

UNIVERSITY OF MALTA

MASTER OF SCIENCE IN MEDICAL PHYSICS

An initial SWOT study of the Medical Physics profession in Malta: The Perspective of Medical Physicists

Jason SCHEMBRI

A dissertation submitted in partial fulfilment of the requirements of the award of

Master of Science in Medical Physics of the University of Malta

2022



University of Malta Library – Electronic Thesis & Dissertations (ETD) Repository

The copyright of this thesis/dissertation belongs to the author. The author's rights in respect of this work are as defined by the Copyright Act (Chapter 415) of the Laws of Malta or as modified by any successive legislation.

Users may access this full-text thesis/dissertation and can make use of the information contained in accordance with the Copyright Act provided that the author must be properly acknowledged. Further distribution or reproduction in any format is prohibited without the prior permission of the copyright holder.

To Abel Makkonen Tesfaye

For fuelling my motivation with his sonically versatile studio albums

Acknowledgements

Firstly, I would like to thank my principal supervisor Professor Carmel J. Caruana for all the help and knowledge he passed on to me in this academic year especially during this dissertation.

I would also like to thank my co-supervisor Mr Sam Aguis, President of the Malta Association of Medical Physics (MAMP), for his input in my dissertation whilst also allowing me to make use of MAMP to recruit local clinical medical physicists in Malta to conduct this study.

I also want to thank Ms Chantelle Said, secretary of MAMP, who kindly acted as intermediary in this study and encouraged other medical physicists to respond to my questionnaire.

I also cannot forget to thank all the local medical physicists who took the time out of their very busy schedules to take part in this study as without their insight I would not have been able to fully complete this dissertation.

My family and friends for continually encouraging me during this postgraduate course, and any past educators of mine who were all part of my educational journey, you have my deepest thanks.

An initial SWOT study of the Medical Physics profession in Malta: The Perspective of Medical Physicists

Abstract

Background: In Malta, medical physics is a relatively new profession and thus far, there has not yet been a systematic situational analysis to determine how the profession is currently progressing. A SWOT study of the medical physics profession in Malta was therefore deemed necessary such that a strategic plan may be developed.

Objectives: (a) To develop a vision statement for the profession in Malta. (b) To analyse the current state of the profession in Malta by establishing internal strengths and weak-nesses of the profession, and external opportunities and threats with respect to the desired vision statement as perceived by the medical physicists themselves. (c) To develop an initial set of strategic objectives for the medical physics profession in Malta.

Research Methodology: A mixed-method research approach was used in this study, the philosophical perspective was pragmatic, the research strategies were a qualitative survey of relevant documents for the vision statement and a mostly quantitative survey of the opinions of Maltese medical physicists to identify the SWOT themes. The data collection techniques were document analysis for the vision statement and a semi-structured online anonymous questionnaire for the SWOT themes, the data collection tool for the SWOT was a specially designed thematic template, the data analysis technique was SWOT analysis with the use of Microsoft Excel for numerical data.

Results: A vision statement was proposed and the SWOTs of the medical physics profession in Malta as perceived by the local medical physicists with respect to the vision statement were identified.

Conclusions and Recommendations: The medical physics profession in Malta has many strengths and opportunities. However, there are also some weaknesses and threats. Strategic objectives for the profession were proposed. It is suggested that further research be conducted to clarify the SWOTs further by including in-depth interviews with both medical physicists and other stakeholders.

Keywords: Medical physics profession, medical physicists, SWOT analysis, vision statement.

Contents

Ac	knov	ledgements	ii
Ał	ostrac	t	iii
Ca	onten	S	iv
Li	st of '	fables	viii
Li	st of l	Definitions	ix
Li	st of A	Acronyms	xi
1	Intr	oduction to the study	1
	1.1	Introduction	. 1
	1.2	Problem Statement	. 1
	1.3	Background and Context	. 1
	1.4	Objectives of the Study	. 2
	1.5	Scope of the Study	. 2
	1.6	Research Design	. 3
	1.7	Ethical Considerations	. 3
	1.8	Relevance of the Study	. 4
	1.9	Conclusion	. 4
2	Lite	rature Review	5
	2.1	Introduction	. 5
	2.2	Medical Physics in Healthcare	. 5
	2.3	Employment Status and Specialties of Medical Physicists	. 6
	2.4	Competences and Responsibilities of Medical Physics Professionals .	. 7
		2.4.1 Responsibilities of the Medical Physics Profession as specified	
		by EU Directive 2013/59/Euratom	. 8
		2.4.2 Competences of a Medical Physics Professional as specified by the 'European Guidelines on Medical Physics Expert' project (RP174)	. 8

		2.4.3	Responsibilities of the Medical Physics Profession as specified by the European Federation of Organisations for Medical Physics (EFOMP)	9
		2.4.4	Responsibilities of the Medical Physics Profession as specified	-
		2	by the International Atomic Energy Agency (IAEA)	11
		2.4.5	Responsibilities of the Medical Physics profession as specified by Maltese legislation	12
	2.5	The M	edical Physics Profession in Malta – A Brief History	13
	2.6	Import	ance of Vision Statements for a profession	13
	2.7	Import	ance of Strategic Planning for the development of the profession	14
	2.8	Use of	SWOT Analysis for the development of strategic plans	15
	2.9	SWOT	'Analysis and its use in Healthcare	16
	2.10	Conclu	ision	18
3	Rese	arch M	lethodology	19
	3.1	Introdu	action	19
	3.2	Resear	ch Approach	19
	3.3	Philoso	ophical Perspective	20
	3.4	Resear	ch Strategies	20
		3.4.1	Research strategy for developing the vision statement for the Medical Physics profession in Malta	20
		3.4.2	Research Strategy for the identification of SWOT themes	20
	3.5	Data C	Collection Techniques/Procedures/Tools	21
		3.5.1	Data Collection Techniques/Procedure/Tool for the vision state-	21
		352	Data Collection Techniques/Procedure/Tool for identifying the	21
		5.5.2	SWOT themes	21
	3.6	Data A	nalysis Techniques	23
		3.6.1	Method of analysis for the vision statement	23
		3.6.2	Method of analysis for the SWOT themes	23
	3.7	Pilotin	g the Questionnaire	24
	3.8	Ethical	Considerations	24
	3.9	Limita	tions of the Research Methodology	24
	3.10	Conclu	ision	24
4	Resu	lts and	Discussion	25
	4.1	Introdu		25
	4.2	Vision	Statement for the Medical Physics profession in Malta	25
	4.3	Questi	onnaire Section 1: General Questions	26
	4.4	Questi	onnaire Section 2: Possible Internal Strengths of the Medical s profession in Malta	37
	45	Ouecti	onnaire Section 3: Possible Internal Weaknesses of the Medical	51
	т.Ј	Physic	s profession in Malta	40
	46	Onectio	onnaire Section 4. Possible External Opportunities for the Medical	-10
	1.0	Physic	s profession in Malta	45

	4.7	Questionnaire Section 5: Possible External Threats to the Medical	40
	18	Questionnaire Section 6: Concluding Questions	49 51
	4.0 1 Q	Summary: Strengths of the Medical Physics profession in Malta with	51
	ч.у	respect to the vision statement	52
	4 10	Summary: Weaknesses of the Medical Physics profession in Malta with	52
	7.10	respect to the vision statement	53
	4 1 1	Summary: Opportunities of the Medical Physics profession in Malta	00
	7.11	with respect to the vision statement	54
	4 12	Summary: Threats of the Medical Physics profession in Malta with	51
	1.12	respect to the vision statement	55
	4 13	List of strategic objectives for the profession	55
	1.15	4 13 1 Strengthening of Internal Strengths	55
		4 13 2 Reducing Internal Weaknesses	55
		4.13.2 Grasping Available Opportunities	56
		4.13.4 Countering External Threats	57
	1 1 1	Conclusion	57
	4.14		57
5	Cone	clusions and Recommendations	58
	5.1	Introduction	58
	5.2	Summary of Conclusions from the Study	58
	5.3	Recommendations for Professional Practice	59
	5.4	Avenues for Future Research	59
	5.5	Conclusion	60
Re	feren	ces	61
A	EU I	Directive 2013/59/Euratom Article 57(1b) and Article 83(2)	67
B	List	of knowledge, skills and competences for the MPE as specified by the	
	EU g	guidelines for the MPE project	70
С	Com	petences of a MPP specific to the specialties of Radiation therapy,	
	Nucl	ear Medicine, and Diagnostic and Interventional Radiology as specified	
	by [A	AEA	85
	0		
D	Οςςι	ipational and Public Radiation Protection as specified by IAEA	92
E	Subs	idiary Legislation 585.01, 2018, Regulation 107	97

F	Characteristics that suggest the need for strategic planning	100
G	List of processes for creating a strategic plan by using SWOT methodology	102
Н	Five points to follow for the development of effective questions	104
I	Questionnaire	106
J	Questionnaire Evaluation Sheet	121
K	FREC Approval	124

List of Tables

3.1	Benefits and limitations of anonymous questionnaires	22
3.2	Design of the SWOT template.	23
4.1	Crosstab for Specialty Area and Device Category of Q1.8	28
4.2	Crosstab for Specialty Area and Device Category of Q1.9.	29
4.3	Crosstab for Specialty Area and Device Category of Q1.10	30
4.4	Crosstab for Specialty Area and Device Category of Q1.11	31
4.5	Crosstab for Specialty Area and Device Category of Q1.12	32
4.6	Crosstab for Specialty Area and Device Category of Q1.13	33
4.7	Crosstab for Specialty Area and Device Category of Q1.14	33
4.8	Crosstab for Specialty Area and Device Category of Q1.15	34
4.9	Crosstab for Specialty Area and Device Category of Q1.16	35
4.10	Statistical summary of the results obtained for Q1.17 till Q1.19	36
4.11	Statistical summary of the results obtained for Q2.1 till Q2.8	38
4.12	Statistical summary of the results obtained for Q3.1 till Q3.10	41
4.13	Statistical summary of the results obtained for Q4.1 till Q4.11	46
4.14	Statistical summary of the results obtained for Q5.1 till Q5.5	50
4.15	Statistical summary of the results obtained for Q6.3	51

List of Definitions

- **competence** means having the capacity to effectively use information and/or methodological skill sets in work or study contexts and for the advancement of one's career and self. The European Qualifications Framework (EQF) defines competence with reference to autonomy and responsibility (EQF, 2008).
- **laser safety expert** means a person, also known as Laser Safety Adviser in some countries, who is certified to provide advice with regards to the safe utilisation of lasers as per Directive 2006/25/EC of the European Parliament and of the Council of 5 April 2006 on the minimum health and safety requirements regarding the exposure of workers to risks arising from physical agents (artificial optical radiation) (19th individual Directive within the meaning of Article 16(1) of Directive 89/391/EEC) (OJ L 114, 27.4.2006, p. 38) (2006/25/EC, 2006).
- **magnetic resonance safety expert** means a person that is capable to offer expert guidance on the engineering, analytical, and managerial elements of the secure clinical utilisation of magnetic resonance devices (Hand et al., 2012).
- **medical physics expert** means an individual or, if provided for in national legislation, a group of individuals, having the knowledge, training, and experience to act or give advice on matters relating to radiation physics applied to medical exposure, whose competence in this respect is recognised by the competent authority (2013/59/EURATOM, 2013).
- **physical agents** means a group of sources of energy. Examples of physical agents include noise, vibration, electromagnetic radiation, electricity, and extremes of temperature (OSHwiki, 2022).
- **radiation protection expert** means an individual or, if provided for in the national legislation, a group of individuals having the knowledge, training and experience

needed to give radiation protection advice in order to ensure the effective protection of individuals, and whose competence in this respect is recognised by the competent authority (2013/59/EURATOM, 2013).

SWOT analysis means a framework that determines the internal strengths and external opportunities a profession may use to achieve its goals, as well as seeking to reduce any internal weaknesses and protect against external threats (Lewis and Littler, 1997).

List of Acronyms

- AI Artifical Intelligence.
- **CPCM** Council of Professions Complementary to Medicine.
- CQMP Clinically Qualified Medical Physicist.
- **D&IR** Diagnostic and Interventional Radiology.
- **DRL** Diagnostic Reference Level.
- **EFOMP** European Federation of Organisations for Medical Physics.
- **ENEN+** European Nuclear Education Network.
- **EQF** European Qualifications Framework.
- **ESMPE** European School for Medical Physics Experts.
- **EUTEMPE** European Network for Training and Education of Medical Physics Experts.
- fMRI Functional Magnetic Resonance Imaging.
- FREC Faculty Research Ethics Committee.
- GDP Gross Domestic Product.
- IAEA International Atomic Energy Agency.
- **ICT** Information and Communication Technology.
- **IPEM** Institute of Physics and Engineering in Medicine.
- **ISCO** International Standard Classification of Occupations.
- LINAC Linear Particle Accelerator.

- LN Legal Notice.
- LSE Laser Safety Expert.
- **MAMP** Malta Association of Medical Physics.
- MP Medical Physics. Medical Physicist. .
- MPE Medical Physics Expert.
- **MPP** Medical Physics Professional (i.e., Medical Physicists and Medical Physics Experts).
- MR Magnetic Resonance.
- **MRI** Magnetic Resonance Imaging.
- MRSE Magnetic Resonance Safety Expert.
- MRSO Magnetic Resonance Safety Officer.
- NM Nuclear Medicine.
- PEST Political, Economic, Social-Psychological, Technological, and Scientific.
- PET Positron Emission Tomography.
- **QA** Quality Assurance.
- QC Quality Control.
- **RO** Radiation Oncology.
- **RP** Radiation Protection.
- RPC Commission for the Protection against Ionising and Non-Ionising Radiation.
- **RPE** Radiation Protection Expert.
- **RT** Radiotherapy.
- SL Subsidiary Legislation.
- SMP Specialist Medical Physicist.

- SWOT Strengths, Weaknesses, Opportunities, Threats.
- **TB** Tuberculosis.
- **UREC** University Research Ethics Committee.
- **US** Ultrasound.

Chapter 1

Introduction to the study

1.1 Introduction

This chapter outlines the problem statement of this study and why it was chosen. The background and context of the problem are discussed along with the objectives, scope, and the research design. This is followed by a brief discussion of the ethical considerations and relevance of the study. The chapter ends with an overview of the dissertation.

1.2 Problem Statement

As the Medical Physics (MP) profession in Malta is still in its initial stages, there has not yet been a systematic situational analysis to determine how the profession is currently progressing. A Strengths, Weaknesses, Opportunities, Threats (SWOT) study of the MP profession in Malta is therefore crucial in order to increase strengths, take advantage of opportunities, overcome current weaknesses, and counter threats, such that a strategy plan for the profession in Malta may be developed.

1.3 Background and Context

In Malta, MP is a relatively new profession. Discussions on the profession's development and recognition began in 2007 with a partnership between a professor at the University of Malta and the three existing clinical Medical Physicists (MPs). At the time, clinical MPs were employed by the Ministry for Health under the title of 'scientific officers' because the profession was relatively unknown and was not legally recognised. Additional MP trainees were employed in 2012, when the Ministry of Health acknowledged the relevance of the profession. A grant from the European Social Fund was won due to a strong

partnership between the Ministry of Health and the University of Malta's MP Department (Caruana and Grech, 2016). According to the Council of Professions Complementary to Medicine (CPCM) (2014), MP was classified as a regulated profession under the said council by a legal notice issued by the Maltese government in 2014. The national Commission for the Protection against Ionising and Non-Ionising Radiation (RPC) (2021) recognises and certifies Medical Physics Experts (MPEs) in Malta. Malta currently has a total of 20 clinically qualified MPs who at present work within the specialties of Radiation Oncology (RO), Diagnostic and Interventional Radiology (D&IR), and Nuclear Medicine (NM) (Caruana and Grech, 2016, CPCM, 2014). The Malta Association of Medical Physics (MAMP) holds meetings and seminars as part of its national activities, with members presenting clinical issues, research studies and accomplishments (MAMP, 2019). MAMP encourages members to participate in the activities of the Association and the European Federation of Organisations for Medical Physics (EFOMP). Legal Notice (LN) 397 of (2012) and LN 288 of (2019) established the M.Sc. in Medical Physics and B.Sc.(Hons) in Physics, Medical Physics and Radiation Protection respectively in Malta. The latter was set up to ensure a steady number of applicants to the Masters and hence new entrants to the profession (given the low number of physics and engineering graduates) and that Masters graduates are well prepared for entry to their MP traineeship since a one year M.Sc. in Medical Physics was simply not enough to acquire the necessary knowledge and skills given the rapid expansion of medical device technology and use of physical agents in medicine.

1.4 Objectives of the Study

The objectives of the study were the following:

- a) To develop a vision statement for the MP profession in Malta.
- b) To analyse the current state of the profession by establishing internal strengths and weaknesses of the profession, and external opportunities and threats with respect to the desired vision statement and as perceived by the MPs themselves.
- c) To develop an initial set of strategic objectives for the MP profession in Malta.

1.5 Scope of the Study

The study was delimited to the perspective of the local registered clinical Medical Physics Professional (i.e., Medical Physicists and Medical Physics Experts) (MPP).

1.6 Research Design

The research design provides a methodological framework for the conduct of the research study and for data collection and analysis whilst also providing a conceptual framework to interpret and acknowledge the research (Clark et al., 2021). A research design consists of a research approach (quantitative, qualitative or mixed), philosophical perspective, and research methodology (research strategy, data collection technique, data collection tool and analysis).

- a) Research Approach: A mixed-method research approach was adopted for this study. Qualitative document analysis was used to develop the vision statement whilst a mostly quantitative survey was used to discover SWOT issues as seen from the perspective of the clinical MPs.
- b) Philosophical Perspective: A pragmatic philosophical perspective was utilised since a real-life setting is being studied (Dou, 2021).
- c) Research Methodology: The methodologies for each objective are given below:
 - i) Objective 1: An initial vision statement was developed via a document analysis of EFOMP policy statements and international literature with regard to the MP profession.
 - ii) Objective 2: An online semi-structured anonymous questionnaire was used to collect SWOT themes from the perspective of the local MPs.
 - iii) Objective 3: The SWOT analysis results were used to create an initial list of strategic objectives via standard SWOT strategic planning techniques.

1.7 Ethical Considerations

An information letter, consent form and link to an anonymous online semi-structured questionnaire was delivered to all local clinical MPPs via the Secretary of the MAMP acting as intermediary to the research study. Participation in the study was entirely voluntary and the participants were advised that filling in and submitting the questionnaire implied consent. Furthermore, the participants were also advised that the data collected could not be traced back to them. In fact, they were informed that, the researcher, would not be able to identify from whom the information came from nor whether any particular MP participated in the study or otherwise. This study has been approved by the University Research Ethics Committee (UREC) of the University of Malta.

1.8 Relevance of the Study

The relevance of the study for the various stakeholders is as follows:

For patients: A committed team of MPPs who understand the vision of how the profession can evolve guarantees that the profession's members have the capacity to grow and develop, and hence provide patients with effective, secure, and efficient services.

For the MP profession and professional practice: The MPPs involved would be able to acquire a future vision for the profession and a research-based list of strategic objectives to attain the vision and hence enhance the state of the profession in Malta.

1.9 Conclusion

This chapter presented an introduction to the study. Chapter two provides a critical review of the literature. Chapter three describes the research methodology whilst chapter four presents the results together with their analysis and discussion. Chapter five summarises the main conclusions of the study, proposes recommendations arising from the study and suggests directions for future research.

Chapter 2

Literature Review

2.1 Introduction

This chapter presents a review of the literature about the MP profession focusing in particular on relevant EU directives and guidelines, documents of international and European organisations such as the International Atomic Energy Agency (IAEA), and EFOMP, and also Maltese legislation. In addition, this chapter presents a critical review of the literature regarding strategic planning and the utilisation of the SWOT methodology in healthcare.

2.2 Medical Physics in Healthcare

The history of MP may be attributed to have started with Sanctorius, who made use of a measuring technique in physics, which was weighing, and applied it to medicine as a method of health monitoring in the early seventeenth century. The name 'Medical Physics' (or, more precisely, Physique médicale) was first used in 1778 by Félix Vicq d'Azir who was the general secretary of the Société Royale de Médecine. According to Vicq d'Azir medical training should include MP as well as other basic sciences. 'La Médecine eclairée par les sciences physiques' was a brief-lived newspaper that Comte de Fourcroy founded in 1791. He puts forth his own viewpoint in the preface to the first issue stating that the study of medicine always begins with the study of physics and without having studied physics, one cannot become a physician (Duck, 2014). The 1814 updated version of Nysten's medical lexicon eventually contained a definition of MP. This definition is noteworthy for being thorough and concise: "Physics applied to the knowledge of the human body, to its preservation and to the cure of its illnesses." In 1824, the first physics textbook for medical students was published by Pelletan (Pelletan, 1824). Carlos Matteucci was a physics professor who published a book in 1844 with the aim of having an adequate combination of physics principles and its new medical applications (Matteucci, 1844). An addendum on medical physics for 'Müller-Pouillet's Lehrbuch der Physik' was published in 1850s by a physiologist Adolf Fick. Fick later on published 'Die medizinische Physik' in 1855. A book with the title 'A Text-Book of Medical Physics for the use of Students and Practitioners of Medicine' was published as the first MP book in English by John Christopher Draper (Draper, 1885).

After a century of sporadic advancements in therapeutic medical treatment, physics applied to medicine saw a significant relaunch during the last ten years of the nineteenth century. The discovery of radioactivity by Henri Becquerel in 1896, Röntgen's discovery of X-rays in 1895, and Marie Curie's subsequent discovery of radium are what gave rise to medical imaging and radiotherapy. The Nobel Prize winning Danish physician Niels Finsen was a pioneer in the use of infrared and ultraviolet light for medicinal purposes. He developed the "red room" method of treating smallpox lesions in 1893, and in 1895 he demonstrated that UV therapy could be used to cure the deforming type of Tuberculosis (TB) known as lupus vulgaris. Thus, phototherapy came into effect. The birth of Ultrasound (US) may be traced back to the discovery of piezoelectricity by Jacques and Pierre Curie in 1881 (Duck, 2010, 2014). Heinrich Hertz's discovery of radio waves in 1888 led to Arséne d'Arsonval, in 1895, to develop the first therapeutic high-frequency heat treatment device (Duck, 2014). MPs now contribute significantly to the majority of healthcare fields.

2.3 Employment Status and Specialties of Medical Physicists

MPs work in hospitals or as consultants to hospitals, the administrative sector, industry, or academia (Evans et al., 2016). Regardless of where the MP works, all MPs apply their physics expertise to the solution of problems in healthcare, and they do so in various but complementary contexts.

The academic and industrial sectors are home to the first category of MP. Teaching to MP students and other medical and healthcare professions and/or doing research that is directly or indirectly related to healthcare are the primary responsibilities of academic MPs. MPs carry out research in a wide range of fields connected to the use of physics in

healthcare. This sort of work is far wider in scope than that of a hospital based MP and includes not just the use of medical devices but also the study of the effects of physical agents on human health, physiological modelling, computer applications in medicine, and the development of novel imaging tools and technologies (Evans et al., 2016).

The primary responsibility of hospital based MP is to use their physics expertise to ensure the optimised, ethical, efficient, safe and scientific use of medical devices as well as development of new clinical services (Christofides et al., 2016). Regarding medical devices, several EU directives have been promulgated.

MPs employed in hospitals are frequently found in departments such as D&IR, RO, NM, and Radiation Protection (RP) that use equipment based on x-rays or radioactive sources and these are traditionally the four principal areas of specialty for MPs. Ionising radiations, in the healthcare industry, are commonly referred to as the photons or particles associated with having adequate energy to ionise molecules in the body if their energy is absorbed by tissues. MPs also offer services in bioengineering, medical electronics, Information and Communication Technology (ICT), physiological measurements, audiology, clinical applications of non-ionising radiation (e.g., lasers, ultraviolet light, and microwaves), and Magnetic Resonance (MR) and US imaging (Evans et al., 2016).

A person who is qualified to practise independently and is eligible to register as a healthcare professional with the title 'Medical Physicist' in one or more of the aforementioned specialties is referred to as a Clinically Qualified Medical Physicist (CQMP). A CQMP will then progress to become a Specialist Medical Physicist (SMP) by accumulating more advanced clinical experience and completing specialist training in their specialty area (Christofides et al., 2016). EU Directive 2013/59/EURATOM (2013) uses the title 'Medical Physics Expert' in lieu of 'Specialist Medical Physicist'.

2.4 Competences and Responsibilities of Medical Physics Professionals

The following subsection presents the competences and responsibilities of a MPP as specified by EU Directive 2013/59/EURATOM (2013), the 'European guidelines on MPE' project (EU Commission, 2014), European and international organisations such as EFOMP (EFOMP, 1984), (Hand et al., 2012), (Caruana et al., 2014), (Evans et al.,

2016), (Christofides et al., 2016), (Caruana et al., 2018), (Busoni et al., 2021) and IAEA (IAEA, 2013), and Maltese Subsidiary Legislation (SL) (S.L.585.01, 2018).

2.4.1 Responsibilities of the Medical Physics Profession as specified by EU Directive 2013/59/Euratom

According to the EU Directive 2013/59/EURATOM (2013), to guarantee the proper safety of patients undergoing medical radiodiagnostic and radiotherapeutic treatments, a high level of competence and a clear description of roles and tasks among all specialists participating in medical exposure are essential. This is applicable to MPs and other equivalent experts involved in medical radiological procedures. In the case of MPEs this is provided in Article 57(1b) and Article 83(2) of the Directive as shown in Appendix A.

2.4.2 Competences of a Medical Physics Professional as specified by the 'European Guidelines on Medical Physics Expert' project (RP174)

The EU guidance on medical physics expert (EU Commission, 2014) is intended to improve the implementation of provisions related to the MPE in the Directive and to facilitate harmonisation of the role and education and training of MPs among the member states with the goal of enhancing cross-border mobility. The core competences for the key activities of an MPE as specified by EU guidelines on MPE project are the following:

- a) Scientific problem solving service: Identifying instances of less than ideal performance while using radiological instruments and verifying that suggested adjustments have improved performance of the device.
- b) Dosimetry Measurements (and other physical agents as appropriate): Evaluation, calibration, and upkeep of dosimetry-related instrumentation; independent verification of dose-related quantities supplied by devices (including software devices).
- c) Patient Safety / Risk Management: Monitoring of radiological devices and analysis of clinical protocols involving dosage audit and assessment of incidents that include inadvertent medical exposures.
- d) Occupational & Public Safety / Risk Management (when there is an impact on medical exposure or own safety): Balancing occupational risk and patient requirements in the correlation of medical and occupational exposures.

- e) Clinical Medical Device Management: Offer technical assistance and partake in the specification, selection, acceptance testing, Quality Assurance (QA), commissioning, installation design, and decommissioning of medical radiological devices.
- f) Clinical Involvement: Performing routine radiation protection and Quality Control (QC) measures for the benefit of the patient. Making use of phantoms to make the necessary testing and calibration on the modalities before the device may be used on patients.
- g) Development of Service Quality and Cost-Effectiveness: Implementation of new MP services, and clinical protocols whilst providing appropriate consideration to financial concerns.
- h) Expert Consultancy: Providing professional guidance to the general public in clinics that might not have MP knowledge.
- i) Education of Healthcare Professionals (including MP trainees): A MPP should also be involved in the training of MP students and the clinical training programmes.
- j) Health Technology Assessment: Assuming control over the physics portion of health technology evaluations pertaining to medical radiological equipment and applications of radioactive sources.
- k) Innovation: Creating new or changing already existing equipment (including software) and using procedures to better handle clinical issues that have not yet been solved.

Appendix B presents a detailed list of competences that an MPE is expected to possess for each activity.

2.4.3 Responsibilities of the Medical Physics Profession as specified by the European Federation of Organisations for Medical Physics (EFOMP)

Maintaining a high level of service in the hospital is the clinical MPs primary duty (EFOMP, 1984). There are two distinct functions of an MPP according to the EFOMP. One function is the one that MPs can perform their clinical roles through their scientific expertise in the field of MP. A second function is that they exercise managerial roles at a high level when they serve as administrators for the MP service (Evans et al., 2016). In close collaboration with other healthcare professionals, MPs are in charge of the standardisation and calibration of medical equipment as well as the precision

and safety of physical procedures used in clinical applications. The creation of new procedures, physical methods, and equipment is one of their other duties, along with research, clinical oversight of the use of medical devices, health technology assessment, and service development. They also have a duty to educate and teach future MPs as well as technical employees in the MP department. This obligation extends to physicians, nurses, medical technical assistants, and more. In their managerial role, often the head of the MP department, will be accountable to central administration and be in charge of financial control as well as general hospital and/or a local region organisation (Evans et al., 2016). Caruana et al. (2014) provides the updates necessary to describe the key activities of a MPP following the publication of (EU Commission, 2014). Caruana et al. (2018) elaborates on the role and competences of a MPP as per EU Directive 2013/59/EURATOM (2013) and by the European guidelines on the MPE project (EU Commission, 2014).

EU Directive 2013/59/EURATOM (2013) is concerned with the role of the MPE in ionising radiation only. EFOMP has always pushed for a defined role for MPPs also in non-ionising radiations. For example, the Magnetic Resonance Safety Officer (MRSO) and the Magnetic Resonance Safety Expert (MRSE) are two professionals that EFOMP advises for managing safety inside MR units (Hand et al., 2012). Occupational group 2111 of the International Standard Classification of Occupations (ISCO) describes MPs as the preferred profession for the position of magnetic resonance safety expert when it states that physicists have the duty of "ensuring the safe and effective delivery of radiation (ionising and non-ionising) to patients to achieve a diagnostic or therapeutic result as prescribed by a medical practitioner" (Hand et al., 2012, International Labour Organisation, 2012).

Specialised Magnetic Resonance Imaging (MRI) techniques have grown significantly over the past few years, including spectroscopy, diffusion and perfusion weighted imaging, Functional Magnetic Resonance Imaging (fMRI), and synthetic MRI. Many of these techniques are being combined with Artifical Intelligence (AI) which is becoming more prevalent in these fields. Moreover, there has been an emergence and development of novel MRI settings (Positron Emission Tomography (PET)/MRI, Linear Particle Accelerator (LINAC)/MRI, interventional MRI), as well as the approval for the clinical use of extremely high field MRI systems (7 T), with the associated new safety areas concern (Busoni et al., 2021).

EFOMP in its Malaga Declaration (2007) insists that the role of radiation protection of workers and the general public (today known as the radiation protection expert) should be always assigned to MPPs. The Malaga declaration is at present being updated to create an updated vision of the role of MPPs. This updated vision includes a role for MPPs in all medical devices and physical agents used in medicine and including patient, workers, and the general public (Caruana C.J., personal communication, August 7, 2022).

2.4.4 Responsibilities of the Medical Physics Profession as specified by the International Atomic Energy Agency (IAEA)

The duty of the CQMP in clinical practice is to collect and analyse clinical physics data for the diagnosis or treatment of diseases, as well as to optimise or guide other health care professionals to optimise the use of radiation to guarantee the quality and safety of therapeutic or diagnostic procedures (IAEA, 2013). According to IAEA (2013) the responsibilities for a MP are the following:

- a) Calibration and verification of measurement instruments: Regular calibration of the instruments utilised.
- b) Technical supervision of equipment operation and maintenance: Setting up acceptance and commissioning processes for therapeutic, and diagnostic equipment. Working closely with service engineers.
- c) Records and documentation: Demonstrating adherence to the equipment and process guidelines.
- d) Clinical computing and networking: Ensure systems are operational by testing network integration and data transmission.
- e) Research and development: Assessing new technology and looking at the adoption of new practises while supporting the clinical staff's preparation for their use.
- f) Education and training: Providing medical practitioners, nurses, MP trainees, and students, with lectures and training in MP.

IAEA issued the core tasks of a CQMP specific to the specialties of Radiotherapy (RT), D&IR, NM, and RP (IAEA, 2013). The IAEA does not distinguish between MP and MPE. A CQMP is equivalent to the MPE. It also does not distinguish between MPE and Radiation Protection Expert (RPE) in healthcare. For the IAEA, RP is part of the duties

of the MP.

Appendices C and D present a detailed list of competences that a MPP is expected to possess for each corresponding specialty according to the IAEA.

2.4.5 Responsibilities of the Medical Physics profession as specified by Maltese legislation

The fundamental legislation for radiation protection in Malta the Basic Safety Standards for Ionising Radiation Regulations, 2018 are stipulated in LN 210 (L.N.210, 2018). This LN provides the general responsibilities of an MPE under regulation 107. As of 2020, LN 210 was amended by LN 184 (L.N.184, 2020) and the latest regulation is the SL 585.01 (S.L.585.01, 2018). It can be stated that there is a general agreement when comparing the duties of an MPE as stipulated in the Maltese legislation and in the EU Directive 2013/59/EURATOM (2013) Article 82. Regulation 107 of the SL can be found in Appendix E.

MPs in Malta specialise in one of three areas: RO, NM, and D&IR (Caruana and Grech, 2016). According to the Ministry for Health in Malta (Duties DIR, 2022), a MPE in D&IR shall provide guidance about medical radiological tools and make a contribution to the following:

- a) Dosimetry: Using appropriate equipment to analyse the dosage to be given to the patients during treatment.
- b) Radiation protection programme: Participating in the department such as contributing to equipment selection in the programme.
- c) Safety Culture: Dedicated to question any negligence and improve own knowledge in terms of safety. To maintain ongoing efficiency and optimum usage of medical radiological devices, such as overseeing routine patients.
- d) Diagnostic Reference Levels (DRLs) and Dose Audits: Assuming responsibility of the dosage distributed to an individual. Dose audits of various medical modalities are also to be carried out.
- e) Optimisation: Optimising procedures prior to their use on patients by utilising phantoms and simulations first.

- f) Quality Assurance Programme: Offer technical guidance and take part in specification, selection, and acceptance testing, commissioning, installation design and decommissioning of medical radiological devices.
- g) Accidental or unintended medical exposures: The MPE is responsible for any incidents due to unintended medical exposures. The MPE must also report, where needed, any incidents to the local authorities responsible for such matters.
- h) Education and training: The MPE must contribute to the education of MP students and the clinical training of MP trainees and other staff in relevant departments related to D&IR.

Similar documents describe the duties of a MPP within RO and NM.

2.5 The Medical Physics Profession in Malta – A Brief History

Insight to the development of the MP profession in Malta has already been provided in Chapter 1, Section 1.3. In addition, most Maltese CQMPs obtained their clinical training in the UK (Caruana et al., 2021). This has been advantageous for the profession because they were taught by some of the leading European MPs. The Maltese coordinator of the project insisted with the UK MPs that they utilise the IAEA training manuals (IAEA, 2009), (IAEA, 2010), (IAEA, 2011) in all three specialty areas: RO, D&IR, and NM. The MP profession in Malta is now developing its own training programme locally, which will again be in accordance with the IAEA standards (Caruana et al., 2021).

2.6 Importance of Vision Statements for a profession

Vision statements are important because they can assist any entity (e.g., profession, organisation, individual) to define long-term direction and objectives that must be achieved. According to (Bart, 2001), the vision statement describes the desired future state of the entity within the ambit of its mission statement. A vision statement should be able to contribute to the organisation's or profession's identity (Cardona and Rey, 2006). The social impact that a vision statement leaves to the various stakeholders should also be considered (Collis and Rukstad, 2008). These definitions can be reduced to three core elements that a vision statement should embody: a fundamental ideology, a vision for the future, and acknowledgement of service to stakeholders. A description and purpose of a profession's ideals is contained in the vision statement's fundamental ideology. The profession's envisioned future is a description of the services it can provide in the future and a specification of how the profession benefits its stakeholders, including patients, other healthcare professionals as well as community and society.

MPPs have historically been associated with ionising radiation based medical devices. The role of MPPs must evolve along with the medical device technologies and the utilisation of physical agents aside from ionising radiation. For instance, non-ionising radiation based imaging and therapeutic technologies need MPPs attention as well because they are already well-established. The MPP frequently receives requests for help in other fields such as sophisticated physiological monitoring, AI, and nanotechnology because the variety of medical radiological tools as well as physical agents employed in hospitals nowadays transcend the utilisation of both ionising and non-ionising radiation. EFOMPs competent committee is currently updating its vision, possibly as an update to the Malaga Declaration, for the MP profession in the direction of medical devices and physical agents (Caruana C.J., personal communication, August 7, 2022).

2.7 Importance of Strategic Planning for the development of the profession

In order to determine a profession's strategies, it is necessary to identify the needs of society, environmental factors, organisational capabilities, and any internal weaknesses of the profession and external threats. The strategy of any profession is a list of objectives to help it achieve its vision and the means to achieve those objectives. Lacking strategic planning, a profession may degrade due to constantly responding to external pressures (Lawlor, 2010). According to Perera and Peiró (2012) all kinds of professions, including those in healthcare, may be guided by strategic planning, which is a perfectly legitimate and practical technique. According to Caruana (2020) "strategic planning is a systematic process for developing and achieving a desired future vision."

It is strongly advised that the processes in creating a strategic plan be carried out in collaboration with the profession's members as this will guarantee that everyone is aware of what is occurring and why certain actions were taken in a particular way. Additionally, it provides a forum for the members of the profession to engage in the creation and execution of the vision. This would help prevent unpleasant emotions in the future,

when the strategic plan is being implemented (Caruana, 2020). Perera and Peiró (2012) outlined five characteristics that when combined, suggest the need for strategic planning. These are:

- a) Clientele who are more aware and demanding.
- b) Increasing numbers of competent and professional rivals.
- c) Limited organisational resources.
- d) The client is given more attention than the service.
- e) Size and intricacy of the organisation.

Further insight can be obtained in Appendix F.

Each of the above mentioned factors place a profession in a situation where their internal and external circumstances as well as those of their clients are always changing. It should be emphasised that there is no single strategic planning model for all professions. Each profession's distinct qualities must be considered whilst designing the formal strategic planning framework (Steiner, 2010).

2.8 Use of SWOT Analysis for the development of strategic plans

SWOT analysis is one of several methods that may be applied to the strategic planning process of an entity whether organisation or profession (Phadermrod et al., 2019). SWOT analysis' primary benefit is its simplicity, which has led to its continuous usage in different organisations and top businesses (Ghazinoory et al., 2011). The majority of strategy development frameworks being utilised are currently based on a SWOT technique model. Caruana (2020) outlined a list of processes for creating a strategic plan by utilising the SWOT methodology. This list can be accessed in Appendix G.

Caruana (2020) outlined all the aspects of SWOT which are summarised hereunder:

a) Strength (S): The profession's strengths are those attributes that will enable it to accomplish the vision. To prevent the strength from becoming a threat the profession needs to maintain the quality of its strengths. The core competences are very significant strengths that are seen as a strong advantage when compared to other competing professions.

- b) Weakness (W): Weaknesses of the profession will hinder it from fulfilling the vision. The critical weaknesses are of great significance. Contrary to core competences, which guarantee a profession survives, critical weaknesses are a disadvantage that, if not addressed, can hinder it severly.
- c) Opportunity (O): They are advantageous issues arising from the surrounding environment that may be used to accomplish the vision. The profession will be able to advance and expand if the key opportunities, that is those opportunities with the greatest benefit value, are identified.
- d) Threat (T): These are the drawbacks from the surrounding environment that, if not addressed, would prevent the vision from being accomplished. Existential threats, are threats whose effects might entirely disrupt the vision of the profession. They must be quickly recognised and eradicated.

SWOT analysis offers a way to pinpoint the factors that obstruct or improve the achievement of the intended vision. Such issues must be identified before the strategic plan can be developed.

2.9 SWOT Analysis and its use in Healthcare

Incorporating a SWOT analysis in the health sector is unquestionably a useful tool for medical staff, and its applications range from improving patient care, increased effectiveness in equipment use, and higher standards in hospital administration. Given its many benefits, SWOT analysis has a bright future in healthcare (Turankar et al., 2014). SWOT analysis is also very beneficial in healthcare as it assures that there are imperative advancements in the medical field (Gandolf, 2021).

SWOT analysis was employed by Caruana et al. (2010) for producing a strategic plan for the role of biomedical physicists in Europe for the education of professionals in healthcare. A qualitative analysis of biomedical physics curricula offered to professionals in healthcare throughout Europe recognised the internal strengths and weaknesses. A review of the literature in the fields of healthcare professional education, and higher education led to the identification of external opportunities and threats, which were then categorised according to the traditional Political, Economic, Social-Psychological, Technological, and Scientific (PEST) categories. Aslan et al. (2014) made use of the SWOT analysis to study Turkey's developing healthcare system. The authors address the demands of an expanding population, Turkey requires an increase in healthcare staff. To lessen the burden of hospitalisation on public hospitals, which are mostly focused on chronic illnesses, nursing homes can be utilised. To raise the proportion of healthcare spending in Turkey's Gross Domestic Product (GDP), a more open and unbiased performance management system emphasising patient and employee satisfaction must be created.

Terzic-Supic et al. (2015) employed SWOT analysis such that a list strategic objectives could be obtained for the utilisation of hospital management. The authors believed that programs for formal and informal training are increasingly in demand, particularly for healthcare staff in executive roles. This was described as a necessity for hospitals in an environment that is very competitive.

Von Kodolitsch et al. (2015) used SWOT to develop specific medicinal plans for individuals with aortic illness. The authors began by defining the therapy's objective and listing all evidence-based treatment choices. In a subsequent stage, they evaluated each therapy option's advantages and disadvantages using a SW matrix. The authors employed the SWOT matrix to create a customised medical plan by matching "SW" with "OT" in the final stage, where they evaluated opportunities and threats specific to the particular patient.

Figueira et al. (2016) conducted a SWOT analysis of the current MP profession in Portugal with the purpose of identifying potential measures that may be performed to improve the state of the profession. Based on the response of 83 Portuguese MPs, the most common strength was the dedication of the MPs towards the profession. A weakness mentioned was the lack of professional recognition. The growing need of new installations was seen as an opportunity and one main threat mentioned was the pressure from other professionals such as radiographers that are performing some MPs tasks. The authors concluded that it was beneficial to use SWOT in assisting the MPs in creating a strategic action plan to enhance the existing state of affairs.

Topor et al. (2018) made use of SWOT in interprofessional healthcare education at academic medical centres. A workshop consisting of a webinar and an in-person presentation was conducted and the main objective of this workshop was to educate participants on how to create interprofessional learning activities using SWOT analysis. The session

was well-received by the participants, who felt that it achieved the intended learning goals.

Through the use of a SWOT analysis Andersson et al. (2021) concluded that AI will alter the MP profession by increasing opportunities linked to the use of AI in healthcare. The survey amassed a 40% response rate of which 90% believed that AI might alter how MP is practised, and 81% believe that AI does not pose a risk to the MP profession. The majority of respondents supported using AI in educational programmes. The author's understanding is that occupational readiness for AI was usually poor among the participants and was a weakness that needed to be addressed with some urgency.

Vella et al. (2022) used SWOT to analyse the constancy testing programme in a major hospital in Malta for medical imaging devices. The study showed that the present constancy testing programme has several advantages due to the team having analytical expertise, but also some disadvantages, mostly due to the profession in Malta being so new. Several opportunities for the programme to be enhanced were identified such as there being a rise in the utilisation of diagnostic medical imaging devices which is beneficial for the MPPs. Nevertheless, there are some threats such as possible loss of MPPs who conduct these tests.

2.10 Conclusion

The literature related to the study was critically reviewed in this chapter. The research methodology that was employed in the study will be described and discussed in the next chapter.

Chapter 3

Research Methodology

3.1 Introduction

The project's methodology is described in this chapter. The methodology used needs to be designed in a way that addresses the objectives of the study. A methodology is made up of a research approach, philosophical perspective, research strategy, data collection technique/procedure/tool, and a data analysis technique. A mixed-method research approach was used in this study, the philosophical perspective was pragmatic, the research strategies were a qualitative survey of relevant documents for the vision statement and a mostly quantitative survey of the opinions of Maltese MPs to identify the SWOT themes, the data collection techniques were document analysis for the vision statement and a semi-structured online questionnaire for the SWOT themes, the data collection tool for the SWOT was a specially designed thematic template, the data analysis technique was SWOT analysis.

3.2 Research Approach

A mixed-method research approach was adopted for this study. A mixed-method study is one that uses both qualitative and quantitative methods of data collection and analysis. This kind of research helps a researcher to qualitatively comprehend complex issues as well as to utilise statistical analysis when necessary (Creswell, 1999). According to Rossman and Wilson (1994) this approach is particularly useful because it has the potential to better understand the issues through different lenses and obtain eclectic data that better respond to the perspectives and opinions of the stakeholders relevant to the study. In this study, a qualitative document analysis of relevant documents (legislative and documents from international and professional organisations) was used to formulate the vision statement that would guide the strategic planning whilst a mostly quantitative survey was utilised to generate comprehensive SWOT thematic data from CQMPs in Malta. In this way the data collected would better reflect the actual state of the profession with respect to the vision statement.

3.3 Philosophical Perspective

One of the most prevalent philosophical schools of thought in research today is pragmatism, which was employed in this study. Pragmatism allows for a free choice to use any methodology and research analyses such as to ensure that the study's research objectives are achieved (Legg and Hookway, 2008).

3.4 Research Strategies

A research strategy is a comprehensive plan for carrying out a research study that enables the methodical and timely execution of research to yield high-quality findings and thorough reporting. The research strategies for formulating the vision statement and identifying SWOT themes are described in Subsections 3.4.1 and 3.4.2 below.

3.4.1 Research strategy for developing the vision statement for the Medical Physics profession in Malta

Since we live in a globalised world and since Malta is part of Europe the creation of the vision statement was carried out via a document analysis of relevant EU (and associated national) legislation and the international and European literature about the MP profession.

3.4.2 Research Strategy for the identification of SWOT themes

The research strategy for the identification of the SWOT themes was a mostly quantitative survey among MP professionals in Malta. The aim of the survey was to capture the SWOT themes. The formulation of the queries in the questionnaire was guided by the vision statement. The questionnaire was designed to identify what strengths the profession has with respect to the vision statement, to identify if there were any weaknesses that would hinder the achievement of the vision, any opportunities available to MPs to help them achieve the vision and any threats that would stop them from achieving the vision.

3.5 Data Collection Techniques/Procedures/Tools

The data collection techniques/procedures/tools for the vision statement and the SWOT themes are identified in Subsections 3.5.1 and 3.5.2 respectively.

3.5.1 Data Collection Techniques/Procedure/Tool for the vision statement

The data collection techniques used for the vision statement was document analysis. Document analysis was performed on documents from European (EU Commission, 2014) and/or International organisations that seek to represent or assist MP societies such as EFOMP (EFOMP, 1984), (Hand et al., 2012), (Caruana et al., 2014), (Evans et al., 2016), (Christofides et al., 2016), (Caruana et al., 2018), (Busoni et al., 2021) and IAEA (IAEA, 2013) as well as legal instruments provided by the EU such as the Directive 2013/59/EURATOM (2013). Moreover, local SL (S.L.585.01, 2018) was also analysed.

3.5.2 Data Collection Techniques/Procedure/Tool for identifying the SWOT themes

The primary data collection technique used for identifying the SWOT themes was an online anonymous semi-structured questionnaire. The Google Forms software was used to collect the data. The questionnaire was made available to all CQMPs in Malta and the survey was filled on a voluntary basis. The semi-structured questionnaire with the MPPs aided in presenting viewpoints that were required to assess the current situation of the MP profession in Malta with respect to the suggested vision statement.

Data was collected following ethical clearance from the Faculty Research Ethics Committee (FREC). The questionnaire was placed online so that prospective participants would be able to complete the questionnaire at their own convenience. An online survey was also seen as beneficial when compared to an interview since the data would not have to be transcribed manually which enabled the time that would have been taken to transcribe the surveys, to be used more effectively. It was decided that the questionnaire would be anonymous so that participants would be free to express their opinions without fear of repercussion. A questionnaire is considered anonymous if it does not seek personal information, such as their name or email address. As a result, it is impossible to discern any variables in the replies that may be connected to a respondent. Some advantages and
disadvantages of anonymous questionnaires are listed in Table 3.1.

Establishing the main objectives would be the first stage in developing a questionnaire. What is it that you want to learn? What are the particular research questions? Five criteria were outlined by Fowler Jr and Fowler (1995) as a roadmap for creating effective questions. These five points were used to guide the formulation of the questions. These five criteria can be found in Appendix H.

The online semi-structured questionnaire, took around one hour for the participant to complete and mainly consisted of close-ended questions with an attached Likert scale of the form: 'Strongly Agree', 'Agree', 'Not Sure', 'Disagree', 'Strongly Disagree' whilst other questions required simple 'yes' or 'no' replies. However, a few open-ended questions were also used when it was deemed that more insight needed to be gathered on certain queries. The questionnaire consisted of six sections. Section one consisted of some general questions with the aim of identifying any accreditations the participant might possess whilst introducing them to the roles and responsibilities identified in the proposed vision statement. Section two and three sought to identify further possible internal strengths and weaknesses of the MP profession in Malta respectively with respect to the proposed vision statement whilst sections four and five sought to identify external environmental opportunities and threats of the MP profession in Malta respectively. Finally, section six contained some concluding questions. The full questionnaire can be found in Appendix I.

Benefits	Limitations
Participants are more inclined to respond hon- estly and in more details on certain matters.	Unable to conduct any follow ups.
Higher participation rate.	Specific issues cannot be referenced to the re- spondent to see if the situation has been rectified.
Instantaneous recording of participant responses into a database.	People talk more easily than they write.
Less pressure is exerted on the participants to give an immediate response.	

TABLE 3.1: Benefits and limitations of anonymous questionnaires.

3.6 Data Analysis Techniques

The data analysis techniques used to analyse the vision statement and the SWOT themes are presented in Subsections 3.6.1 and 3.6.2 respectively.

3.6.1 Method of analysis for the vision statement

The EU, EFOMP, IAEA, and LN documents were analysed by searching for the following words within the documents: vision, future, role, competences, responsibilities, duties, key activities. A vision statement was then formulated which encompassed the future mission for MPs as envisaged by EFOMP.

3.6.2 Method of analysis for the SWOT themes

The main aim of the design of the questionnaire was to encourage SWOT topics to be identified. Table 3.2 presents the thematic template that was used to analyse the data and explicitly showing the SWOT categories.

Numerical data analysis was performed by using Microsoft Excel. The quantitative data from the questionnaire were analysed using descriptive statistics. To analyse such data the mean score was obtained. Averaging was used to analyse questions that have an ordinal categorical (Likert) scale (e.g., strongly agree, agree, not sure, disagree, strongly disagree). A level of agreement score was achieved by assigning values +2.0 = 'Strongly Agree', +1.0 = 'Agree', 0.0 = 'Not Sure', -1.0 = 'Disagree', -2.0 = 'Strongly Disagree' to the Likert scale and computing the average. A value between +0.5 and +2.0 indicated agreement with the statement with a value above +1.0 indicating strong agreement. Corresponding negative values indicated disagreement and strong disagreement whilst

TABLE 3.2	: Design	of the	SWOT	template.
-----------	----------	--------	------	-----------

Internal Factors					
Strengths	Weaknesses				
Externa	l Factors				
Opportunities	Threats				

a value in the range -0.5 < Mean < +0.5 indicated a more neutral, indecisive or ambiguous stance.

3.7 Piloting the Questionnaire

This research study was aiming at obtaining responses from all CQMPs in Malta, which accounts to a maximum of 20 possible participants. To avoid further loss of participants due to the fact that any participant taking part in a pilot study would have had to be excluded from the main study, the questionnaire was not tested for content validity and reliability on the participants but by intensively consulting the principal supervisor and co-supervisor. The principal supervisor is a world authority on the use of strategic planning in MP, and the co-supervisor is the president of the MAMP. Both analysed the questionnaire several times, each time improving the content and simplifying the structure and language to increase clarity. As a double check, a short evaluation of the questionnaire was also added as a supplement to the main questionnaire. The evaluation sheet and a summary of the responses are shown in Appendix J.

3.8 Ethical Considerations

The research study was approved by the FREC of Health Sciences of the University of Malta. Since the questionnaire was totally anonymous the research proposal did not present any potential issues in the domain of research ethics and/or data protection, therefore data collection was authorised by FREC. FREC approval can be found in Appendix K.

3.9 Limitations of the Research Methodology

The main limitation in such questionnaires is usually the low response rate. This will be discussed at a later stage of the dissertation.

3.10 Conclusion

The study's methodology was discussed in this chapter. Chapter 4 presents the results obtained as well as a thorough discussion of the data.

Chapter 4

Results and Discussion

4.1 Introduction

This chapter provides the proposed vision statement for the MP profession in Malta. Data obtained from the semi-structured questionnaire is documented and discussed. This includes internal strengths and weaknesses and external opportunities and threats of the profession with respect to the vision statement from the perspective of the local MPs. This chapter ends by proposing a list of strategic objectives for achieving the proposed vision.

4.2 Vision Statement for the Medical Physics profession in Malta

Given the present work being done by the EFOMP Professional Matters committee (EFOMP, n.d) regarding the future role of MPs, it was decided to adopt the version of the vision statement as being suggested by the EFOMP professional matters committee at the time that the questionnaire was being finalised. The proposed vision statement being proposed at the time was:

"The Medical Physics profession will be recognised by all healthcare stakeholders as contributing to maintaining and improving the quality, safety and cost-effectiveness of healthcare services through patient-oriented activities requiring expert action, involvement or advice regarding the specification, selection, acceptance testing, commissioning, quality assurance, decommissioning and optimized clinical use of present and future medical devices and regarding patient risks from all associated physical agents including protection from such physical agents, installation design and surveillance, and the prevention of unintended or accidental exposures to physical agents; all activities will be based on current best evidence or own scientific research when the available evidence is not sufficient. The scope includes risks to volunteers in biomedical research and carers and comforters. Medical Physics Professionals will also take on at the expert level the added function of the protection of workers and the general public from physical agents (such as Radiation Protection Expert, Magnetic Resonance Safety Expert and Laser Safety Expert)." (Caruana C. J., personal communication, March 22, 2022).

4.3 Questionnaire Section 1: General Questions

Questions 1.1 to 1.3 provided some background information about the participants. The questionnaire garnered a total of eight responses out of the 20 clinical MPs in Malta resulting in a response rate of 40%, which is a normal response rate to such questionnaires. The areas of specialty of the MPs which took part in the survey were as follows: three from RO, three from D&IR, and two from NM. There were therefore a similar number of responses from all three specialty areas. The sample was therefore considered as sufficiently representative of the Maltese MP profession.

In Q1.4, the MPs were asked if they were certified as MPEs by the RPC, where eight (100%) replied as yes whilst none (0%) replied as no. Having such a high percentage of MPs certified as MPEs is a strength for the profession.

Questions 1.5 to 1.7 queried the participants regarding the following sections of the vision statement "Medical Physics Professionals will also take on at the expert level the added function of the protection of workers and the general public from physical agents (such as Radiation Protection Expert, Magnetic Resonance Safety Expert and Laser Safety Expert)." In Q1.5, the MPs were asked if they are certified also as RPEs by the RPC, where six (75%) replied as yes whilst two (25%) replied as no. This is also considered a strength since the majority are recognised also as RPEs. The two respondents that replied no in Q1.5 were from RO, but in fact it is common knowledge that there are many RPEs in RO. Q1.6 and Q1.7 asked if the participants were certified as MRSE and Laser Safety Expert (LSE) respectively by any accredited organisation to which all (100%) replied no for both questions. With regard to the vision statement this can be seen as a weakness since none of the MPs in Malta are recognised as a MRSE nor

as a laser safety expert. This is indeed a pity as courses for certification in these areas are easily available (the Institute of Physics and Engineering in Medicine (IPEM), UK holds such course regularly). MPs should be encouraged by MAMP to attend such courses.

Questions 1.8 to 1.16 queried the participants with regard to the following section of the vision statement "the specification, selection, acceptance testing, commissioning, quality assurance, decommissioning and optimized clinical use of present and future medical devices and regarding patient risks from all associated physical agents including protection from such physical agents, installation design and surveillance, and the prevention of unintended or accidental exposures to physical agents" with respect to the following device categories: ionising radiation based medical devices, non-ionising radiation based medical devices. The tables generated for each question show the observed frequencies for each Device Category by Specialty Area of the respondents.

In Q1.8, participants were asked for which categories of medical devices are MPs in Malta involved with respect to specification and procurement. The results are shown in Table 4.1. Out of the eight participants, seven (87.5%) answered that MPs in Malta are involved in the specification and procurement of ionising radiation based medical devices and five out of eight (62.5%) answered non-ionising radiation based medical devices. This means that MPs are quite strong in the specification and procurement of theses two specific categories of devices. However, only two out of eight (25%)answered that MPs are involved in the specification and procurement of all medical devices which is a weakness with respect to the vision statement. Overall, the perception is therefore that MPs in Malta are involved in the specification and procurement of ionising and non-ionising radiation based medical devices but very little outside these two traditional categories. Hence with respect to the vision statement which states that MPs should endeavour to involve themselves in all medical devices the position of the profession is still weak. This is expected since the profession is a very young profession in Malta. However, the profession should endeavour to expand its scope of practice to beyond radiation-based devices as recommended by the vision statement. In its meeting of August 2022 in Dublin the EFOMP Council approved a new Work Group to make proposals regarding the involvement of MPs in Physiological Measurement (Caruana C.J., personal communication, 5 September, 2022). MPs are already involved in this specialty in several countries in Europe (notable Finland and the UK). In addition, MPs are also involved in Audiology in Germany, Holland and Austria (Caruana et al.,

			Device Category				
			Ionising radiation based medical	Non-Ionising radiation based medical	All medical devices		
			devices	devices			
Specialty	Radiation Oncology	Count	2	1	2		
Area	Diagnostic and Interventional Radiology	Count	3	3	0		
	Nuclear Medicine	Count	2	1	0		
		Total	7	5	2		

TABLE 4.1: Crosstab for Specialty Area and Device Category of Q1.8.

2014). The MAMP should work with the University of Malta to introduce such areas in the curriculum. The author of this thesis has been advised that the University will be introducing Physiological Measurement in a revised MSc Medical Physics curriculum for October 2023 (Caruana C.J., personal communication, 5 September, 2022).

There are higher percentages of MPs from D&IR (100%) than MPs from RO and NM (33.3% and 50% respectively) who perceive that MPs are involved in the specification and procurement of non-ionising radiation based medical devices. This is probably because the MPs in D&IR are increasingly involving themselves in quality control of ultrasound and MRI. There are presently however moves to introduce an MR-LINAC and this would of necessity nudge MPs in RO to involve themselves more in non-ionising radiation devices. Meanwhile only two MPs from RO (66.6%) perceive that MPs are involved in the specification and procurement of all medical devices. This is probably due to the fact that MPs in RO also use a large number of ancillary medical devices (e.g., applicators, moulds and other patient restraint devices).

In Q1.9, participants were asked for which medical device categories are MPs in Malta involved in acceptance testing. The results are shown in Table 4.2. Out of the eight respondents, seven (87.5%) answered that MPs in Malta are involved in the acceptance testing of ionising radiation based medical devices and five out of eight (62.5%) answered non-ionising radiation based medical devices. This means that MPs are again quite strong in these two specific categories of devices. However, again only two out of eight (25%) answered that MPs are involved in the acceptance testing on all medical devices which is a weakness with respect to the vision statement. Overall, the perception

			Device Category				
			Ionising radiation based medical	Non-Ionising radiation based medical daviage	All medical devices		
	Radiation Oncology	Count	$\frac{uevices}{2}$	1	2		
Specialty Area	Diagnostic and Interventional Radiology	Count	3	3	0		
	Nuclear Medicine	Count	2	1	0		
		Total	7	5	2		

TABLE 4.2:	Crosstab f	for Specialty	Area and Device	Category of Q1.9
		1 2		

is therefore again that MPs in Malta are involved in the acceptance testing of ionising and non-ionising radiation based medical devices but very little outside these two categories. Hence with respect to the vision statement which states that MPs should endeavour to involve themselves in the acceptance testing of all medical devices, the position of the profession is still weak. The profession should strive to expand acceptance testing to all medical devices as discussed earlier.

There are again larger percentages of MPs from D&IR (100%) than MPs from RO and NM (33.3% and 50% respectively) who perceive that MPs are involved in the acceptance testing of non-ionising radiation based medical devices.

In Q1.10, participants were asked for which medical device categories are MPs in Malta involved in commissioning. The results are shown in Table 4.3. All of the participants (100%) answered that MPs in Malta are involved in the commissioning of ionising radiation based medical devices and six out of eight (75%) answered that they are involved in the commissioning of non-ionising radiation based medical devices. This means that MPs are again very strong in these two specific categories of devices. However, none of the participants indicated that they are involved in the commissioning of all medical devices which is a weakness with respect to the vision statement. This could be due to the fact that other healthcare professionals are doing the commissioning on these medical devices or that little is being done. Overall, the perception is therefore that MPs in Malta are involved in the commissioning of ionising and non-ionising radiation based medical devices only. Hence with respect to the vision statement, the MPs should endeavour to involve themselves in the commissioning of all medical devices are involved in the commissioning of all medical devices only.

			Device Category				
			Ionising radiation based medical	Non-Ionising radiation based medical	All medical devices		
			devices	devices			
Specialty	Radiation Oncology	Count	3	2	0		
Area	Diagnostic and Interventional Radiology	Count	3	3	0		
	Nuclear Medicine	Count	2	1	0		
		Total	8	6	0		

TABLE 4.3:	Crosstab for	[•] Specialty	Area and	Device	Category	of Q)1.1	10
							- · ·	

All the participants (100%) in the study indicated that MPs in each respective specialty area perceive that MPs in Malta are involved in the commissioning of ionising radiation based medical devices. On the other hand, there are larger percentages of MPs from RO and D&IR (66.7% and 100% respectively) than MPs from NM (50%) who perceive that MPs are involved in the commissioning of non-ionising radiation based medical devices. This can be explained by the fact that PET-MRI is not yet available in Malta.

In Q1.11, participants were asked which medical devices do MPs in Malta perform QA on. The results are shown in Table 4.4. Out of the eight respondents, seven (87.5%) answered that MPs in Malta are involved in the QA of ionising radiation based medical devices and five out of eight (62.5%) answered non-ionising radiation based medical devices. Hence, meaning that MPs are again quite strong in these two specific categories of devices. However, only one of the participants (12.5%) answered that MPs are involved in the acceptance testing of all medical devices which is a weakness with respect to the vision statement. Overall, the perception is therefore that MPs in Malta are again mostly involved in the QA of ionising and non-ionising radiation based medical devices. With respect to the vision statement, which states that MPs should endeavour to involve themselves in the QA of all medical devices, the position of the profession is still weak. The profession should work to increase its involvement in the QA of all medical devices.

There are larger percentages of MPs from D&IR and NM (both 100%) than MPs from RO (66.7%) who perceive that MPs are involved in the QA of ionising radiation based medical devices. This is probably because the MPs in D&IR and NM involve themselves

			Device Category				
			Ionising radiation based medical	Non-Ionising radiation based medical	All medical devices		
			devices	devices			
Specialty	Radiation Oncology	Count	2	1	1		
Area	Diagnostic and Interventional Radiology	Count	3	3	0		
	Nuclear Medicine	Count	2	1	0		
		Total	7	5	1		

TABLE 4.4: Crosstab for Specialty Area and Device Category of Q1.11.

extensively in QA. It might also be because QA in RO might also involve the manufacturer to a higher extent given that the profession is still young in Malta and therefore the level of experience in RO is less than desirable.

In Q1.12, participants were asked for which medical device categories are MPs in Malta involved in decommissioning. The results are shown in Table 4.5. Out of the eight respondents, four (50%) answered that MPs in Malta are involved in the decommissioning of ionising radiation based medical devices, two (25%) answered non-ionising radiation based medical devices, one (12.5%) answered all medical devices, and three (37.5%) answered that they do not partake in the decommissioning of any medical devices. Overall, the perception is therefore that MPs in Malta are barely involved in the decommissioning process with only 50% stating that they are involved with regards to ionising radiation medical devices, whilst 37.5% do not perform this duty on any medical device. However, one must remark that the process of medical device decommissioning is still a grey area in Malta (Caruana C. J., personal communication, September 5, 2022). The 37.5%of the participants that indicated that they do not perform decommissioning on any medical devices correspond to MPs in D&IR. However the duties of a MP in D&IR, according to the Ministry for Health in Malta do include that they should contribute to the decommissioning of medical radiological devices as shown in Subsection 2.4.5 under (f). It could be that MPs are not performing such duties due to other more pressing work. Hence with respect to the vision statement which states that MPs should endeavour to involve themselves in the decommissioning of all medical devices, the position of the profession is very weak. As indicated by the vision statement, MPs should strive to be part of the decommissioning of all medical devices.

				Device Car	tegory	
			Ionising radiation based medical	Non-Ionising radiation based medical devices	All medical devices	None of the above
Specialty	Radiation Oncology	Count	2	1	1	0
Area	Diagnostic and Interventional Radiology	Count	0	0	0	3
	Nuclear Medicine	Count	2	1	0	0
		Total	4	2	1	3

TABLE 4.5: Crosstab for Specialty Area and Device Category of Q1.12.

In Q1.13, participants were asked which medical devices do they optimise. The results are shown in Table 4.6. Out of the eight respondents, seven (87.5%) answered that MPs in Malta are involved in optimising the clinical use of ionising radiation based medical devices and five out of eight (62.5%) answered non-ionising radiation based medical devices so again MPs are quite strong in these two specific categories of devices. However, only one of the eight participants (12.5%) answered that MPs optimise the clinical use of all medical devices. With respect to the vision statement this is considered as a weakness. Overall, the perception is therefore that MPs in Malta are mostly involved in optimising the clinical use of ionising and non-ionising radiation based medical devices. With respect to the vision statement, which states that MPs should endeavour to involve themselves in optimisation of the clinical use of all medical devices, the position of the profession is still weak. As advised by the EFOMP vision statement, MPs should involve themselves in optimising the clinical use of all medical devices.

In Q1.14, participants were asked if MPs in Malta provide advice regarding patient risks from medical device associated physical agents including protection from such physical agents for the following categories of physical agents: ionising radiation, nonionising radiation, and all physical agents (vision statement: "*patient risks from all associated physical agents including protection from such physical agents, installation design and surveillance, and the prevention of unintended or accidental exposures to physical agents*"). The results are shown in Table 4.7. Out of the eight respondents, all (100%) answered that MPs in Malta provide advice in cases of ionising radiation and four out of eight (50%) answered that advice is provided in cases of non-ionising radiation. Hence, meaning that MPs are again quite strong in these two specific categories of devices. However, none of the MPs (0%) provide advice regarding patient risks

			Device Category				
			Ionising radiation based medical	Non-Ionising radiation based medical	All medical devices		
			devices	devices			
Specialty	Radiation Oncology	Count	2	1	1		
Area	Diagnostic and Interventional Radiology	Count	3	3	0		
	Nuclear Medicine	Count	2	1	0		
		Total	7	5	1		

TABLE 4.7: Crosstab for Specialty Area and Device Category of Q1.14.

			Device Category				
			Ionising radiation based medical devices	Non-Ionising radiation based medical devices	Physical agents		
Snecialty	Radiation Oncology	Count	3	1	0		
Area	Diagnostic and Interventional Radiology	Count	3	2	0		
	Nuclear Medicine	Count	2	1	0		
		Total	8	4	0		

and protection from all physical agents which is a weakness with respect to the vision statement. This indicates that the general notion of physical agents is not well established here but just ionising and non-ionising radiation and in the case of the latter this is limited to only MRI and Ultrasound (US) – but not lasers, ultraviolet, infra-red, visible. As advised by the EFOMP vision statement, the profession should work to increase their knowledge and skills with regards to patient risks and protection from all physical agents.

There is a very high percentage of MPs from RO, D&IR, and NM (all 100%) who perceive that MPs in Malta are involved in providing advice regarding patient risks from ionising radiation. On the other hand, there are larger percentages of MPs from D&IR (66.7%) than MPs from RO and NM (33.3% and 50% respectively) who perceive involvement in patient risk from non-ionising radiation. This is almost certainly because

of the involvement of D&IR physicists in MRI and US imaging and the fact that in Malta there is not an MR-LINAC or PET-MRI.

In Q1.15, participants were asked for which medical device categories are MPs in Malta involved in the design and installation of the new facilities. The results are shown in Table 4.8. All of the participants (100%) answered that MPs in Malta are involved in the design and installation of ionising radiation based medical devices and five out of eight (62.5%) answered that they are involved in the design and installation of non-ionising radiation based medical devices and five out of eight (62.5%) answered that they are involved in the design and installation of non-ionising radiation based medical devices. This means that MPs are again very strong in these two specific categories of devices. However, none of the participants indicated that they are involved in the design and installation of all medical devices which is a weakness with respect to the vision statement. This could be attributed to the fact that the MP team in Malta is very small and there is simply not enough physicists to cater for installation and surveillance of all medical devices. Overall, the perception is therefore that MPs in Malta are involved in the design and installation of ionising and non-ionising radiation based medical devices. However, with respect to the vision statement which states that MPs should endeavour to involve themselves in the design and installation of all medical devices, the position of the profession is still weak.

In Q1.16, participants were asked for which of the medical devices do MPs in Malta provide advice regarding the prevention of unintended or accidental exposures to physical agents. The results are shown in Table 4.9. All of the participants (100%) answered that MPs in Malta provide advice regarding the prevention of unintended or accidental exposures of ionising radiation based medical devices. However, three out of eight

				Device Cate	gory
			Ionising radiation based medical	Non-Ionising radiation based medical	All medical devices
			devices	devices	
Specialty	Radiation Oncology	Count	3	2	0
Area	Diagnostic and Interventional Radiology	Count	3	2	0
	Nuclear Medicine	Count	2	1	0
		Total	8	5	0

TABLE 4.8: Cross	stab for Specialty	Area and Device	Category of Q	1.15.
------------------	--------------------	-----------------	---------------	-------

				Device Cate	gory
			Ionising radiation based medical devices	Non-Ionising radiation based medical devices	All medical devices
Specialty	Radiation Oncology	Count	3	2	0
Area	Diagnostic and Interventional Radiology	Count	3	0	0
	Nuclear Medicine	Count	2	1	0
		Total	8	3	0

TABLE 4.9: Crosstab for Specialty Area and Device Category of Q1.16.

(37.5%) answered that they provide advice regarding the prevention of unintended or accidental exposures of non-ionising radiation based medical devices.

Questions 1.17 to 1.19 queried the participants with regard to the following section of the vision statement "*The Medical Physics profession will be recognised by all healthcare stakeholders as contributing to maintaining and improving the quality, safety and cost-effectiveness of healthcare services through patient-oriented activities requiring expert action*". The responses were analysed by attaching a categorical Likert scale with +2.0 = 'Strongly Agree', +1.0 = 'Agree', 0.0 = 'Not Sure', -1.0 = 'Disagree', -2.0 = 'Strongly Disagree' and a mean computed. A value between +0.5 and +2.0 indicated agreement with the statement with a value above +1.0 indicating strong agreement. Corresponding negative values indicated disagreement and strong disagreement whilst a value in the range -0.5 < Mean < +0.5 indicated a more neutral, indecisive or ambiguous stance. Table 4.10 presents a statistical summary of the results where *N* refers to the number of participants and *Mean* is the average score of the responses.

In Q1.17, participants were asked if the MP profession in Malta is seen by stakeholders as contributing significantly to the over-arching mission of the wider healthcare system of which it forms part. Results show that on average the MPs generally agree (mean Likert score = +0.75) with this statement which is a strength with respect to the vision statement.

In Q1.18, participants were asked if the MPs in Malta are aware of what is internationally considered best practice in their specialty area. Results show that the MPs strongly agree (mean Likert score = +1.50) that they are well informed of the procedures and techniques

	N	Mean	Strongly Agree %	Agree %	Not Sure %	Disagree %	Strongly Disagree %
Q1.17	8	+0.75	12.5	50	37.5	0	0
Q1.18	8	+1.50	25	75	0	0	0
Q1.19	8	0	12.5	25	25	25	12.5

TABLE 4.10: Statistical summary of the results obtained for Q1.17 till Q1.19.

in their corresponding specialty area. This can be considered as a core competence with respect to the vision statement.

In Q1.19, the participants were asked to what extent they agree with the statement that "There has never been a shortage of material resources that are required to enable me to deliver my professional services to the healthcare system". The results were neutral (mean Likert score = 0), with three out of eight (37.5%) participants disagreeing and two out of eight (25%) not being sure with this statement which is worrying because this means that MPs may be suffering from a shortage of the resources necessary for them to deliver an effective and safe service, which can be considered as a weakness of the MP profession in Malta with respect to the vision statement.

In Q1.20 when asked what specific unique benefits the MP profession in Malta provides and to whom (patients, other healthcare professionals, hospital management), the respondents generally agreed that the MP profession provides safe and accurate MP services to patients, other healthcare staff, and the general public. It was also identified that the MP profession provides education to students from various professions. The following are some quotes by the respondents.

"Medical Physics education to students, patient safety advice to healthcare staff, diagnostic systems are up-to-date due to QA testing."

"In Malta, in Nuclear Medicine, the physicists have significant contact with patients, both for radiation protection and therapy aspects."

"Since we are a small country and most clinical physicist are based in the single largest hospital, our impact can reach hundreds of thousands of patients every year."

One respondent noted the integrity and scientific excellence that each specialty strives to provide to the hospital management:

"In-depth knowledge in the specialty area helps hospital management better understand certain requirements for improvement with respect to patient services, provide guidance to hospital management when certain goals are to be reached, and thus help advance patient clinical services in both diagnostic and therapeutic areas."

It is clear from the responses that there is a wide range of people that benefit from services the MP profession in Malta provides which is a strength with respect to the vision statement.

4.4 Questionnaire Section 2: Possible Internal Strengths of the Medical Physics profession in Malta

Questions 2.1 till 2.8 were analysed by attaching the previous Likert scale and their mean score was obtained from the responses. Table 4.11 presents a statistical summary of the results.

In Q2.1 MPs were asked to what extent they agree with the statement that "Medical Physicists in Malta have a deep techno-scientific expertise regarding medical devices used in their specialty and their clinical use." The MPs agree that this is a strength of the profession with 37.5% saying they strongly agree (mean Likert score = +1.25). With respect to the vision statement a MP should have a deep understanding of medical devices used in their specialty area. MP is the only profession in healthcare with a strong understanding of the technology and optimised clinical application.

In Q2.2, participants were asked to what extent they agree with the statement that "Medical Physicists in Malta have a deep techno-scientific expertise regarding the protection of patients, workers and the general public from ionising radiation and other physical agents used in their specialty." The respondents strongly agree with this statement (mean Likert score = +1.25). MPs in Malta consider that they are highly capable of handling such agents which is a strength with respect to the vision statement. Physical agents provide considerable medical benefits, and their usage is steadily rising. However, there is a risk component as well, and when employed by individuals who lack appropriate understanding, they constitute a source of unnecessary risk.

In Q2.3 participants were asked to what extent they agree with the statement that "Medical Physicists in Malta have strong analytical, problem-solving, and trouble-shooting

	N	Mean	Strongly Agree %	Agree %	Not Sure %	Disagree %	Strongly Disagree %
Q2.1	8	+1.25	37.5	50	12.5	0	0
Q2.2	8	+1.25	37.5	50	12.5	0	0
Q2.3	8	+1.25	25	75	0	0	0
Q2.4	8	+1.13	12.5	87.5	0	0	0
Q2.5	8	+0.75	0	75	25	0	0
Q2.6	8	+1.25	25	75	0	0	0
Q2.7	8	+1.00	25	50	25	0	0
Q2.8	8	+0.50	0	62.5	25	12.5	0

TABLE 4.11: Statistical summary of the results obtained for Q2.1 till Q2.8.

skills." The survey showed that the respondents strongly agree with this statement (mean Likert score = +1.25). MP services require strong problem solving skills. The requisite that MPs possess a first degree with a strong physics and mathematics component guarantees that only those of particularly high analytical skills are allowed to enter the profession.

In Q2.4 participants were asked to what extent they agree with the statement that "*Medical Physicists in Malta have strong mathematical, statistical and data analysis skills.*" MPs in Malta strongly agree that they have such skills (mean Likert score = +1.13). MPs lead over all other healthcare specialties in terms of quantitative data analysis, statistics, and mathematics. Quantitative imaging, AI, and machine learning are seeing a surge in therapeutic applications, opening up yet another higher tier dimension. Courses in MP now include deep learning.

In Q2.5 participants were asked to what extent they agree with the statement that "*The Medical Physics profession in Malta has a strong legal foundation.*" From the responses obtained, there is a general consensus, though not very strong, that MPs in Malta do have a strong legal foundation (mean Likert score = +0.75). The Maltese authority issued legal Act (L.N.210, 2018) stating MPEs are qualified to provide specialist advice on radiation physics related matters. The MP profession is a legally established health care profession and regulated via the CPCM. Having a profession legally recognised by the local authority is considered a strength with respect to the vision statement.

In Q2.6 participants were asked to what extent they agree with the statement that "Medical Physicists in Malta have strong scientific research skills." There is a strong consensus that MPs have strong scientific research skills (mean Likert score = +1.25). MPs have unmatched scientific quantitative research abilities that have been perfected through many years of training. Over the course of their lengthy study and training, MPs are taught concepts including accuracy, reliability, and estimations of uncertainty. Since MPs often have to introduce new technologies in healthcare and since often protocols are not sufficiently optimised, they need to have research skills to drive the use of the technology forward. With the vision statement stating that "*all activities will be based on current best evidence or own scientific research when the available evidence is not sufficient*", this is a critical strength.

In Q2.7 participants were asked to what extent they agree with the statement that "*Medical Physicists in Malta have strong ICT skills*." On average the results showed that the respondents agree that they possess such a strength (mean Likert score = +1.00). Those who answered not sure are probably not aware of the skills of some of their colleagues. MPs in Malta possess extremely high ICT skill levels, particularly programming and this is considered as a strength with respect to the vision statement.

In Q2.8 participants were asked to what extent they agree with the statement that "*Present Medical Physics services in Malta compare well with other EU countries*." The MPs agree, though not strongly, that services offered locally are similar to those in other EU regions (mean Likert score = +0.50).

In Q2.9 the participants were asked whether they believe that the qualification and curriculum framework leading to the certification of MP professionals in Malta was sufficient to ensure the production of high calibre professionals. Where five replied as yes (62.5%) whilst three replied as no (37.5%). When asked in Q2.10, which element of the curriculum can be improved, the respondents generally agreed that part of the clinical training can be enhanced by doing some training at foreign hospitals such that exposure to MP services in other countries is obtained to widen the vision. One respondent proposed that the qualification to apply for a MSc in medical physics should also be changed from "Bachelor in Physics/Engineering to Bachelor in Physics, Medical Physics and Radiation Protection." Although the latter would certainly lead to even more competent professionals, it must be kept in mind that a first degree in physics/engineering is recommended by the 'European Guidelines on the Medical Physics' project (EU Commission, 2014) and EFOMP Policy Statement 12.1 (Caruana et al., 2014). Hence the present Bachelor in 'Physics, Medical Physics and Radiation Protection' may be

considered as an ideal compromise (Caruana C.J., personal communication, September 5, 2022).

In Q2.11 the respondents were given the opportunity to highlight any other notable strengths for the MP profession in Malta. The MPs identified the determination, competitivity and drive for improvement with the profession being quite a young team with modern ideas and a very new profession in Malta with many opportunities where to advance. One of the respondents identified that Malta as a small country made it easier for the profession to adapt to new changes:

"Since we are a small country, we can be highly cohesive and adaptable. For example, in Germany things are a bit fragmented because of the many states that are involved, and professionals would be hesitant to comment or commit without approval from all states."

4.5 Questionnaire Section 3: Possible Internal Weaknesses of the Medical Physics profession in Malta

Questions 3.1 to 3.10 were of the ordinal categorical scale and a Likert scale mean score was obtained as previously. The following are the questions and their respective result. Table 4.12 presents a statistical summary of the results.

In Q3.1 participants were asked to what extent they agree with the statement that "One of the problems that the profession faces in Malta is the absence of a universally known mission statement for the profession for Malta." The majority of the MPs are not sure whether this is a weakness of the profession (mean Likert score = +0.38). Lack of a mission statement that is brief, unambiguous, universally recognised, and most importantly, easily marketable is one of the issues the profession could be currently dealing with. A mission statement would serve as a guide for the MPs and should be quickly implemented to show the importance of the role to healthcare management and even society at large. This can be considered as a weakness, according to the vision statement the MPs should be recognised by all healthcare stakeholders, and this cannot be achieved without a known mission statement.

In Q3.2 participants were asked to what extent they agree with the statement that *The Medical Physics profession in Malta has a narrow range of specialisations.*" On average, the respondents are quite neutral about this statement (mean Likert score = -0.38). The

	N	Mean	Strongly Agree %	Agree %	Not Sure %	Disagree %	Strongly Disagree %
Q3.1	8	+0.38	12.5	12.5	75	0	0
Q3.2	8	-0.38	12.5	25	37.5	25	0
Q3.3	8	-0.13	12.5	12.5	37.5	25	12.5
Q3.4	8	+0.25	12.5	25	50	0	12.5
Q3.5	8	+0.50	0	75	12.5	0	12.5
Q3.6	8	-0.63	12.5	0	25	37.5	25
Q3.7	8	0	12.5	25	25	25	12.5
Q3.8	8	-0.25	0	25	25	50	0
Q3.9	8	+0.75	12.5	62.5	12.5	12.5	0
Q3.10	8	+0.63	25	37.5	12.5	25	0

TABLE 4.12: Statistical summary of the results obtained for Q3.1 till Q3.10.

use of ionising radiation in D&IR, NM, and RO remains the primary clinical MP specialty in Malta. The focus of the profession should be broadened to include all physical agents and medical equipment as indicated by the vision statement so it can be considered as a weakness if MPs do not specialise in all physical agents.

In Q3.3 participants were asked to what extent they agree with the statement that "*The Medical Physics profession in Malta still has not realised the importance of strategic and robust leadership.*" Results showed that on average the MPs are quite neutral about this statement (mean Likert score = -0.13). Strategic leadership is important as it allows for the desired vision to be developed and achieved. Perhaps there needs to be a debate on the development of strategic leadership for the profession in Malta.

Q3.4 stated that "Medical Physicists in Malta have low marketing skills leading to too low profile of the profession within healthcare." Results show that the participants are not sure whether as a profession they have low marketing skills (mean Likert score = +0.25). One respondent that strongly disagreed stated that:

"Medical physicists have gone a long way to prove the importance of the profession in healthcare."

However, results show that many MPs in Malta might struggle with the idea of promoting and advertising their services. This is a weakness with respect to the vision statement because the profession should be recognised by all healthcare stakeholders as being a profession that improves the quality, safety, and cost-effectiveness of healthcare services. One respondent however stated that this issue is being targeted:

"We are trying to improve this by being actively involved in Allied healthcare matters where we interact with key individuals from many different professions."

In Q3.5 participants were asked to what extent they agree with the statement that "Medical physicists in Malta take part in regional/international networking." The MPs agree at a very basic level with this statement (mean Likert score = +0.50). Regional and international networking is an integral way to strengthen the profession's professional and political position. With networking, the profession would achieve a more harmonised perception of itself and its role in the healthcare world today.

In Q3.6 participants were asked to what extent they agree with the statement that "*The lack of an independent MP department will negatively impact the achievement of the vision.*" Results show that the participants disagree (mean Likert score = -0.63). The options for the profession to develop and perform strategic planning depend on where the MP group is located within the organisational structure of healthcare management. One respondent even argued that:

"There are of course many benefits to having an independent MP department. This can only be realised once we have enough staff and a sufficient number of PhDs."

However, it appears that the MPs in Malta currently do not consider it to be an important issue. One respondent even argued that:

"In terms of SL 585.01, having a MP department will not change the current work being done within healthcare."

This suggests that there is a significant amount of mutual assistance and cross-fertilisation of ideas between the three specialties thus the three specialties do not feel isolated from each other. However not having an independent department does mean that there is little direct communication with the CEO of the hospital, and everything needs the approval of the respective medical chair.

In Q3.7 participants were asked to what extent they agree with the statement that "*There is a reluctance by some Medical Physicists in Malta to be part of the wider healthcare*

picture." On average, the respondents are not sure whether their peers are reluctant to be part of the wider healthcare picture (mean Likert score = 0). Some MPs could tend to believe that physics is the only thing that matters and that they do not need to worry about other disciplines because they do not consider them as vital. However, inter-professional teamwork and cooperation is the future. Overall, with respect to the vision statement this can be seen as a weakness.

In Q3.8 participants were asked to what extent they agree with the statement that "Medical Physicists in Malta have insufficient communication and pedagogical skills." The MPs are neutral on this issue (mean Likert score = -0.25). MPs should have sufficient communication skills, as this would help the profession market its services and its vision to other professions and management.

In Q3.9 participants were asked to what extent they agree with the statement that "*Research by Medical Physicists in Malta has barely taken off.*" The MPs in Malta generally agree with the idea that research by MPs is still far behind (mean Likert score = +0.75). Research has barely taken off in Malta due to either other work/personal commitments or else lack of motivation. This is considered as a weakness with respect to the vision statement.

In Q3.10 participants were asked to what extent they agree with the statement that "Our work duties in Malta are too overwhelming to be able to have enough free time to enrol in new training courses or attend conferences regarding the profession." The MPs generally agree with the idea that their work duties are too much and to the extent that they have no free time to further increase their knowledge and skills in the profession (mean Likert score = +0.63). One of the respondents stated:

"We are essentially a start-up operating within a larger government organisation. There is foundational work to be completed and more staff to be recruited before we can reach a steady working state to be able to take on higher level duties."

This is considered as a weakness for the profession with respect to the vision statement since the MP team in Malta will not have sufficient time to involve itself in new training courses and seminars.

As a general note, most of the possible weaknesses presented in the questionnaire obtained, on average, a neutral result from the participants as shown in Table 4.12. More research needs to be done to unravel the actual situation.

In Q3.11 when asked whether there are any legal problems that might hinder the idea of the profession from achieving this vision, three replied as yes (37.5%) whilst five replied as no (62.5%). Some issues were raised with regard to the involvement of a MP when using some modalities:

"Currently the MP/MPE is not legally required for acceptance testing of radiological medical equipment. Involvement of the MPE determined by the modality, leaving the lower dose-per-patient modalities with lacking MPE oversight (these modalities however have a very high dose to the population due to the high number of imaged patients). Additionally, there is currently no enforcement for MPE involvement in dental and cone-beam computed tomography (CBCT) systems."

With respect to the vision statement this is a weakness for the profession since it states that MPs should be involved in the acceptance testing of all medical devices, regardless of area of specialty. In fact it is strictly not in conformance with EU directive 2013/59/EURATOM (2013).

However, the majority believe that there are no legal problems that can hinder the profession from achieving this part of the vision and this issue regarding acceptance testing will eventually resolve itself when there are more MPs once the bachelor's students graduate.

In Q3.13 and Q3.14 the respondents were given the opportunity to highlight any other notable weaknesses for the MP profession in Malta and explain how they could be eliminated. Some MPs considered that there is a lack of engagement by MPs in the development of the profession, and this can be eliminated with good leadership. Clinical work is overwhelming, leaving no time for independent research to be done and that there is a lack of sufficient MPs in Malta. They stated that such issues can be solved by employing more MP staff, by having a better structure in the department such as introducing the idea of dosimetrists in radiotherapy and more automated software to ease the work of the MPs. Also, one respondent raised the issue of expanding the knowledge and skills which depends on work done in own free time:

"Maintenance and growth in the profession is difficult for most as since we are again quite young, there are many 'big life projects' such as housing, marriage, children that impact the amount of free time that could otherwise be dedicated to extracurricular work such as involvement in professional societies, development of high quality presentations and courses to keep a high standard."

4.6 Questionnaire Section 4: Possible External Opportunities for the Medical Physics profession in Malta

Questions 4.1 till 4.11 were also based on an ordinal categorical Likert scale. Table 4.13 presents a statistical summary of the results.

In Q4.1 participants were asked to what extent they agree with the statement that "*The Medical Physics profession can expand as a result of the increase in number and sophistication of hospital medical devices.*" The MPs in Malta strongly agree that the profession can grow if there is investment in hospital medical devices (mean Likert score = +1.13). MPs must make the most of the opportunities presented by the fast increase in the variety and sophistication of medical equipment. One respondent even stated that: "If we are proactive, we can be at the forefront for adoption of these technologies." Any other medical or healthcare profession falls well short of the amount of competence that the MP profession possesses in this field. Though advances in AI and theranostics are significant, relatively few members of the healthcare industry are familiar with these words, let alone be qualified to optimise their usage or conduct acceptance testing and quality control. This is seen as an opportunity with respect to the vision statement. The utilisaion of lasers, biomedical optics, and nanotechnology are all rapidly developing, and the profession should carefully consider increasing its participation in these fields.

In Q4.2 participants were asked to what extent they agree with the statement that "Medical Physicists in Malta should link up with patient organisations and project themselves as patient advocates regarding safety standards in healthcare." The MPs are not sure whether they should link up with patient organisations as patient advocates regarding safety standards in healthcare (mean Likert score = +0.25). This could be due to the overwhelming work duties already at hand.

In Q4.3 participants were asked to what extent they agree with the statement that "Medical Physicists in Malta should link up with occupational health and safety organisations

	N	Mean	Strongly Agree %	Agree %	Not Sure %	Disagree %	Strongly Disagree %
Q4.1	8	+1.13	25	62.5	12.5	0	0
Q4.2	8	+0.25	12.5	25	37.5	25	0
Q4.3	8	+0.13	12.5	25	25	37.5	0
Q4.4	8	+1.25	50	37.5	0	12.5	0
Q4.5	8	+1.00	50	25	12.5	0	12.5
Q4.6	8	+0.50	25	37.5	12.5	12.5	12.5
Q4.7	8	0	0	37.5	25	37.5	0
Q4.8	8	+0.38	12.5	50	0	37.5	0
Q4.9	8	+1.13	37.5	50	0	12.5	0
Q4.10	8	+1.50	50	50	0	0	0
Q4.11	8	+1.63	62.5	37.5	0	0	0

TABLE 4.13: Statistical summary of the results obtained for Q4.1 till Q4.11.

and environmental groups and project themselves as occupational/public advocates with regard to safety from physical agents." The MPs are neutral about this statement (mean Likert score = +0.13). This could be because currently occupational/public safety falls under the duties of a RPE. However, this is a great opportunity for MPs that are interested in expanding their role since it is legally required that healthcare professionals' education and training curriculum include competences regarding protection against physical agents. With respect to the vision statement this approach can be seen as an opportunity as it mentions that a MP should take the added function of the role of a RPE.

In Q4.4 participants were asked to what extent they agree with the statement that "*The Medical Physics profession should create further opportunities for itself by exerting pressure on legislators to recognise that patient safety should not be limited to ionising radiation and strive to ensure that the rigorous standards of safety to be found in ionising radiation are extended to other areas such as magnetic resonance imaging, lasers and in the future nanodevices.*" There was a strong agreement between the participants that the legislators should recognise that patient safety is not limited to ionising radiation only but is extended to areas such as magnetic resonance imaging, lasers and nanodevices (mean Likert score = +1.25). Such legislations would further strengthen the legal basis of the profession. For instance, the EU has implemented laws for staff safety related to electromagnetic radiation, but there is no specific legislation for patients as there is for ionising radiation. This is a chance for MPs to raise awareness of these challenges among the broader public, pushing the field into the public eye and increasing its visibility. The vision statement also mentions that a MP should take on at expert level the added

function of protection from workers and the general public from physical agents (e.g., MRSE and LSE) so this would be an opportunity for the profession to expand its role.

In Q4.5 participants were asked to what extent they agree with the statement that "Due to the escalating cost of healthcare, the need for efficient use of medical devices and Health Technology Assessment is vital, and we as Medical Physicists need to take this opportunity and involve ourselves in such issues, as it would raise our profile with health economists." Results show that participants agree with this statement (mean Likert score = +1.00). This is seen as an opportunity with respect to the vision statement which indicates that MPs should provide cost-effective healthcare services. Health economics, which is the practise of utilising limited resources in the best way, is very important as healthcare costs are on the rise whilst patient expectations rise as well. This is an opportunity for MPs to be involved in clinical trials and health technology assessment. One respondent noted:

"Involvement in clinical patient pathways to emphasise that examinations are really justified."

In Q4.6 participants were asked to what extent they agree with the statement that "The rapid expansion in the number of home-use, self-testing and wearable devices are not only a clinical opportunity for Medical Physicists but also can be turned into a business opportunity. Shops selling such devices should be manned by Medical Physicists who can give advice to customers on the most appropriate devices to purchase, proper and safe use, re-calibration frequencies and other information. Medical Physicists can also offer quality control services for such devices." The MPs generally agreed, though not strongly, that self-testing and wearable devices can be turned into a business opportunity (mean Likert score = +0.50). One respondent who strongly disagreed argued that this should be the role of a biomedical engineer. A medical gadget designed for users at home is one that is used outside of a setting where professionals provide healthcare, these devices can be used to gather information for tracking a user's health, example heart rate. According to the vision statement this would be a great opportunity for the profession as MPs can provide QC services to such equipment in addition to offering recommendations on which gadget should be bought.

In Q4.7 participants were asked to what extent they agree with the statement that "*The Government of Malta makes available a lot of funding for me to able to develop my*

competences." The participants have come to a consensus that they are not sure whether such fundings are available (mean Likert score = 0). This could be since very few to none are actively involved in research hence the government would not receive requests for funding from MPs leading to such grants being used somewhere else.

Q4.8 stated, "I take part in IPEM courses." The majority of the participants do take part in IPEM courses with five respondents (62.5%) agreeing on taking such courses whilst three respondents (37.5%) do not. Q4.9 stated, "I take part in EFOMP school for medical physics experts." All (87.5%) except one (12.5%) of the respondents take part in EFOMP courses. Q4.10 stated, "I take part in IAEA courses." All the respondents (100%) partake in IAEA courses. Q4.11 stated, "I take the opportunity to go to conferences regarding the profession." All the respondents (100%) attend conferences regarding the MP profession. Taking part in these courses organised by these organisations is a great opportunity for the profession in Malta. The vision statement encourages MPs to take any opportunity that allows them to expand their scope of practice and these courses can help the profession do just that.

In Q4.12 the MPs were asked whether there are opportunities for research in their specialty area. Six replied as yes (75%) whilst two replied as no (25%). The respondents that replied with a no are both from RO. However, one of the participant that replied with a yes is also from RO. This could indicate that there are opportunities in all specialty areas however some MPs might not be sufficiently informed. It also indicates that there is more push for research in D&IR and NM. The MPs in Malta fully agree that they have strong scientific research skills as suggested by the results obtained in Section 4.4, Table 4.11, Q2.6, so these abilities should not be left unattended.

In Q4.13 and Q4.14 the respondents were asked to highlight any other notable opportunities for the MP profession in Malta and explain how the profession would benefit. The respondents agreed they should get more involved with foreign organisations such as EFOMP and IAEA and the knowledge gained can be used in Malta as well as participating in public relations activities to show the public the importance of having MPs. Lecturing and project supervision was also mentioned such that there would be an increase in awareness and knowledge within the university. One respondent also mentioned the importance of being part of the RPC: "Having a say within the radiation safety authority as a stakeholder in deciding certain standards and legal notices."

4.7 Questionnaire Section 5: Possible External Threats to the Medical Physics profession in Malta

Questions 5.1 till 5.5 were of the ordinal categorical Likert scale and their mean score was obtained from the responses. Table 4.14 presents a statistical summary of the results.

In Q5.1 participants were asked to what extent they agree with the statement that "The BSc(Hons) Physics, Medical Physics and Radiation Protection has solved the threat of low number of medical physicists due to low number of Physics/Engineering graduates." The majority of the MPs seem to agree that by introducing a BSc in Medical Physics solves the threat of low physics/engineering graduates (mean Likert score = +0.75). When there are not enough physics/engineering students, it is critical that undergraduate students are found without relying on other sources. A consistent supply of physics/engineering graduates is essential for the profession. With more members, the MP profession in Malta will be able to be more involved in the clinical use of present and future medical devices and physical agents which is a strength with respect to the vision.

In Q5.2 participants were asked to what extent they agree with the statement "*Malta* needs more Medical Physicists to achieve western European standards in my specialty area." The MPs agree that an increase in staff is required for a specialty area to prosper (mean Likert score = +1.00). A lack of MPs is a threat as it would not allow the profession to achieve what is being proposed by the vision statement.

In Q5.3 participants were asked to what extent they agree with the statement "Austerity economics and unrestrained commoditisation, are a threat to the Medical Physics profession in Malta." The respondents agree (mean Likert score = +0.75). Austerity would lead to a very low yearly intake of MPs, with the profession looking at a small rate of growth which would be a threat because the workload is steadily increasing but staffing levels are not. The political leadership of the profession has actively countered the threat of commoditisation.

In Q5.4 participants were asked to what extent they agree with the statement "*There is role poaching from other professions in Malta*." The participants have mixed views on

	N	Mean	Strongly Agree %	Agree %	Not Sure %	Disagree %	Strongly Disagree %
Q5.1	8	+0.75	37.5	25	25	0	12.5
Q5.2	8	+1.00	12.5	75	12.5	0	0
Q5.3	8	+0.75	25	25	50	0	0
Q5.4	8	+0.50	25	37.5	12.5	12.5	12.5
Q5.5	8	+0.38	0	62.5	12.5	25	0

TABLE 4.14: Statistical summary of the results obtained for Q5.1 till Q5.5.

this. In fact, 25% strongly agree, 37.5% agree, 12.5% are not sure, 12.5% disagree, and 12.5% strongly disagree (mean Likert score = +0.50). Role-poaching can be very harmful to the profession and inter-professional relations. One of the respondents even argued that when some MPs only do routine constancy testing, people from other professions might get the idea that they are able to perform such duties: "We have for many years only really been performing machine QC. This has given the impression that our job can be done by other healthcare professionals as well as some machine technicians." With respect to the vision this is a threat to the quality of services that should be offered by MPs.

In Q5.5 participants were asked to what extent they agree with the statement "*The profession is facing unfair competition from other professions*." Again, there are mixed views on this issue. Having 62.5% agree, 12.5% are not sure, and 25% disagree (mean Likert score = +0.38). The respondents have stated that other professions such as technicians have been doing some MP work. One of the respondents argued that they are faced with this problem because they are short staffed meaning that they are not able to be present at all times to intercept such incidents and do the work themselves.

In Q5.6 and Q5.7 the respondents were asked to highlight any other notable threats that the MP profession in Malta is facing and explain how the profession can reduce their impact respectively. A few concerns were raised with regards to the CPCM, more specifically that the clinical training programme in Malta which is decided by the CPCM is still pending recognition, threatening the possibility for new MP trainees to kickstart their two year clinical training. Two respondents agreed that an independent MP council should be established by MAMP to deal with the recognition of traineeship programmes, qualification entry requirements, and professional clinical registration. Not having own MP council is not a major threat to the profession with respect to the vision. However, delays in the approval of clinical training does pose a threat for the growth of the profession.

Another respondent mentioned the threat of having the private sector loosely enforced by the authorities and that the MP association should have regular meetings with said authorities. With regards to acceptance testing and commissioning, one respondent stated:

"Acceptance testing and commissioning is not required to be done by an MPE by law."

Followed by:

"There should be peer pressure to restrict acceptance testing to engineers for installation purposes with the hospital MPE being present and MPE for commissioning and putting the equipment into clinical use."

The vision statement proposes that acceptance testing and commissioning of all medical devices be performed by a MPP, so this issue needs to be tackled by the profession.

4.8 Questionnaire Section 6: Concluding Questions

The participants were asked whether they have any ideas on how the vision statement presented in Section 4.2 could be developed further. In which 100% of the participants replied with no. When asked about if the MP profession in Malta is prepared to face the challenges of future technologies in Q6.3, the participants are ambivalent 50% agree, 37.5% are not sure, and 12.5% disagree (mean Likert score = +0.38). Table 4.15 presents the results obtained for Q6.3. Some MPs might not be entirely sure whether the profession can face future technologies because they are not certain what other physicists are doing in other specialties. Strategic leaders with a vision are needed.

The questionnaire concluded by asking the MPs if they would like to mention any points regarding the MP profession in Malta which were not covered by the survey. Where 100% of the participants responded with no.

TABLE 4.15: Statistical summary of the results obtained for Q6.3.

	N	Mean	Strongly Agree %	Agree %	Not Sure %	Disagree %	Strongly Disagree %
Q6.3	8	+0.38	0	50	37.5	12.5	0

4.9 Summary: Strengths of the Medical Physics profession in Malta with respect to the vision statement

- 1) Most MPs are certified by the RPC as MPEs.
- 2) Most MPs are certified by the RPC as RPEs.
- 3) MP profession recognised by stakeholders as being a contributor to healthcare.
- 4) High knowledge and skills in specialty areas.
- 5) MPs provide education to students.
- 6) MPs have sufficient scientific expertise regarding medical devices used in their specialty area and their clinical use.
- 7) MPs have strong competences regarding the protection of patients, workers and the general public from ionising radiation and other physical agents used in their specialty.
- 8) MPs have strong analytical, problem-solving, and trouble-shooting skills.
- 9) MPs have strong mathematical, statistical, and data analysis skills.
- 10) Strong legal foundation for the profession.
- 11) MPs have strong scientific research skills.
- 12) MPs have strong ICT skills.
- 13) Services in Malta compare well with other EU countries.
- 14) High level qualification and curriculum framework leading to the certification of a MP.
- 15) Young team that is able to adapt with deterministic views striving for improvement.
- 16) MPs take part in regional/international networking.
- Introduction of a B.Sc.(Hons) degree in Physics, Medical Physics and Radiation allowing for the profession to grow.

4.10 Summary: Weaknesses of the Medical Physics profession in Malta with respect to the vision statement

- 1) No MPs certified as a MRSE nor as a LSE.
- 2) Shortage of material resources.
- 3) MPs are mainly involved in the specification and procurement of ionising and nonionising radiation based medical devices rather than all medical devices.
- MPs in Malta are involved in the acceptance testing of ionising and non-ionising radiation based medical devices but this is not entrenched in law and not to all medical devices.
- 5) MPs in Malta are involved in commissioning but this is not entrenched in law. None of the participants indicated that they are involved in the commissioning of medical devices other than radiation based devices.
- 6) The perception is that MPs in Malta are mostly involved in the QA of ionising and non-ionising radiation based medical devices and not all medical devices.
- 7) MPs in Malta are little involved in the decommissioning process of modalities.
- 8) As a MP profession, it is mostly involved in optimising the clinical use of ionising and non-ionising radiation based medical devices only.
- 9) MPs are able to provide advice regarding patient risks and protection from ionising and non-ionising radiation but not from all physical agents.
- 10) MPs are involved in the design and installation of radiation devices but not all medical devices.
- 11) Lack of a universally acknowledged easy to promote mission statement for the profession.
- 12) Narrow range of specialties.
- 13) Insufficient robust leadership and strategic skills.
- 14) Low marketing skills.
- 15) Some MPs are reluctant to be part of the wider healthcare picture.
- 16) Some MPs have insufficient communication and pedagogical skills.

- 17) Low level of research on professional and educational issues.
- 18) Low number of MPs.

4.11 Summary: Opportunities of the Medical Physics profession in Malta with respect to the vision statement

- 1) Increase in number of hospital medical devices.
- Link up with patient organisations as patient advocates regarding safety standards in healthcare.
- Link up with occupational health and safety organisations and environmental groups and project themselves as occupational/public advocates with regard to safety from physical agents.
- 4) Exerting pressure on legislators to recognise that patient safety is not limited to ionising radiation.
- 5) Raise profile with health economists.
- 6) Self-testing and wearable devices can be turned into a business opportunity.
- 7) Research grants from the Government of Malta.
- 8) Involvement in IPEM/EFOMP/IAEA courses and conferences.
- 9) Research activities in each specialty area.
- 10) Scientific visits and fellowship programs to other hospitals.
- 11) Lecturing and project supervision at the University of Malta.
- 12) Being included as a stakeholder in RPC allows for deciding certain standards and legal notices.

4.12 Summary: Threats of the Medical Physics profession in Malta with respect to the vision statement

- 1) Low number of MPs limiting achievable EU standards.
- 2) Austerity economics.
- 3) Role poaching from other professions.
- 4) The private sector having very loose enforcing from the authority.
- 5) Lack of MP council.

4.13 List of strategic objectives for the profession

The following strategic objectives are being proposed based on the SWOT analysis.

4.13.1 Strengthening of Internal Strengths

- 1) Pursue the available regional/international approved courses on the MP profession in general to upgrade the current skills and learn new ones.
- 2) MPs should involve themselves more in the education of all healthcare professionals and research using all medical devices and protection from all physical agents, rather than limiting their function to ionising and non-ionising radiation based medical devices.
- 3) Exchange experiences between specialty areas to further increase cohesion between MPs. Schedule discussions for each specialty area and exchange written reports, say annually, outlining the objectives, goals, any obstacles, and the accomplishments of each specialty. These reports can then be presented at the MAMP annual conference.
- 4) Include more specialty areas, e.g., physiological measurement, and prepare prospective MP students more for the traineeship in the MSc curriculum in Malta.

4.13.2 Reducing Internal Weaknesses

 The main competences outlined in the suggested mission statement must get the MP members attention and any weaknesses eliminated. The MPs should not waste valuable time on areas such as daily constancy testing that can be easily replaced by other disciplines, but rather be focused on higher order competences such as quantitative methods, advanced optimisation, ICT, personalised medicine and future technologies. Focusing on higher order competences renders replacement more difficult and ensures the growth of the MP profession.

- 2) Exert pressure on local legislators such that the key competences mentioned in (1) are regulated such that a MP is involved by law.
- 3) Attend courses in LSE and MRI, more specifically, a course for MRSE. IPEM offers these courses remotely.
- 4) In order to succeed in the modern world, MPs ought to take steps to improve their limited strategic planning, advertising, and public relations skills. They cannot afford to limit their operations to the walls of their laboratories and offices; instead, they must become active participants in the larger society that surrounds them. While MPs should be proud of their scientific achievements, they should also understand the value of other skills acquired from other fields, such as business and economics. At MAMP conferences, specific courses should be organised on these subjects. Materials must be created to direct and assist MPs in pitching their expertise to stakeholders.
- 5) The MP profession should adopt the vision statement and focus all its planning in this direction.

4.13.3 Grasping Available Opportunities

- The MP members ought to be using their relationships with external organisations to request that training and education in technical competences that are necessary for the cost-effective use of medical devices and safety concerns involving physical agents be imposed through adequate legislation for the benefit of the healthcare industry.
- 2) Attend courses organised by country/regional/international organisations particularly those involving new technologies such as nanotechnology.
- 3) Apply for funds such that there is a push in research activities in each specialty area.
- 4) Link with the RPC such that the profession has a stronger presence and MPs can exchange views with the commission on a regular basis.
- 5) MAMP should encourage its members to take part in European Network for Training and Education of Medical Physics Experts (EUTEMPE) (EUTEMPE, n.d.) consortium and EFOMPs European School for Medical Physics Experts (ESMPE) courses.

Courses are available for such areas as leadership, advanced quality control, radiation protection and advanced MRI, all of which would elevate the competences of MPs in Malta to exceptional levels. The European Nuclear Education Network (ENEN+) (ENEN+, n.d.) consortium will be making funds available for scholarships for participants to these courses.

4.13.4 Countering External Threats

- Insist with the CPCM that approval of training schemes be accelerated. Lack of MPs in Malta is the root of several problems that the profession is facing, an increase in MPs would allow for the vision statement to be realised.
- Take proactive measures to combat any attempt in introducing the idea that MPs can be replaced by other members of other professions, which is very hazardous for the MP profession.
- 3) Exert pressure on legislators to impose stricter regulations on the private sector with respect to the employment of MPs.

4.14 Conclusion

This chapter presented the results from the SWOT analysis with respect to the vision statement based on the data obtained from the semi-structured anonymous questionnaire. The data gathered was then thoroughly discussed and suggested strategic objectives for the MP profession in Malta were offered. Chapter five will present conclusions of the study and recommendations for the professional practice and future research.
Chapter 5

Conclusions and Recommendations

5.1 Introduction

A summary of the findings in this study are presented in Section 5.2. Section 5.3 and 5.4 present possible recommendations for the professional practice and future research respectively in relation to the dissertation.

5.2 Summary of Conclusions from the Study

The main findings of the study are listed hereunder:

- a) The MPs provided a comprehensible viewpoint of the profession, they pointed out several strengths and most of the respondents mentioned several goals that they would like to see in fruition for the MP profession in Malta. The participants also pointed some weaknesses, ways in which the profession can grow by grasping available opportunities, whilst highlighting the external threats they are currently facing.
- b) There are both merits and issues in the present state of the MP profession. The MP profession is endowed with strong competences, each specialty area having a team of high calibre professionals due to their analytical and scientific expertise owing to their backgrounds in physics and engineering. This allows the MPs to grasp in-depth understanding of medical devices and be highly trained with regards to the protection of patients and the general public from ionising radiation and other physical agents. However, due to the problems brought on by the fact that MP members are still a relatively new and small profession in Malta, most of the possible advancements have still not been realised. MPs have very strong involvement in ionising and non-ionising radiation based medical devices, overall, the perception is that the profession

is very weak when it comes to having MPs involved in other medical devices and other associated physical agents, which is a weakness with respect to the vision statement. On the other hand, several opportunities that can help in eliminating weaknesses were also recognised. The MPs identified the possibility of attending courses provided by several international organisations such that their competences could be improved, whilst also identifying ways in which the profession can further improve its profile. However, some threats were also derived from the questionnaire. The MPs acknowledge the fact that even though there are many opportunities, the lack of new MPs might hinder the profession from achieving EU standards. Other threats mentioned were role poaching from other healthcare professionals and austerity economics.

c) Based on the perspective of the local MPs in Malta, the SWOTs of the profession were analysed, and a list of strategic objectives were presented. The list consisted of ways to further strengthen internal strengths, reduce internal weaknesses, grasp available opportunities, and counter external threats of the profession.

5.3 **Recommendations for Professional Practice**

Recommendations for the MP professional practice in Malta are:

- a) The MP profession and MAMP should consider adopting the strategic objectives suggested by this study.
- b) Till more MPs are forthcoming, MPs should offer to dedicate more of own time in pursuing activities such as involvement in professional societies, development of high quality presentations, collaborative research with other departments abroad, and courses to further develop standards.
- c) Following the transition to local MP clinical training, the future of the MP profession in Malta depends on the present local MPs. With each passing year, the intricacy of procedures in RO, D&IR, and NM is rising. In order to reflect the shifting environment in hospitals abroad and in Malta, it is crucial that training programmes and competences be constantly evaluated and updated.

5.4 Avenues for Future Research

Recommendations for future research are:

- a) Clarify the SWOTs further by including in-depth interviews with both MPs and other stakeholders. It is important to study not only how MPs see themselves but also how they are seen by other stakeholders.
- b) Carrying out a SWOT study of the MP profession from the perspective of MPs across Europe. This study was conducted from the viewpoint of the MPs in Malta, incorporating the various perspectives of other MPPs across an entire continent would have more far reaching effects and the outcome will surely allow for the development of a highly comprehensive strategic plan for the MP profession at a European level.

5.5 Conclusion

The objectives of this study were to develop a vision statement for the MP profession in Malta, to identify the SWOTs of the MP profession with respect to that vision as perceived by the MPs themselves and to generate a strategic plan for the way forward. These objectives have been largely achieved. However in-depth interviews with both MPs and other stakeholders would certainly lead to a more sophisticated and comprehensive strategic plan.

References

- 2006/25/EC. Directive 2006/25/EC of the European parliament and of the council of 5 april 2006 on the minimum health and safety requirements regarding the exposure of workers to risks arising from physical agents (artificial optical radiation) (19th individual directive within the meaning of article 16(1) of directive 89/391/eec). *Official Journal of the European UnionL 114/38*, 2006. URL http://data.europa.eu/eli/dir/2006/25/oj.
- 2013/59/EURATOM, C. D. Council directive 2013/59/Euratom of 5 december 2013 laying down basic safety standards for protection against the dangers arising from exposure to ionising radiation, and repealing directives 89/618/Euratom, 90/641/Euratom, 96/29/Euratom, 97/43/Euratom and 2003/122/Euratom. *Official Journal of the European Union L 13/1*, 2013. URL http://data.europa.eu/eli/dir/2013/59/oj.
- 288, L. N. Bye-laws for the degree of bachelor of science (honours) in physics, medical physics and radiation protection - B.Sc.(Hons) - under the auspices of the faculty of health sciences and the faculty of science. *SUBSIDIARY LEGISLATION 327.637*, 2019.
- 397, L. N. Bye-laws for the degree of master of science in medical physics under the auspices of the faculty of health sciences. SUBSIDIARY LEGISLATION 327.434, 2012.
- Andersson, J., Nyholm, T., Ceberg, C., Almén, A., Bernhardt, P., Fransson, A., and Olsson, L. E. Artificial intelligence and the medical physics profession-A Swedish perspective. *Physica Medica*, 88:218–225, 2021.
- Aslan, I., Çınar, O., and Özen, Ü. Developing strategies for the future of healthcare in turkey by benchmarking and SWOT analysis. *Procedia-Social and Behavioral Sciences*, 150:230–240, 2014.
- Bart, C. K. Measuring the mission effect in human intellectual capital. *Journal of intellectual capital*, 2001.

- Busoni, S., Bock, M., Chmelik, M., Colgan, N., De Bondt, T., Hanson, L., Israel, M., Kugel, H., Maieron, M., Mazzoni, L., et al. Addendum to EFOMP policy statement no. 14 "the role of the medical physicist in the management of safety within the magnetic resonance imaging environment: EFOMP recommendations". *Physica Medica: European Journal of Medical Physics*, 89:303–305, 2021.
- Cardona, P. and Rey, C. Management by missions: how to make the mission a part of management. *Problems and Perspectives in Management*, 4(1):164–174, 2006.
- Caruana, C. J. Leadership and Challenges in Medical Physics: A Strategic and Robust Approach. 2053-2563. IOP Publishing, 2020. ISBN 978-0-7503-1395-7. doi: 10.1088/978-0-7503-1395-7. URL https://dx.doi.org/10.1088/978-0-7503-1395-7.
- Caruana, C. J. and Grech, K. Cost optimised medical physics education and training: an innovative E&T scheme in Malta. *Medical Physics International Journal*, 4(1):17–19, 2016.
- Caruana, C. J., Wasilewska-Radwanska, M., Aurengo, A., Dendy, P. P., Karenauskaite,
 V., Malisan, M. R., Meijer, J. H., Mihov, D., Mornstein, V., Rokita, E., et al. A comprehensive SWOT audit of the role of the biomedical physicist in the education of healthcare professionals in Europe. *Physica medica*, 26(2):98–110, 2010.
- Caruana, C. J., Christofides, S., and Hartmann, G. H. European federation of organisations for medical physics (EFOMP) policy statement 12.1: recommendations on medical physics education and training in europe 2014. *Physica Medica*, 30(6): 598–603, 2014.
- Caruana, C. J., Tsapaki, V., Damilakis, J., Brambilla, M., Martín, G. M., Dimov, A., Bosmans, H., Egan, G., Bacher, K., and McClean, B. EFOMP policy statement 16: the role and competences of medical physicists and medical physics experts under 2013/59/Euratom. *Physica Medica*, 48:162–168, 2018.
- Caruana, C. J., Agius, S., Bickle, J. M., Vella, N., Said, C., Napoli, N., Farrugia, J., Bezzina, M., and Pace, E. Medical physics in malta: Past, present and future. *MEDICAL PHYSICS*, 9(2), 2021.
- Christofides, S., Isidoro, J., Pesznyak, C., Cremers, F., Figueira, R., van Swol, C., Evans, S., and Torresin, A. The European federation of organisations for medical physics policy statement no. 10.1: recommended guidelines on national schemes for

continuing professional development of medical physicists. *Physica Medica*, 32(1): 7–11, 2016.

- Clark, T., Foster, L., Sloan, L., and Bryman, A. *Bryman's social research methods*. Oxford University Press, 2021.
- Collis, D. J. and Rukstad, M. G. Can you say what your strategy is? *Harvard business review*, 86(4):82–90, 2008.
- CPCM. Benchmarking document for medical physics. Coun-2014. URL cil of Professions *Complementary* to Medicine, https://deputyprimeminister.gov.mt/en/regcounc/cpcm/Documents/Medical Physics Benchmark.pdf.
- Creswell, J. W. Mixed-method research: Introduction and application. In *Handbook of educational policy*, pages 455–472. Elsevier, 1999.
- Dou, L. Quantitative vs qualitative. [PDF document]. University of Buckingham, 2021.
- Draper, J. C. A Text-book of Medical Physics: For the Use of Students and Practitioners of Medicine. Lea Brothers, 1885.
- Duck, F. The curie papers on piezoelectricity. Nowotwory J Oncol, 60:118e-124e, 2010.
- Duck, F. A. The origins of medical physics. *Physica medica*, 30(4):397–402, 2014.
- Duties DIR. Duties of MPE in the area of diagnostic interventional radiology. *PDF Document*, 2022.
- EFOMP. Policy statement 2: The roles, responsibilities and status of the clinical medical physicist. 1984. URL https://www.efomp.org/uploads/policy_statement_nr_2.pdf.
- EFOMP. Professional matters committee. Retrieved on September 5, 2022, n.d. URL https://www.efomp.org/index.php?r=structure/view&id=9.
- ENEN+. European nuclear education network. Retrieved on September 10, 2022, n.d. URL https://enen.eu/.
- EQF. Recommendation of the European parliament and of the council of 23 april 2008 on the establishment of the European qualifications framework for lifelong learning. *European Qualifications Framework*, 2008. URL http://www.eucen.eu/EQFpro/GeneralDocs/FilesFeb09/GLOSSARY.pdf.

- EU Commission. European guidelines on medical physics expert. radiation protection series 174. 2014. URL http://ec.europa.eu/energy/sites/ener/files/documents/174.pdf.
- EUTEMPE. European network for training and education of medical physics experts. Retrieved on September 10, 2022, n.d. URL https://eutempe-net.eu/.
- Evans, S., Christofides, S., and Brambilla, M. The European federation of organisations for medical physics. Policy statement no. 7.1: The roles, responsibilities and status of the medical physicist including the criteria for the staffing levels in a medical physics department approved by EFOMP council on 5th february 2016. *Physica Medica*, 32 (4):533–540, 2016.
- Figueira, R., Poli, E., and Isidoro, J. SWOT analysis of the current situation of medical physicists in portugal. *Physica Medica*, 32:332, 2016.
- Fowler Jr, F. J. and Fowler, F. J. *Improving survey questions: Design and evaluation*. Sage, 1995.
- Gandolf, S. SWOT: The high-level self exam that boosts your bottom line. *Healthcare Success*, 2021.
- Ghazinoory, S., Abdi, M., and Azadegan-Mehr, M. SWOT methodology: a state-of-theart review for the past, a framework for the future. *Journal of business economics and management*, 12(1):24–48, 2011.
- Hand, J., Bosmans, H., Caruana, C., Keevil, S., Norris, D. G., Padovani, R., and Speck,
 O. The European federation of organisations for medical physics policy statement no
 14: The role of the medical physicist in the management of safety within the magnetic resonance imaging environment: EFOMP recommendations. *Physica medica*, 29(2):
 122–125, 2012.
- IAEA. Clinical training of medical physicists specializing in radiation oncology. training course series, 37. *International Atomic Energy Agency*, 2009. URL https://www-pub.iaea.org/MTCD/Publications/PDF/TCS-37_web.pdf.
- IAEA. Clinical training of medical physicists specializing in diagnostic radiology. training course series, 47. *International Atomic Energy Agency*, 2010. URL https://wwwpub.iaea.org/MTCD/Publications/PDF/TCS-47_web.pdf.
- IAEA. Clinical training of medical physicists specializing in nuclear medicine.training course series, 50. *International Atomic Energy Agency*, 2011. URL https://www-pub.iaea.org/MTCD/Publications/PDF/TCS-50_web.pdf.

- IAEA. Roles and responsibilities, and education and training requirements for clinically qualified medical physicists. *HUMAN HEALTH SERIES No. 25*, 2013. URL https://www-pub.iaea.org/MTCD/Publications/PDF/Pub1610_web.pdf.
- International Labour Organisation. International Standard Classification of Occupations 2008 (ISCO-08): Structure, group definitions and correspondence tables. International Labour Office, Geneva, 2012. ISBN 978-92-2-125953-4. URL https://www.ilo.org/global/publications/ilo-bookstore/ order-online/books/WCMS_172572/lang--en/index.htm.
- Lawlor, J. E. The importance of strategic planning. *London: Observatory on Borderless Higher education Long Range Planning*, 21(1):73–81, 2010.
- Legg, C. and Hookway, C. Pragmatism. 2008.
- Lewis, B. R. and Littler, D. *The Blackwell encyclopedic dictionary of marketing*. Blackwell Publishing Ltd., 1997.
- L.N.184. 184 of 2020 Basic safety standards for ionising radiation (amendment) regulations,. *Gazzetta tal-Gvern ta' Malta Nru. 20,400, 5 ta' Mejju, 2020, 2020.* URL https://legislation.mt/eli/ln/2020/184.
- L.N.210. 210 of 2018 Basic safety standards for ionising radiation regulations,. *Gazzetta tal-Gvern ta' Malta Nru. 20,020, 3 ta' Lulju, 2018, 2018.* URL https://legislation.mt/eli/ln/2018/210.
- MAMP. Statute and rules, 2019. URL http://mamp.org.mt/wp-content/ uploads/2021/11/statute.pdf.
- Matteucci, C. Sui fenomeni fisico-chimici dei corpi viventi lezione vii. *Il Cimento*, 2(1): 5–18, 1844.
- OSHwiki. Physical agents. *Occupational Safety and Health wikipedia*, Retrieved on August 31, 2022, 2022. URL https://oshwiki.eu/wiki/Physical_agents.
- Pelletan, P. *Traité élémentaire de physique générale et médicale*, volume 1. Gabon, 1824.
- Perera, F. d. P. R. and Peiró, M. Strategic planning in healthcare organizations. *Revista Española de Cardiología (English Edition)*, 65(8):749–754, 2012.

- Phadermrod, B., Crowder, R. M., and Wills, G. B. Importance-performance analysis based SWOT analysis. *International journal of information management*, 44:194–203, 2019.
- Rossman, G. B. and Wilson, B. L. Numbers and words revisited: Being "shamelessly eclectic". *Quality and quantity*, 28(3):315–327, 1994.
- RPC. Document 54: Criteria for the recognition of medical physics experts under the basic safety standards for ionising radiation regulations (sl585.01). *Commission for the Protection From Ionising and Non-Ionising Radiation*, 2021. URL https://rpc.gov.mt/en/Documents/DOC 54 MPE Criteria.pdf.
- S.L.585.01. S.L.585.01 of 2018 Basic safety standards for ionising radiation regulations, SUBSIDIARY LEGISLATION 585.01, 2018. LEGAL NOTICE 210 of 2018, as amended by Legal Notice 184 of 2020. URL https://eur-lex.europa.eu/legalcontent/EN/TXT/PDF/?uri=NIM:286730.
- Steiner, G. A. Strategic planning. Simon and Schuster, 2010.
- Terzic-Supic, Z., Bjegovic-Mikanovic, V., Vukovic, D., Santric-Milicevic, M., Marinkovic, J., Vasic, V., and Laaser, U. Training hospital managers for strategic planning and management: a prospective study. *BMC medical education*, 15(1): 1–9, 2015.
- Topor, D. R., Dickey, C., Stonestreet, L., Wendt, J., Woolley, A., and Budson, A. Interprofessional health care education at academic medical centers: using a SWOT analysis to develop and implement programming. *MedEdPORTAL*, 14:10766, 2018.
- Turankar, A., Motghare, V., Kinage, P., Deshpande, R., Sinha Sh, T. S., et al. SWOT analysis in medical sciences. *J Rational Pharmacother Res*, 2(2):39–43, 2014.
- Vella, K., Caruana, C., Vella, N., Schembri, K., Camilleri, M., Busuttil, M., and Pace, E. Strategic planning: Case study for a diagnostic radiology constancy testing programme in a major hospital in malta. *MEDICAL PHYSICS*, 10(1), 2022.
- Von Kodolitsch, Y., Bernhardt, A. M., Robinson, P. N., Kölbel, T., Reichenspurner, H., Debus, S., and Detter, C. Analysis of strengths, weaknesses, opportunities, and threats as a tool for translating evidence into individualized medical strategies (I-SWOT). *Aorta*, 3(03):98–107, 2015.

Appendix A

EU Directive 2013/59/Euratom Article 57(1b) and Article 83(2)

Official Journal of the European Union

 Member States shall ensure that for each medical or biomedical research project involving medical exposure:

(a) the individuals concerned participate voluntarily;

- (b) these individuals are informed about the risks of exposure;
- (c) a dose constraint is established for individuals for whom no direct medical benefit is expected from exposure;
- (d) in the case of patients who voluntarily accept to undergo an experimental medical practice and who are expected to receive a diagnostic or therapeutic benefit from this practice, the dose levels concerned shall be considered on an individual basis by the practitioner and/or referrer prior to the exposure taking place.

4. Member States shall ensure that the optimisation includes the selection of equipment, the consistent production of adequate diagnostic information or therapeutic outcomes, the practical aspects of medical radiological procedures, quality assurance, and the assessment and evaluation of patient doses or the verification of administered activities, taking into account economic and societal factors.

- 5. Member States shall ensure that:
- (a) dose constraints are established for the exposure of carers and comforters, where appropriate;
- (b) appropriate guidance is established for the exposure of carers and comforters.

6. Member States shall ensure that in the case of a patient undergoing treatment or diagnosis with radionuclides, the practitioner or the undertaking, as specified by Member States, provides the patient or their representative with information on the risks of ionising radiation and appropriate instructions with a view to restricting doses to persons in contact with the patient as far as reasonably achievable. For therapeutic procedures these shall be written instructions.

These instructions shall be handed out before leaving the hospital or clinic or a similar institution.

Article 57

Responsibilities

1. Member States shall ensure that:

- (a) any medical exposure takes place under the clinical responsibility of a practitioner;
- (b) the practitioner, the medical physics expert and those entitled to carry out practical aspects of medical radiological procedures are involved, as specified by Member States, in the optimisation process;

- (c) the referrer and the practitioner are involved, as specified by Member States, in the justification process of individual medical exposures;
- (d) wherever practicable and prior to the exposure taking place, the practitioner or the referrer, as specified by Member States, ensures that the patient or their representative is provided with adequate information relating to the benefits and risks associated with the radiation dose from the medical exposure. Similar information as well as relevant guidance shall be given to carers and comforters, in accordance with point (b) of Article 56(5).

 Practical aspects of medical radiological procedures may be delegated by the undertaking or the practitioner, as appropriate, to one or more individuals entitled to act in this respect in a recognised field of specialisation.

Article 58

Procedures

Member States shall ensure that:

- (a) written protocols for every type of standard medical radiological procedure are established for each equipment for relevant categories of patients;
- (b) information relating to patient exposure forms part of the report of the medical radiological procedure;
- (c) referral guidelines for medical imaging, taking into account the radiation doses, are available to the referrers;
- (d) in medical radiological practices, a medical physics expert is appropriately involved, the level of involvement being commensurate with the radiological risk posed by the practice. In particular:
 - (i) in radiotherapeutic practices other than standardised therapeutic nuclear medicine practices, a medical physics expert shall be closely involved;
 - (ii) in standardised therapeutical nuclear medicine practices as well as in radiodiagnostic and interventional radiology practices, involving high doses as referred to in point (c) of Article 61(1), a medical physics expert shall be involved;
 - (iii) for other medical radiological practices not covered by points (a) and (b), a medical physics expert shall be involved, as appropriate, for consultation and advice on matters relating to radiation protection concerning medical exposure.

17.1.2014

17.1.2014 EN

L 13/33

 The radiation protection expert may be assigned, if provided for in national legislation, the tasks of radiation protection of workers and members of the public.

Article 83

Medical physics expert

 Member States shall require the medical physics expert to act or give specialist advice, as appropriate, on matters relating to radiation physics for implementing the requirements set out in Chapter VII and in point (c) of Article 22(4) of this Directive.

2. Member States shall ensure that depending on the medical radiological practice, the medical physics expert takes responsibility for dosimetry, including physical measurements for evaluation of the dose delivered to the patient and other individuals subject to medical exposure, give advice on medical radiological equipment, and contribute in particular to the following:

- (a) optimisation of the radiation protection of patients and other individuals subject to medical exposure, including the application and use of diagnostic reference levels;
- (b) the definition and performance of quality assurance of the medical radiological equipment;
- (c) acceptance testing of medical radiological equipment;
- (d) the preparation of technical specifications for medical radiological equipment and installation design;
- (e) the surveillance of the medical radiological installations;
- (f) the analysis of events involving, or potentially involving, accidental or unintended medical exposures;
- (g) the selection of equipment required to perform radiation protection measurements;
- (h) the training of practitioners and other staff in relevant aspects of radiation protection;
- The medical physics expert shall, where appropriate, liaise with the radiation protection expert.

Article 84

Radiation protection officer

1. Member States shall decide in which practices the designation of a radiation protection officer is necessary to supervise or to perform radiation protection tasks within an undertaking. Member States shall require undertakings to provide the radiation protection officers with the means necessary for them to carry out their tasks. The radiation protection officer shall report directly to the undertaking. Member States may require employers of outside workers to designate a radiation protection officer as necessary to supervise or perform relevant radiation protection tasks as they relate to the protection of their workers.

Depending on the nature of the practice, the tasks of the radiation protection officer in assisting the undertaking, may include the following:

- ensuring that work with radiation is carried out in accordance with the requirements of any specified procedures or local rules;
- (b) supervise implementation of the programme for workplace monitoring;
- (c) maintaining adequate records of all radiation sources;
- (d) carrying out periodic assessments of the condition of the relevant safety and warning systems;
- (e) supervise implementation of the personal monitoring programme;
- (f) supervise implementation of the health surveillance programme;
- (g) providing new workers with an appropriate introduction to local rules and procedures;
- (h) giving advice and comments on work plans;
- (i) establishing work plans;
- (j) providing reports to the local management;
- (k) participating in the arrangements for prevention, preparedness and response for emergency exposure situations;
- (l) information and training of exposed workers;
- (m) liaising with the radiation protection expert.

 The task of the radiation protection officer may be carried out by a radiation protection unit established within an undertaking or by a radiation protection expert. **Appendix B**

List of knowledge, skills and competences for the MPE as specified by the EU guidelines for the MPE project

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K1. Explain statutory and institutional requirements for Medical Physics Services in own area of medical physics practice with respect to	 S1. Apply the general concepts, principles, theories and practices of physics to 	C1. Take responsibility for statutory and institutional requirements for Medical Physics Services in own area of
	Scientific Problem Solving Service.	the solution of clinical problems	medical physics practice with respect to Scientific
	K2 LIse physics concepts principles and theories to explain in detail	concerning the optimised clinical use	Problem Solving Service.
	and quantitatively. the structure, functioning, characteristics	of medical devices and safety / risk	C2. Take responsibility for the setting-up and organization of
	strengths and limitations and use of the medical devices used in own	management with respect to	a Medical Physics Service in own area of medical
	area of medical physics.	associated romany radiations and other physical agents	physics practice.
	K3. Explain in detail and quantitatively the properties of ionising		C3. Take responsibility for applying the general concepts,
	radiations (electromagnetic, electrons, ions, neutrons) and other	 Use the general concepts, principles, 	principles, theories and practices of physics to the
	physical agents (e.g., electrical energy, static electric / magnetic	theories and practices of physics to	solution of clinical problems concerning the optimal use
	fields, non -ionising electromagnetic radiation, vibration, sound and	analyze the research literature	of medical devices and management of risk from
ə:	ultrasound, heat energy and laser) to be found in the healthcare	concerning the optimised use of	associated ionizing radiations and other physical agents
oiv	environment.	medical devices and safety / risk	in own area of medical physics practice.
eL	K4. Explain quantitatively using biological models the beneficial and/or	management with respect to ionizing	C4. Take responsibility for applying the general concepts,
s	adverse biological effects of ionizing radiations and the various	radiations and other associated	principles, theories and practices of physics to analyze
бι	physical agents associated with medical devices, the factors	physical agents.	the research literature concerning the optimal use of
νir	influencing the magnitude of the biological effect and the way these	 Use physics research skills to develop 	medical devices and management of risk from associated
0	can be manipulated to improve clinical outcomes e.g., in the case of	the experimental evidence base for the	ionizing radiations and other physical agents and to
s	ionizing radiation this would include radiobiological models, radiation	optimal use of medical devices and	transfer relevant published research results to the clinical
we	epidemiology, mutagenesis, carcinogenesis (including	safety / risk management from	environment in own area of medical physics practice.
piq	leukaemogenesis), genetic effects on offspring from irradiation of	associated ionizing radiations and	C5. Take responsibility to apply the general concepts,
LO LO	gametes, teratogenic effects on the conceptus, skin effects, eye	physical agents when present	principles, theories and practices of physics for the
Ы	cataracts, cell survival curves, linear-quadratic model, absorbed	evidence is insufficient.	selection and insertion of new medical devices within
oi'	dose, type of radiation (RBE, radiation weighting factor), tissue	 S4. Use the general concepts, principles, 	own area of medical physics practice and to facilitate the
hđr	radiosensitivity (LET, RBE, tissue weighting factor), dose rate,	theories and practices of physics to	effective, safe and economical use of said devices.
ıəi	presence of radiosensitisers, oxygen and radioprotectors, age, dose-	ensure effective and safe practice in	C6. Take responsibility to apply physics research skills to
25	effect relationships.	own area of medical physics practice.	develop the evidence base for the optimal use of medical
;	K5. Explain the application of the terms deterministic/stochastic,	 Use the general concepts, principles, 	devices in own area of medical physics practice when
	early/late and teratogenic/genetic effects in relation to each physical	theories and practices of physics for	present evidence is insufficient.
	agent.	the transfer of new medical devices	
	K6. Explain the main sources of evidence from within the general	and associated techniques to the	
	physics, medical physics and healthcare literature (e.g., the	clinical environment in an effective,	
	Cochrane Collaboration) essential for the carrying out of a systematic	safe and economical manner.	
	survey in own area of medical physics practice.	S6. Design quantitative clinical and	
		biomedical studies based on rigorous	
		statistical design.	
		S7. Use statistical packages for the	
		analysis of clinical and biomedical	
		data	

Annex 1/12

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K7. Explain and explain the statutory and institutional requirements of	S8. Select and use instruments for	C7. Take responsibility for statutory and institutional
(Medical Physics Services with respect to Dosimetry Measurements	dosimetric quantities for the various	requirements for Medical Physics Services in own area of
do	(including non-ionising radiations as appropriate).	types of ionizing radiations and other	medical physics practice with respect to Dosimetry
bro	K8 Define and evulain the dosimetric quantities (including units and	physical agents for patients, workers	Measurements (including non-ionising radiations as
de	ive. Ectime and expansions are desiring a series durations and and a inter-relationships) used to assess beneficial or adverse biological	and public in own area of medical	appropriate).
s	effects for ionizing radiations and the various types of physical	physics practice.	C8. Design and equip an appropriate space for the
e s	agents in own area of medical physics practice (use ICRU 85, 2011	 Develop rigorous dosimetry protocols 	measurement of dosimetric quantities for the various
ţu	definitions for ionizing radiation).	in own area of medical physics	types of ionizing radiations and physical agents for
əß	K9. Define patient dosimetric quantities for each clinical procedure in	practice.	patients, workers and public in own area of medical
е	own area of medical physics practice and explain the method used	S10. Interpret the results of dosimetry	physics practice.
e	for their measurement / calculation.	measurements.	 C9. Take responsibility for the selection, acceptance testing,
bie	K10. Explain the relationship between the various dosimetric	S11. Maintain calibration of dosimetry	commissioning and quality control of instruments for the
Λu	quantities used (e.g., between energy fluence, kerma and absorbed	instruments.	measurement of dosimetric quantities for ionizing
d	dose for photon beams including the concept of charged particle	S12. Implement cross-calibration	radiations and other physical agents in own area of
er	equilibrium).	procedures for dosimetry instruments.	medical physics practice. In the case of acceptance
цţ	K11. Define operational quantities (including units and inter-	S13. Convert dosimetry quantities	testing this should be done in cooperation with the
0	relationships) used in personal dosimetry e.g., ambient H*(10),	measured in air or other medium to	vendor.
ou	directional H'(0.07, angle) and personal dose equivalents i.e., depth	relevant dosimetric quantities in tissue.	C10. Take responsibility for the handling, management,
e)	dose equivalent H _p (10) and skin dose equivalent H _p (0.07) for		calibration and maintenance of dosimetry instruments in
st	external photon radiation and explain the method used for their		own area of medical physics practice.
uə	measurement / calculation.		C11. Take responsibility for dosimetric investigations and
u	K12. Explain in detail and quantitatively the structure, operation and		the supervision of dosimetry measurements.
ə.ir	advantages / disadvantages of the various types of patient and		
ISI	personal dosimeters and area monitors available for the various		
69	types of ionising and non-ionising radiation including criteria for		
W	selection (e.g., accuracy, precision, uncertainties, linearity, any dose		
٨ı	rate / energy / directional dependence, spatial resolution, physical		
j ⊖ι	size, read out convenience and convenience of use), management,		
nia	calibration, traceability (including international traceability framework)		
0			
	cavity theory).		
	K13. Explain the principles of biological monitoring / dosimetry.		

Appendix B. List of knowledge, skills and competences for the MPE as specified by theEU guidelines for the MPE project72

Annex 1/13

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K14. Explain the statutory and institutional requirements for Medical	S14. Calculate patient risk from	C12. Take responsibility for statutory and institutional
	Physics Services in own area of medical physics practice with	measurement data or the dosimetry	requirements for medical Physics Services in own area of
	respect to Patient Safety / Kisk Management.	quantities used to assess adverse biological officite for the verticus times	Diek Monocomport
	K15. Explain the classification of medical devices based on patient	of ionizing radiations and other	
	risk. 1746 – Frankrig Mendelsen of entropy with monotonical to	physical agents.	C13. Inventorise sources of ionizing radiations and other
	A TO. Explain the principles of patient risk management as applied to		priysical agents present in the nospital environment with
	medical devices and associated ionizing radiations and other abusical counts in cum accorde modical abusics accetion	515. Assess patient risks from given	respect to patient salety. C14 Take research the for the proving estimization of
	priysical agents in own area of incurval priysics practice. K17 Evoloin the heneficial and nossible adverse highorizal effects	procedures in OWI area OFITIEULA physics practice from mossified	C 14. I dive responsibility for une origonity ophimization of evicting and newly introduced protocols in own area of
	ALL: Explain the beneficial and possible adverse biological effects (including mechanisms) to patients of ionizing radiations and other	priysics practice inon measured patient dose data and dose-effect	existing and rewry introduced protocols in own area of medical physics practice with respect to patient
	physical agents including the factors impacting the magnitude of the	relationships.	protection and in accordance with the latest published
	biological effect.	S16. Apply the principles of justification	evidence or own research when the available evidence is
ţ	K18. Explain the possible impact of human factors with regard to	(risk / benefit assessment),	not sufficient.
uə	patient safety in the use of medical devices and associated ionizing	optimization (including ALARA) and	C15. Carry out an ionizing radiation and other physical
w	radiations and other physical agents.	the setting up of reference levels to	agent dose audit with respect to patient safety in own
aß	K19. Explain the difference between deterministic/stochastic,	protect the patient from unnecessary	area of medical physics practice.
eu	early/late and teratogenic/genetic effects of the various ionizing	risk from ionizing radiations and other	C16. Take responsibility for the development of patient
ey	radiations and other physical agents in relation to patient risk.	physical agents.	safety teams in own area of medical physics practice.
N)	K20. Explain relevant international, EU, national and local legislation,	S17. Apply the various means of dose	C17. Implement corrective procedures with regard to
lsi	recommendations and documentation regarding risk from ionizing	reduction (appropriate source	patient safety in own area of medical physics practice.
Ы	radiations and other physical agents with the purpose of hazard	strengths, exposure time, distance,	C18. Take responsibility for the planning for emergency
,	prevention and emergency preparedness in the healthcare	shielding) in protocol optimization.	situations with regard to patient safety in own area of
àə,	environment with regard to patient safety / risk management.	S18. Calculate risks to the unborn child	medical physics practice.
tei	K21. Explain the procedures for the prevention, investigation and	in the case of exposure to ionizing	C19. Implement a detailed organisational policy to support
s i	handling of adverse incidents (including use of Root Cause Analysis /	radiations and other physical agents.	the safety of patients in own area of medical physics
uə	Failure Modes and Effects Analysis or alternative methodology;	S19. Develop an organisational policy	practice.
oite	recommendations of appropriate remedial actions) with respect to	to achieve regulatory compliance for	C20. Take responsibility for the establishment and use of
^s d	patients in own area of medical physics practice.	patient safety from ionizing radiations	appropriate reference levels with respect to risks from
	K22. Explain the process and practical implementation of patient risk	and other physical agents in own area	ionizing radiations and other physical agents.
	assessments in own area of medical physics practice, using	of medical physics practice.	C21. Develop contingency plans for emergency
	techniques for the qualitative and quantitative assessment of risk.	S20. Investigate incidents to determine	procedures with respect to patient safety in own area of
	K23. Name and explain the function of the main National, European	the cause(s) and recommending	medical physics practice.
	and International organizations concerned with protection of patients	appropriate remedial action with	C22. Take responsibility for the design of a new facility
	from ionizing radiations and other physical agents (e.g., ICRP,	respect to patient safety in own area of	(including waiting and resting rooms) in own area of
	ICNIRP, IAEA, EC, WHO, UNSCEAR).	medical physics practice.	medical physics practice taking into consideration patient
	K24. Explain how research exposures are managed in own area of	S21. Conduct critical examinations	safety.
	medical physics practice including the processes of ethical review	(interlocks, warning systems, safety	C23. Take responsibility for the surveillance of
	and including the use of dose constraints where appropriate.	design features and barriers) related to	installations with respect to protection of patients from
	K25. Explain the requirements for, and the practical implementation	patient safety in own area of medical	ionizing radiations and other physical agents.
	of, appropriate systems for the monitoring of doses to patients from	physics practice.	C24. Investigate in conjunction with the vendor technical
	ionizing radiations and other physical agents in own area of medical		factors leading to accidental/unintended exposures.
	physics practice.		

Annex 1/14

		Knowledge (facts, principles, theories, practices)	Skills (cognitive and practical)	Competence (responsibility and autonomy)
	Š	Explain the principles and practice of contingency planning and the implementation of emergency procedures with respect to patient safety in own area of medical physics practice.	S22. Give advice on the choice and use of protective equipment related to patient safety in own area of medical	C25. Take responsibility for the management of good and safe practice in the use of ionising radiation beams and sealed / unsealed sources in own area of medical
(cont.)	K27	 Explain the key considerations for the design of a new facility (including waiting and resting rooms) with regards to patient safety in own area of medical physics practice. 	physics practice. S23. Assess patient risks for a given experimental procedure.	physics practice in relation to patient safety.
juəməp	23 23	 Explain the functioning of safety systems (e.g., interlocks) found in own area of medical physics practice with respect to patient safety. Explain how the application of good safety practices and the use 		
eneN	83	or appropriate devices and recrimingues are used to optimize climical protocols. Explain quantitatively and in detail the interactions with organic		
Neisk I		matter of ionising and non-ionising electromagnetic radiations, particulate radiation, ultrasound and electric and magnetic fields at the molecular, cellular, tissue and macroscopic levels in relation to		
ş1ety	K31	patient risks. Define the radiation dosimetry quantities used in patient risk		
sS frie	K32	 assessment and their use in the radiation protection of patients. Explain the principles of the design of radiation safety plans with respect to patient safety in own area of medical physics practice. 		
Iseq	ξ3 Ξ	Explain the fundamental characteristics and limitations of the various models / algorithms used in the quantification of patient		
	K34	doses from external sources or ionising radiation. Explain compartmental / bio-kinetic models and the fundamental characteristics and limitations of the MIRD model and algorithms for internal radionuclide patient dosimetry. 		

Appendix B. List of knowledge, skills and competences for the MPE as specified by theEU guidelines for the MPE project74

Annex 1/15

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K35. Explain statutory and institutional roles of Medical Physics	S24. Perform occupational / public risk	C26. Take responsibility for statutory and institutional
	Services with respect to Occupational and Public Safety / Kisk	assessment based on facility survey	requirements for Medical Physics Services in own area of
	Management in own area of medical physics practice when there is	and estimated / measured dosimetry	medical physics practice with respect to Occupational
	an impact on medical exposure or own safety.	data in own area of medical physics	and Public Safety / Kisk Management when there is an
	K36. Explain the possible adverse biological effects (including	practice.	impact on medical exposure or own safety.
	mechanism) to workers / public from ionizing radiations (and other	S25. Assess occupational risk from	
	physical agents if approp) including the factors impacting the	given procedures in own area of	
(λ,	magnitude of the biological effect.	medical physics practice from ionizing	
jəj	K37. Explain the principles of occupational risk audit and	radiations and other physical agents	
tn tr	management, hazard prevention and emergency preparedness as	using measured occupational dose	
u əu	applied to ionizing radiations (and other physical agents if approp)	data and dose-effect relationships.	
мс uə	associated with the use of medical devices in own area of medical	S26. Carry out a risk audit with respect	
6e	physics practice.	to occupational / public safety from	
o a	K38. Explain relevant international. European. national and local	ionizing radiations and other physical	
w W	legislation. recommendations and documentation regarding risk from	agents in own area of medical physics	
so Xs	ionizing radiations and other physical agents with regard to	practice.	
dx si2	occupational and public safety in own area of medical physics	S27 Evaluate	
а, 1/	practice	facilities/svstems/procedures in terms	
je. Aj	Provident from the activities of institution antimization (institution	of occurational / public pototy from	
e îte	N39. EXplain now the principles of Justinication, optimization (including ALADA) and risk limitation are used for occupational and public	or occupational / public salety from	
) 85			
ш ; Э	protection from the deleterious effects of ionizing radiations and other	agents in own area of medical physics	
ilo	physical agents.	practice.	
na In	K40. Name and explain the function of the main National, European	S28. Assess occupational risks for a	
ac S	and International organizations concerned with protection of workers	given experimental procedure.	
du 8 I	and the general public from ionizing radiations and other physical		
ui i eu	agents (e.g., ICRP, ICNIRP, IAEA).		
oij	K41. Explain how sites and facilities are designed to ensure protection		
sį ec	of workers and the general public.		
<i>ə.</i> Inc	K42. Explain and explain the procedures for the prevention.		
θų po	investigation and handling of adverse incidents with respect to		
μι Ο	workers/public in own area of medical physics practice.		
iət	K43. Explain guantitatively and in detail the interactions with organic		
ļM	matter of ionising and non-ionising electromagnetic radiations.		
)	particulate radiation, ultrasound and electric and magnetic fields at		
	the molecular. cellular. tissue and macroscopic levels in relation to		
	occupational / public risks.		
	K44. Define and measure or calculate the operational quantities		
	(including units and inter-relationships) used in personal dosimetry in		
	own area of medical physics practice (e.g., ambient, directional and		
	personal dose equivalents at recommended depth, annual limit on		
	intake, derived air concentration).		

Annex 1/16

	Knowledge (facts, principles, theories, practices)	Skills (cognitive and practical)	Competence (responsibility and autonomy)
	K45. Explain the possible impact of human factors with regard to occupational / public safety in use of medical devices and associated ionizing radiations and other physical agents.		
	K46. Explain the roles of occupational / public safety personnel associated with ionizing radiations and other physical agents such as Radiation Protection Expert and Radiation Protection Officer as		
(Ju	defined in European, national and local legislation / documentation. K47. Explain the scope, objectives, structure and content of formal systems of work ("local rules").		
∞) tue	K48. Explain in quantitative terms the various means of dose reduction for external radiation (source strengths, exposure times, ristance schedring and internal radiounclides with respect to		
ewe	occupational / public area more reaction account opposed to 0.000 pt or reaction of the state of		
6eue	N+5: State current ucse minus and consulants for workers / public. K50. Explain the process and practical implementation of occup. More and an and a state of the practical abusits practical abusits.		
M Xsi	using techniques for the qualitative and quantitative assessment of risk		
ย/	K51. Explain the key considerations for the design of a new facility		
tety.	(including waiting and resting rooms) with regards to occupational / public safety in own area of medical physics practice.		
eS	K52. Explain the principles and practice of contingency planning and		
plic	the implementation of emergency procedures with respect to occumational / public safety in own area of medical physics practice		
пd	K53. Explain suitable processes for the reporting of radiation incidents		
8 Ie	involving workers / members of the general public in the context of		
enoi	and/or other tools to determine the underlying cause(s).		
teq	K54. Explain the requirements for, and the practical implementation		
nco	u, appropriate systems for the momentation over to the worker, including extremity doses and dose limits for pregnant and		
0	lactating workers, and young workers; and for the public; including		
	selection, management and calibration of devices used to measure such doses, dose records and techniques for dose measurement.		
	K55. Explain how the application of good radiation safety practice and		
	the use of appropriate personal protective equipment minimises		
	worker and public doses in medicine.		
	Noo. Explain the principles factation safety plan design with respect worker / public safety in own area of medical physics practice		
	K57. Explain the functioning of safety systems found in own area of		
	medical physics practice vis-a-vis occupational / public safety.		

Annex 1/17

Clinical Medical Device Management کابماندها Medical Device Management کی کے کی کے کی کے کے کے	 Knowledge (facts, principles, theories, practices) (facts, principles, theories, practices) (facts, principles, theories, practices) (facts, principles, theories, practices) Explain the purpose and practical implementation of formal systems of work (local rules) with regard to safety in own area of medical physics practice. B. Explain statutory and institutional requirements for Medical Physics Services with respect to Clinical Medical device terminology. Define / explain medical devices used in own area of medical physics practice. Define / explain their purpose, modular structure and detailed functioning. Explain the scope and function of national. European and International medical devices bind setting bodies. Explain the Medical Device Directives and associated documentation. Explain the meaning of 'acceptability criteria' as applied to documentation. Explain the function of ICT hardware and software associated accellant the respect to clinical effectiveness and safety, including human-factors. Explain the function of ICT mardware and software associated extenses including firewalls) and systems (e.g., PACS) and data exchange standards used in medicine (e.g., DICOM, DICOM-RT). Include discussions regarding hardware configuration, operating systems. IP discussions regarding hardware configuration, operating systems. IP terminology, port assignment, the tenet, ping testing, network gate/ network propagation, and precautionary measures). 	Skills (cognitive and practical) (cognitive and practical) (cognitive physical / software test objects / phantoms, data acquisition protocols, data recording forms, national / European / international protocols to measure the performance in own area of medical physics, assess deviations from acceptable values (as indicated by manufacturer and international / European / national standard setting bodies), evaluate the relevance of deviations for clinical practice and suggest actions for restoring default performance. S30. Evaluate technical specifications of commercial devices in own area of medical physics practice. S31. Carry out acceptance testing, commissioning and constancy testing procedures in own area of medical physics practice. S32. Adapt national and international and of commercial devices in own area of medical physics practice. S33. Evaluate whether medical devices in own area of medical physics practice.	 Competence (responsibility and autonomy) C27. Take responsibility for statutory and institutional requirements for Medical Physics Services with respect to Clinical Medical Device Management in own area of medical physics practice. C28. Take responsibility for medical device (including software, information systems, PACS) management including planning, evaluation of clinical needs, specification for tender purposes, evaluation of tendered devices, acceptance testing, commissioning, constancy testing (including setting of warming and supement including planning and surveillance, and service contract management in own area of medical physics practice. In the case of acceptance testing this should be done in coortering and surveillance, and service contract C30. Take responsibility for the maintenance of quality control records. C31. Organize infrastructures for display and reading of images and for the reporting and archiving of findings. C32. Organize infrastructures for display and reading of images and for the reporting and archiving of findings. C33. Pursue corrective actions with minimum interference with departmental functionality. C34. Drganize infrastructures for display and reading of images and for the specific activity in own area of medical physics practice.
427 K7 K7	 Explain the operation of data and Data Protection Legislation Explain the operational relationships between hospital information systems (HIS) and information systems specific to own area of medical physics practice (e.g., RIS for imaging). Explain and explain in detail the DICOM standard including its application to own area of medical physics practice. Explain data warehousing for archiving and storage and relevant legislation regarding time such information must be kept. Discuss medical device software standards and types of software licensing. 	service agreements (including software updates) are adequate to ensure service continuity and patient and occupational safety in own area of medical physics practice.	C.35. Take responsibility for the development of an institutional quality assurance / quality control medical device as required by European and national medical device standard setting bodies in own area of medical physics practice. C.36. Take responsibility for the development and ongoing update of departmental quality control protocols for medical devices in own area of medical dical devices in own area of medical dical devices in own area of medical devices in own area of medic

Annex 1/18

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K73. Explain the effects of ionizing radiations and other physical	S34. Analyze the medical devices used	C37. Participate in the installation of new devices in own
	agents on the workings of medical devices in general and in own	in own area of medical physics	area of medical physics practice.
	area of medical physics practice (e.g., electromagnetic interference /	practice and investigate their design,	C38. Negotiate device acceptance with provider and own
	compatibility).	functioning, associated signal / image	department management following acceptance tests
	K74. Define and explain the principles of quality. quality assurance.	processing, safety features, typical	C39. Organize, manage and train guality control teams in
	quality control performance indicators constancy testing quality	specifications and performance	own area of medical physics practice
	control tests. test frequency, tolerances, and action criteria with	indicators.	C40. Decide if actions are required on a medical device to
	respect to medical devices	S35 Design and test physical and	restore default nerformance
	K75 Explain the principles of medical device (including associated	technical methods for quality control of	C41 Define warning and suspension levels for devices
	software) management including planning evaluation of clinical	devices in own area of medical	
(;;	bounded substitution for tender numbers evaluation of tendered	physics practica	
uc	dovince production to tertuci pulposes, evaluation of tertuction dovince productment accontance toeting commissioning	Priyana practuca. 026 Idontifu controle of dovice	
))	devices, procurentent, acceptance testing, continussioning, constancy testing maintanance and decommissioning: service	add. Inctioning sources of device malfunctioning in own area of medical	
ţ		manuncuoming in own area of meanear	
uə		priysics practice.	
w	K/b. Explain the functions of the major international and European	S3/. Autonomously acquire and	
əб	standard (e.g., IEC, CENELEC) setting bodies (and others such as	analyze in detail the literature and user	
e	NEMA) for medical devices and explain the various types of	/ technical manuals for medical	
яr	documentation issued by these bodies and their use in medical	devices in own area of medical	
W	device management.	physics practice.	
ə	K77. Explain in detail international, national and local protocols for	S38. Interpret and apply local	
oi v	assessing the performance of medical devices in own area of	occupational protection rules as	
э	medical physics practice.	applicable to medical device QC	
1	K78 Explain the principles of business planning inventory control	procedures	
ec	And Explain the principles of passives planning, intervely control, subliting honehmorking and handling of controls as applied	C20 Evoluate and narticinate in the	
oil	auduing, benchinalishing and nanuling of service contracts as applied	obs. Evaluate and participate in the	
рə	In medical device management.	selection of medical devices in a	
M	K79. Explain and discuss the main properties of biomaterials relevant	tender in own area of medical physics	
le	to medical device design.	practice.	
oi		S40. Utilize PACS and DICOM in own	
uil		area of medical physics practice.	
c		S41. Apply available systems	
		resources (e.g., RIS, PACS, DICOM	
		data) to QA data elaboration and	
		record.	
		S42. Implement cross-institutional	
		quality control procedures for devices.	
		S43. Perform a documented risk	
		assessment for devices not within	
		susnansion lavals	
		SAA Design mome to accommodate	
		enertific devices in own area of	
		medical physics practice.	

Annex 1/19

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K80. Explain statutory and institutional requirements for Medical	S45. Recognize anatomical /	C42. Take responsibility for statutory and institutional
	Physics Services in own area of medical physics practice with	pathological structures of the human	requirements for Medical Physics Services in own area of
	respect to Clinical Involvement.	body in projection / tomographic and	medical physics practice with respect to Clinical
	K81. Explain the principles of anatomy, physiology, biology (including	3D medical images relevant to own	Involvement.
	radiobiology), pathology as related to the main clinical applications in	area or mearcar priysics practice.	C43. Oversee daily patient safety / risk management
	own area of medical physics practice.	S46. Recognize physiological	involving medical devices and associated ionizing
	K82. Explain trauma / development of diseases, diagnosis, treatment	processes in nuclear / molecular	radiations and other physical agents in own area of
	and follow-up relevant to own area of medical physics practice,	images.	medical physics.
	including primary healthcare and screening programmes.	S47. Participate in clinical discussions	C44. Participate in the evaluation and optimization of
	K83. Explain the International Classification of Diseases (ICD).	within multidisciplinary teams in own	clinical procedures and protocols and risk elimination /
	K84. Explain how medical devices/ ionizing radiations and other	area of medical physics practice.	reduction in own area of medical physics practice in both
	physical agents are used for the solution of clinical problems in own	S48. Participate in the design of patient	routine and non-routine procedures.
	area of medical physics practice.	plans in own area of medical physics	C45. Advise physician in image interpretation and
	K85. Explain the clinical applications and target clinical outcomes in	practice when appropriate.	quantification when appropriate.
	the use of medical devices in own area of medical physics practice.	S49. Adhere to procedures regarding	C46. Take responsibility for semi-quantitative and
ţu	K86. Explain clinical guidelines in own area of medical physics	hygiene.	quantitative data for clinical application.
əu	practice.	S50. Participate in patient preparation	C47. Advise on different patient diagnosis / treatment
uə	K87. Explain the patient's perspective in clinical processes in own	and positioning prior to data	schedule options when appropriate.
^]¢	area of medical physics practice.	acquisition when appropriate.	C48. Participate in the definition of the limits of
	K88. Explain the risk/benefit justification of procedures in own area of	S51. Analyze critically protocol	acceptability of clinical procedures.
uj	medical physics practice.	proposals in terms of feasibility.	C49. Advise on the most appropriate procedure with
le:	K89. Explain protocol optimization principles in own area of medical	effectiveness and safety.	respect to risk/benefit ratio.
oir	physics practice.	S52. Handle and analyze medical	C50. Supervise procedures for paediatric investigations in
iil:	K90. Explain the design principles, the relevant legislation issues and	images including the extraction of	relation to dose optimization.
c	approval procedures for clinical trials.	parametric data / images.	C51. Advise other healthcare professionals on
	K91. Explain the principles and implementation of Good Clinical	S53. Set up devices, experiments and	optimization and safety of individual patient examination /
	Practice (GCP), Good Manufacturing Practice (GMP) and Good	protocols for the measurement of	treatment and examination / treatment protocols.
	Laboratory Practice (GLP) in own area of medical physics practice.	physical variables relevant to clinical	C52. Live up to demands imposed by duty of
	K92. Explain general indications and contra-indications for the use of	practice.	confidentiality, professional secrecy, ethical standards.
	devices in own area of medical physics practice.	S54. Operate medical devices in own	C53. Represent medical physics in clinical conferences.
	K93. Understand the nature of anatomical/ pathological medical	area of medical physics practice	C54. Take responsibility for the prevention, investigation
	images as the visualization of the 3D distribution of physical	effectively and safely.	and handling of adverse incidents (including use of Root
	variables.		Cause Analysis / Failure Modes and Effects Analysis or
	K94. Survey the main sources of evidence from within the general		alternative methods; recommendations of appropriate
	physics, medical physics and general healthcare (e.g., the Cochrane		remedial actions) with respect to patients in own area of
	Collaboration) literature essential for the carrying out of a systematic		medical physics practice.
	survey in own area of medical physics practice.		
	K95. Explain concepts in health informatics such as unique patient		
	identifier, medical record and disease coding (e.g., ICD10).		
	K96. Explain safety and risk related issues associated with the use of		
	ICT in own area of modical physics practice		_

Annex 1/20

|--|

Annex 1/21

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K102. Explain statutory and institutional requirements for Medical	S55. Participate in development of	C55. Take responsibility for statutory and institutional
	Physics Services with respect to development of Service Quality and	service quality and cost-effectiveness	requirements for Medical Physics Services with respect
	Cost-effectiveness in own area of medical physics practice.	in own area of medical physics	to Development of Service Quality and Cost-
	K103 Explain the principles of business strategic planning and cost	practice.	Effectiveness in own area of medical physics practice,
S	effectiveness in the case of Medical Physics Services	S56. Define quality objectives in own	whilst being aware that improvement of the service as a
ອເ	K104. Define and explain the principles of quality. continuous quality	area of medical physics practice.	whole depends on the inputs of other healthcare
19/	improvement, quality audit and total quality management systems as	S57. Define, measure and optimize	protessionals.
٧ij	applied to aspects of clinical audits involving medical devices and	appropriate quality indicators in own	C56. Advise on the technical aspects impacting the clinical
зə	associated ionizing radiations and other physical agents.	area of medical physics practice.	effectiveness and safety of new medical devices or
11 Ξ	K105. Explain why the holistic development of a service depends on	S58. Set up a service quality	techniques prior to their introduction into clinical practice.
3-3	the quality assurance of the parts.	development strategy for own area of	C57. Participate in the design and implementation of QA
so	K106. Explain why the development of service quality for an area of	medical physics practice.	systems in own area of medical physics practice.
С	medical practice requires input from various healthcare	S59. Prepare a business and strategic	C58. Take responsibility for using the methodologies of
pu	professionals.	plan for the development of Medical	Evidence Based Medicine to investigate ways of
IP .	K107. Explain responsibilities of other healthcare professionals	Physics Services in own area of	improving service quality within own area of medical
ity	involved in QA activities in own area of medical physics practice.	medical physics practice.	physics practice.
lei	K108. Explain the intentions and principles of QA systems like ISO	S60. Apply the principles of business,	C59. Assume responsibility for quality management audits
5	9000 and formal systems for external accreditation by	strategic planning and cost	involving medical devices and associated ionizing
) Ə	expert/professional bodies.	effectiveness in own area of medical	radiations and other physical agents.
٥i	K109. Define quality objectives in own area of medical physics practice.	physics practice.	C60. Take responsibility for the design and
AL6	K110. Explain the institutional framework of QA activity and regulation	S61. Set up and continuously develop a	implementation of a monitoring system for Medical
s	in own area of medical physics practice.	feedback system for ongoing	Physics Services in own area of medical physics
ło	K111. Explain the functions of the major International and European	improvement of quality (based on	practice.
JL.	standard setting bodies for healthcare quality; explain the various	assessment of non-conformities and	C61. Take responsibility for the development and
191	types of documentation issued by these bodies and explain its use	accident analysis) in own area of	implementation of a business and strategy plan for
uc	for service quality development.	medical physics practice.	Medical Physics Services in own area of medical physics
lo	K112. Explain the principles of Evidence Based Medicine and explain	S62. Apply available resources (such	practice.
ə/	how the evidence base can be used to improve service quality.	as those in RIS/PACS systems) to	C62. Take responsibility for the formal review and analysis
(ə(K113. Explain the purpose and implementation of local systems for	elaboration and recording of quality	of incidents within own area of medical physics practice.
3	formal incident reporting and internal review with regard to	related data.	
	improvement of service quality.	S63. Measure quality management	
		performance and improvements in	
		own area of medical physics practice.	
		S64. Participate in the reporting, review	
		and analysis of incidents	

Annex 1/22

	(facts, principles, theories, practices)	SKIIIS (cognitive and practical)	Competence (responsibility and autonomy)
	K114. Explain statutory and institutional requirements for Medical Physics Services in own area of medical physics practice with respect to Expert Consultancy.	S65. Apply MPE consultancy skills to specific scenarios in own area of medical physics practice.	C63. Take responsibility for statutory and institutional requirements for Medical Physics Services in own area of medical physics practice with respect to Expert
	K115. Explain the role of a consultant. K116. Explain the role of scientists as consultants in healthcare.	S66. Identify and manage ethical issues involved in delivering a	Consultancy including responsibility for associated ethical issues commensurate with level of personal expertise.
	K11/. Explain the general role of the MPE as consultant in own area of medical physics practice.	consultancy service in own area of medical physics practice (including	C64. Produce and/or audit reports as an independent provider for organizations other than one's own.
	K118. Discuss the specific ethical issues involved in delivering a consultancy service in own area of medical physics practice (including conflict of interest issues).	conflict of interest issues).	C65. Design and evaluate continuous professional courses in own area of medical physics practice for organizations other than one's own.
1	K119. Explain statutory and institutional requirements for Medical Physics Services with respect to the education and training of healthcare professionals (including Medical Physics trainees) in own are of medical physics practice. K120. Discuss the principles of modern adult pedagogy and apply them to the case of healthcare professionals. K121. Discuss the principles of modern adult pedagogy and apply them to the medical device and ionizing radiations and other physical agents educational needs of healthcare professionals (including continuous professional development activities) and including training associated with the introduction of new devices and techniques. K122. Discuss methods for developing and delivering ionizing radiations and other physical agents educational is specific clinical environments. K123. Discuss the factors which impact the choice of learning outcomes for addressing the learning meds of specific healthcare professionals in specific clinical environments. K123. Discuss the factors which impact the choice of learning outcomes for addressing the learning and the usability and safety features of devices). K124. Explain the content of appropriate programmes for healthcare professionals in specific clinical anytonements (such as previous education and training and the usability and safety features of devices).	 S67. Set up an inventory of learning outcomes tailored to the specific healthcare professionals in specific clinical environments in conjunction with the leaders of the respective healthcare professions. S68. Prepare effective and efficient modes of knowledge transfer activities specific to the specific claming needs of specific thealthcare professionals in specific clinical environments in conjunction with the leaders of the respective healthcare professions. S69. Prepare effective modes of the various healthcare professions. S69. Prepare effective modes of the various healthcare professions. S70. Carry out own pedagogical research when the evidence base for decration and training of healthcare professions. 	 C66. Take responsibility for statutory and institutional requirements for Medical Physics Services in own area of medical physics practice with respect to the Education (including continuous professional development) of Heatthcare Professionals (including Medical Physics trainees). C67. Take responsibility for the education of heatthcare professionals (including Medical Physics trainees) regarding the optimised clinical use of medical olevices and safety from ionizing radiations and other physics lagents in specific clinical environments in own area of medical physics practice. C68. Take responsibility for the education of heatthcare professionals (including Medical Physics trainees) in performing QC procedures related to medical devices in own area of medical physics practice. C69. Take responsibility for the education of heatthcare professionals (including Medical Physics trainees) in performing QC procedures related to medical devices in own area of medical physics practice. C69. Take responsibility for the education of heatthcare professionals (including the use of performing QC procedures related to medical devices in own area of monitors and personal protection equipment. C70. In conjunction with other heatthcare professionals take responsibility for raising public awareness of safety issues regording the use of personal dose monitors and personal protection equipment.

Annex 1/23

	Knowledge	Skills	Competence
	(facts, principles, theories, practices)	(cognitive and practical)	(responsibility and autonomy)
	K125. Explain statutory and institutional requirements for Medical	S71. Perform a systematic review of	C72. Take responsibility for statutory and institutional
	Physics Services in own area of medical physics practice with	the existing evidence base to evaluate	requirements for Medical Physics Services in own area c
	respect to Health Technology Assessment (HTA).	the clinical effectiveness and safety of	medical physics practice with respect to Health
	K136 Evolain the principles of HTA as applied to medical devices and	a new medical device or new	Technology Assessment (HTA).
	procedures in own area of medical physics practice.	procedure involving medical devices /	C73. Use the methodologies of HTA to carry out a HTA in
	K127. Explain the steps for the carrying out a HTA, including use of	ionizing radiauons and other priysical	conjunction with other healthcare professionals.
	primary data and secondary sources.	agents.	C74. Take responsibility for the technical component of a
1	K128. Define the roles and responsibilities of all professionals involved	S72. Communicate HTA reports to	HTA related to medical devices and /or ionizing
ue	in an HTA project in own area of medical physics practice.	policy makers.	radiations and other physical agents.
u	K129. Explain the issues that should be considered in an HTA project	S73. Interpret the statutory and	C75. Take responsibility for the technical component of a
SS	in own area of medical physics practice.	institutional requirements of Medical	clinical trial related to medical devices and /or ionizing
98	K130. Explain the value of HTA reports for policy makers at the	Physics Services in HTA activities.	radiations and other physical agents.
ss	European, national, regional and facility levels.	S74. Design and monitor the medical	C76. Take responsibility and communicate with relevant
۲ N	K131. Explain the importance of HTA reports in controlling cost in	physics components of clinical trial	authorities with regards to safety from ionizing radiations
6	relation to benefit for the considered technology in own area of	protocols in own area of medical	and other physical agents in the case of clinical trials.
ojo	medical physics practice.	physics practice.	C77. Apply for approval from a hospital and /or university
bu	K132. Apply research methodologies and statistical techniques used at	S75. Perform statistical analysis and	based ethics committee for a clinical trial involving
чэ	the interface between physical and biomedical science in clinical	report on clinical trials involving	medical devices and /or ionizing radiations and other
еŢ	trials involving medical devices and/or ionizing radiations and other	medical physics services.	physical agents.
. ч	physical agents.	S76. Assemble a suitable physics team	C78. Take responsibility for the evaluation of a clinical tria
tle	K133. Discuss the ethical issues associated with clinical trials involving	for a specific HTA project.	protocol.
əŀ	medical devices and/or ionizing radiations and other physical agents.	S77. Conduct the technical	C79. Ensure good clinical practice (GCP) compliance of
4	K134. Explain how to apply for approval from a hospital and /or	components of an HTA project in own	activities within clinical trials.
	university based ethics committee for a clinical trial involving medical	area of medical physics practice.	C80. Advise on and take responsibility for the preclinical
	devices and /or ionizing radiations and other physical agents.		device aspects of the ethical review of a clinical trial.
	K135. Explain the fundamentals and design models of clinical trials in		C81. Assume the responsibility of statistical and other
	own area of medical physics practice.		mathematical data processing and recording in a clinical
			trial.

Annex 1/25

Appendix C

Competences of a MPP specific to the specialties of Radiation therapy, Nuclear Medicine, and Diagnostic and Interventional Radiology as specified by IAEA

PHY SICIS IS SPECIFIC AND INTERVENTION	OLD THE SPECIALITES OF KADI AL RADIOLOGY	ATION THERAPY, NUCLEAR M	EDICINE, AND DIAGNOSTIC
Area of responsibility	Radiation therapy	Nuclear medicine	Diagnostic and interventional radiology
 (a) Installation design, technical specification, acceptance, commissioning and maintenance of (i) (ii) 	 (i) Be an essential part of the team for shielding and installation design of new or modified radiation therapy rooms, ensuring that safety requirements are complied with; (ii) Lead the development of equipment specifications; (iii) Have responsibility for the acceptance and commissioning of equipment, including radiation therapy treatment and imaging units, brachytherapy sources and treatment planning systems; (iv) Provide advice on equipment decommissioning. 	 (i) Be an essential part of the team for the shielding and installation design of new or modified facilities, ensuring that safety requirements are complied with; (ii) Lead the development of equipment specifications; (iii) Have responsibility for the acceptance and commissioning of equipment; (iv) Provide advice on equipment decommissioning. 	 (i) Be an essential part of the team for the shielding and installation design of new or modified facilities, ensuring that safety requirements are complied with; (ii) Lead the development of equipment specifications; (iii) Have responsibility for the acceptance and commissioning of equipment; (iv) Provide advice on equipment decommissioning.

SUMMARY OF THE ROLES AND RESPONSIBILITIES OF CLINICALLY QUALIFIED MEDICAL 4 TABLE

PHYSICISTS SPECI	FIC TO THE SPECIALTIES OF RAL	JIATION THERAPY, NUCLEAR N	AEDICINE, AND DIAGNOSTIC
AND INTERVENTIC	JNAL KADIULUGY (cont.)		
Area of responsibility	Radiation therapy	Nuclear medicine	Diagnostic and interventional radiology
(b) Radiation safety and protection of patients, staff and the general public	 (i) Develop the clinical radiation safety programme for radiation protection of patients, staff and the public; (ii) Participate in the investigation of radiation incidents and accidents; (iii) Develop procedures for verifying the integrity, safe operation and use 	 (i) Develop the clinical radiation safety programme for radiation protection of patients, staff and the public; (ii) Participate in the investigation of radiation incidents and accidents; (iii) Develop procedures for 	 (i) Develop the clinical radiation safety programme for radiation protection of patients, staff and the public; (ii) Participate in the investigation of radiation incidents and accidents; (iii) Develop procedures for

12

operation and use of diagnostic and interventional radiology

operation and use of nuclear verifying the integrity, safe

medicine equipment and

radioactive sources.

equipment and accessories.

verifying the integrity, safe

of radiation therapy equipment and

accessories.

AND INTERVENTION	NAL RADIOLOGY (cont.)		
Area of responsibility	Radiation therapy	Nuclear medicine	Diagnostic and interventional radiology
(c) Patient radiation dosimetry	 (i) Acquire the data needed for the clinical use of treatment units (part of the acceptance and commissioning process for entry into service); (ii) Develop tables of data for clinical use; (iii) Establish and perform procedures for patient dose calculation and verification; (iv) Have overall responsibility for the treatment planning calculations; (v) Perform patient dose verifications including in vivo dosimetry. 	 (i) Perform activity measurements and calculation of the dose received by different organs following the administration of radiopharmaceuticals in the various clinical procedures; (ii) Perform patient specific dose calculations, establishing tolerances. 	 Establish procedures for estimating the absorbed dose in patients during different clinical procedures; Perform patient specific dose calculations, establishing tolerances; Perform patient dose estimations to establish diagnostic reference levels, or to verify conformity with reference levels.

TABLE 2. SUMMARY OF THE ROLES AND RESPONSIBILITIES OF CLINICALLY QUALIFIED MEDICAL

TABLE 2. SUMMA PHYSICISTS SPECIF AND INTERVENTIO	LRY OF THE ROLES AND RES FIC TO THE SPECIALTIES OF RADI NAL RADIOLOGY (cont.)	SPONSIBILITIES OF CLINICA. IATION THERAPY, NUCLEAR M	LLY QUALIFIED MEDICAL EDICINE, AND DIAGNOSTIC
Area of responsibility	Radiation therapy	Nuclear medicine	Diagnostic and interventional radiology
(d) Optimization of physical aspects of medical procedures	 Optimize the treatment planning process, including image acquisition and treatment delivery; Develop a quality management programme for radiation therapy imaging, dose calculation and treatment delivery systems. 	 (i) Optimize data acquisition processes and procedures to improve image quality while minimizing radiation dose to patients; (ii) Assist nuclear medicine medical practitioners in evaluating examination efficacy and in image quality and perception studies. 	 (i) Optimize data acquisition techniques and procedures to improve image quality while minimizing radiation dose to patients; (ii) Assist diagnostic and interventional radiology medical specialists in evaluating examination efficacy and in image quality and perception studies.

14

PHYSICISTS SPECII AND INTERVENTIO	IC TO THE SPECIALTIES OF RAD NAL RADIOLOGY (cont.)	IATION THERAPY, NUCLEAR N	IEDICINE, AND DIAGNOSTIC
Area of responsibility	Radiation therapy	Nuclear medicine	Diagnostic and interventional radiology
 Quality management of the physical and technical aspects 	 Participate as a team member in designing and implementing a quality management programme, being responsible for: (i) Developing institutional policies and procedures related to the use of radiation; (ii) Establishing and implementing a quality assurance programme for treatment units, treatment planning systems, dosimetry equipment and radiation therapy imaging equipment; (iii) Calibrating radiation generators and brachytherapy sources according to a well established code of practice; (iv) Performing risk assessment, identifying potential radiation exposures, and developing action procedures for such events. 	Participate as a team member in designing and implementing a quality management programme, being responsible for: (i) Developing institutional policies and procedures for the continuous optimization of radiation use; (ii) Establishing and implementing a quality assurance programme with appropriate elements for the handling and measurement of radioactive sources and regulatory compliance of imaging and dosimetry equipment; (iii) Performing risk assessment, identifying potential radiation exposures and developing action procedures for such events; (iv) Investigating unintended or accidental medical exposures.	Participate as a team member in designing and implementing a quality management programme, being responsible for: (i) Developing institutional policies and procedures for the continuous optimization of radiation use; (ii) Developing and implementing procedures for the initial and continuing evaluation of the imaging and associated equipment, as well as calibration of patient dosimetry equipment; (iii) Calibrating X ray units according to a well established code of practice; (iv) Ensuring compliance of imaging radiation equipment with government and accreditation agency regulations and recommendations.
	מרכותבתומו וווכתורמו בעלהסמיריס.	מכנותבווומו וזובתוכמו בעלהמתבס.	

TABLE 2. SUMMARY OF THE ROLES AND RESPONSIBILITIES OF CLINICALLY QUALIFIED MEDICAL

Appendix C. Competences of a MPP specific to the specialties of Radiation therapy, Nuclear Medicine, and Diagnostic and Interventional Radiology as specified by IAEA0

AND INTERVENTIC	NAL RADIOLOGY (cont.)		
Area of responsibility	Radiation therapy	Nuclear medicine	Diagnostic and interventional radiology
(f) Collaboration with other clinical professionals as key team members	 (i) Provide consultation to radiation oncology medical practitioners to establish optimal treatment technique; (ii) Supervise technologists in the implementation of new clinical procedures, including assistance in set-up and correct treatment delivery. 	 (i) Provide consultation to nuclear medicine medical practitioners on special cases of diagnostic exploration or treatment and assist to establish the optimized approach for each case; (ii) Assist to introduce new clinical procedures, develop methods for their quality assurance and control, and supervise their implementation. 	 Provide consultation to diagnostic and interventional radiology medical practitioners on special cases of diagnostic or interventional procedures and assist to establish the optimized approach for each case; (ii) Assist to introduce new clinical procedures, develop methods for their quality assurance and control, and supervise their implementation.

SUMMARY OF THE ROLES AND RESPONSIBILITIES OF CLINICALLY QUALIFIED MEDICAL 2 TABLE

16

Appendix D

Occupational and Public Radiation Protection as specified by IAEA

(ii) Supervision of technologists in the implementation of new clinical procedures: Medical physicists are responsible for supervising the introduction of new clinical imaging procedures in their institution. They are also responsible for developing methods for QC for the new procedures.

4.3. OCCUPATIONAL AND PUBLIC RADIATION PROTECTION

The BSS [3] assign specific responsibilities to the medical physicist for medical exposures and the patient's radiation protection, both intrinsically related to therapeutic and diagnostic procedures using ionizing radiation discussed in previous sections. The BSS [3] also introduce the term 'radiation protection officer' (RPO) for:

"A person technically competent in radiation protection matters relevant for a given type of practice who is designated by the registrant, licensee or employer to oversee the application of relevant requirements."

In many clinical environments, medical physicists have responsibilities not only for the safety of the patient, but also for the protection of the staff and the public, as well as for the safety of radioactive sources. As stated jointly by the IOMP and the International Radiation Protection Association [10], all medical physicists receive adequate training in radiation protection and, as part of their assigned duties, many act as RPOs in health facilities and/or participate as members of the radiation safety committee. One of the specialties of medical physicists trained in this specialty and working mostly in large teaching hospitals undertake the role of RPOs, which may require a higher level of training and more experience in radiation protection.

The main roles and responsibilities of CQMPs in radiation protection in the workplace and for the public are described in this section. They are classified into two major areas, namely safety of the staff and the public, and safety of radioactive sources.

4.3.1. Safety of the staff and the public

(a) Installation design, technical specification, acceptance and commissioning of equipment, including the establishment of criteria for acceptable performance: CQMPs collaborate in the shielding design of installations and ensure compliance with safety requirements, classify work areas into supervised and controlled areas, help to develop and define the technical
specifications for the purchase of new equipment for radiation protection and safety inspections, and develop procedures for the initial and continued evaluation of such equipment. They advise on practical methods to reduce the dose to the staff and the public who work or are in areas adjacent to rooms where radiation equipment is installed or radioactive sources used. In addition, they are responsible for designating the areas in which pregnant, occupationally-exposed women may not work. They also perform calculations and surveys to verify the adequacy of the existing shielding in these rooms using their relevant dimensions, occupancy factors and workload, and establish criteria for the access to rooms that are controlled, with limited access to members of the public, supervising their implementation. CQMPs are also responsible for supervising the installation of new radiation protection equipment and for performing acceptance testing and commissioning of such equipment, including related computer systems, their algorithms, data and results.

- (b) Radiation safety programme for the protection of staff and the general public: CQMPs have responsibilities in the development and implementation of a clinical radiation safety programme for the hospital, including policies and procedures for the radiological protection and safety of the workers and the public. CQMPs carry out hazard assessment of the facilities and procedures, and establish whether the existing procedures are adequate (taking into account the type of radiation sources), the dose rates to which staff members may be exposed, the results of personnel dosimetry obtained for similar activities, control measures in place, and the need for personal protection devices, such as lead aprons, thyroid shields, goggles and other devices. They ensure that the latter are correctly used and tested periodically for integrity. CQMPs have responsibilities in establishing policies and procedures for the safe transport of radioactive material, for precautions in cases of contamination or spillage of unsealed radionuclides, for the management of radioactive waste and for the integrity and proper operation of survey meters and other measuring equipment, as required by the regulations
- (c) Radiation dosimetry: CQMPs organize and provide personnel dosimetry and monitoring systems at a local level, following the local legislative procedures. Records of personnel dosimetry and of estimated doses received by members of the public are maintained for the period of time specified in national regulations. They also have responsibilities in investigating anomalous exposures and determine whether any radiological hazard is present, and if so to what extent, particularly when hazards result from gamma rays emitted by radioactive sources or from ionizing radiation emitted by equipment used for diagnosis or treatment. They develop

procedures and contingency plans to deal with unintended or accidental exposures, and make recommendations on actions required to minimize the likelihood of such unintended exposures happening again. In addition, they provide the required surveys, interpret their results and evaluate compliance with the appropriate regulatory bodies.

- (d) Optimization of the physical and technical aspects of radiation safety procedures: CQMPs carry out radiation protection and safety audits, and identify whether appropriate licences exist. They ensure that radiation areas are properly designated and warning signs are in place, and that radiological checks of the working areas provide evidence for compliance with the existing radiation protection policies and procedures, as well as with regulatory and accreditation agency rules and recommendations.
- (e) Quality management of the physical and technical aspects of radiation safety equipment: CQMPs have responsibilities in developing, implementing and supervising the physical aspects of the quality management programme for the equipment used for radiation protection of the staff and the public. Related tasks include:
 - (i) Developing institutional policies and procedures for the safe and optimal use of equipment used for radiation detection and measurement.
 - (ii) Establishing QA programmes and performing QC for all of the radiation protection equipment: CQMPs are responsible for the selection, periodic calibration and QC processes used to establish the correct operation of equipment used for radiation surveys, as well as for the associated systems used for environmental monitoring, as required by the regulations.
 - (iii) Carrying out risk assessment and management: CQMPs are responsible for the integrity of the survey meters and other equipment used to measure the radiation levels to which staff and/or the public is exposed, and for establishing methods to minimize the radiation dose to the staff and the public, thus reducing the associated risks.
- (f) Collaboration with other clinical professionals: CQMPs work with other clinical professionals, including medical practitioners, technologists and nursing staff, on special cases that may be encountered in the clinical environment and may compromise the safety of the staff and the public, e.g. an accident during the transport of radioactive materials. In such cases, the members of the public and/or the driver that may be involved, injured or contaminated are brought to the hospital for observation and possible treatment. The role of the CQMP, acting in most cases as RPO, is essential in providing instructions on the actions to be taken for triage and decontamination of such victims without spreading the contamination,

and to prevent unnecessary radiation exposure of staff members and other patients during treatment. Collaboration between medical practitioners and medical physicists helps to establish the optimal approach for each case, and the CQMP provides advice on safety for the best and safest outcome of such special situations.

(g) Education and training: CQMPs provide education and continuous training to clinical staff on radiation safety and radiological protection. They ensure that training programmes are in place and deliver lectures and practical training to staff on the basic principles of radiation safety, including the classification of controlled and supervised areas, the expected exposure resulting from different diagnostic or therapeutic procedures using radiation sources, and establish and promote a safety culture and the concept of defence in depth.

4.3.2. Safety of radioactive sources

- (a) CQMPs establish procedures for the safe transport of radioactive sources and equipment that emit or use radiation within the hospital complex, taking into account all regulatory requirements and safety considerations. This includes transfer of ownership of sources during delivery or disposal.
- (b) CQMPs develop a programme of physical security for radioactive sources, including procedures for receiving, storing securely, stock-taking and controlling their fixed or temporary location at the hospital. They plan and supervise regular inventories of all of the radioactive sources, ensuring their safe disposal as radioactive waste when relevant, according to national and international safety regulations and recommendations.
- (c) CQMPs perform risk assessments and identify possible accidents or losses of radioactive sources, develop action procedures to be followed in the event of such occurrences and carry out exercises to verify that they can be implemented correctly.

5. STAFFING AND ORGANIZATION OF A MEDICAL PHYSICS SERVICE

Staffing requirements for providing medical physics services supporting the efficient and safe care of patients need to be defined using criteria consistent with current medical practice. The impact of the continuous development of medical technology and its applications, as well as changes in the regulatory requirements worldwide, make most previous recommendations on staffing outdated. Although **Appendix E**

Subsidiary Legislation 585.01, 2018, Regulation 107

BASIC SAFETY STANDARDS FOR IONISING RADIATION [S.L.585.01 69

undertaking the performance of the tasks of radiation protection of workers and members of the public.

(5) Where a radiation protection expert is consulted pursuant to the requirements of these regulations the undertaking shall appoint that radiation protection expert in writing and shall include in that appointment the scope of the advice which the radiation protection expert is required to give.

106. (1) Radiation protection experts need to be recognized Radiation Protection Experts. by the Commission and will need to meet the following generic criteria:

(a) have a degree in radiography, medicine, or other physical science or a suitable combination of other qualifications and experience; and

sufficient work experience directly concerned (b) with radiation protection practice; and

undertake continued professional development. (c)

The Commission shall issue detailed requirements for (2)different applications based on the generic criteria given in the above.

(3) Radiation protection experts shall need re-certification by the Commission.

experts.

Medical physics experts are to act or give Medical physics 107. (1) specialist advice, as appropriate, on matters relating to radiation physics for implementing the requirements set out in Part VII and in paragraph (b) of regulation 31(1).

(2) Depending on the medical radiological practice, the medical physics expert shall take responsibility for dosimetry, including physical measurements for evaluation of the dose delivered to the patient and other individuals subject to medical exposure, give advice on medical radiological equipment, and contribute in particular to the following:

- radiation protection programme; (a)
- (b) development of safety culture;

optimisation of the radiation protection of (c) patients and other individuals subject to medical exposure, including the application and use of diagnostic reference levels:

the definition and performance of quality (d) assurance programmes of the medical radiological equipment;

(e) testing of medical radiological equipment; 70 [S.L.585.01 BASIC SAFETY STANDARDS FOR IONISING RADIATION

 (f) the preparation of technical specifications for medical radiological equipment and installation design;

(g) the surveillance of the medical radiological installations;

 (h) the analysis of events involving, or potentially involving, accidental or unintended medical exposures;

 the selection of equipment required to perform radiation protection measurements;

 the training of practitioners and other staff in relevant aspects of radiation protection.

(3) The medical physics expert shall, where appropriate, liaise with the radiation protection expert.

108. The Commission shall issue approval certificates for medical physics experts for a validity period of not greater than a period of five years in a specialty (radiotherapy, nuclear medicine, diagnostic and interventional radiology)

109. (1) Medical physics experts shall meet the following criteria for initial certification by the Commission:

> (a) be registered by the Council for the Professions complimentary to Medicine in Malta as a medical physicist; and

> (b) have attained European Qualifications Framework (EQF) Level 7 (ex: Master with 90-120 European Credit Transfer and Accumulation System (ECTS)); and

> two years structured accredited clinical training in Medical Physics in specialty (radiotherapy, nuclear medicine, diagnostic and interventional radiology); and

> (d) two years documented full time work experience, within the preceding 28 months, in specialty (radiotherapy, nuclear medicine, diagnostic and interventional radiology) with documented CPD.

(2) Medical physics experts shall meet the following criteria for re-certification by the Commission:

(a) 48 months full time work experience in their specialty during the valid approval period, but not necessary 48 consecutive months.

(b) Documented CPD since the last certificate issued.

Radiation 110. (1) All undertakings and employers of outside protection officer. workers shall designate a radiation protection officer.

Approval certificates.

Meetings.

Appendix F

Characteristics that suggest the need for strategic planning

Perera, et al., (2012) outlined five characteristics that when combined, suggest the need for strategic planning. These are:

- a) Clientele who are more aware and demanding: Assuming patients are the profession's only direct or indirect clients, it stands to reason that in the years ahead, the profession will be caring for patients who are more informed and more capable of selecting their healthcare provider. This is a choice that has implications for funding.
- b) Increasing numbers of competent and professional rivals: On a global scale, it is becoming easier to locate highly trained experts; the high degree of specialisation and quality across these professionals is becoming more common
- c) Limited organisational resources: The increasing responsibility to spend resources rationally, enabling only the finest and most effective professionals to persist, is one potential reaction to this circumstance.
- d) The client is given more attention than the service: Organisations prioritise on the patient receiving a good service and getting the greatest outcomes rather than just performing tasks as efficiently as feasible. It is not only about what, but also about how a task should be performed.
- e) Size and intricacy of the organisation: The population has grown in number, meaning that the requirements have also grown, and there are now more diagnostic and treatment choices available, which has caused organisations to physically expand in size and organisational complexity to rise.

Appendix G

List of processes for creating a strategic plan by using SWOT methodology

The following list outlines the processes for creating a strategic plan using the SWOT methodology (Caruana, 2020):

- a. Choose the values that will guide the activity in strategic planning.
- b. Create or revise the present strategic mission statement.
- c. While keeping the purpose of the vision in mind, create a desirable future vision for the profession.
- d. Do a gap analysis to determine the gaps between the existing condition of the profession and the anticipated future vision.
- e. Conduct a situational analysis of the profession's SWOTs in relation to the desired vision.
- Determine and rank a number of strategic objectives for bridging the gaps and moving closer to the desired vision.
- g. Create a thorough strategic action plan to accomplish each of the strategic goals.
- h. Carry out the plan.
- i. Control and evaluate the plan and iterate if necessary.

Appendix H

Five points to follow for the development of effective questions

The following list provides the criteria to follow for the development of effective questions as outlined by Floyd J. Fowler, Jr. (1995):

- a) Detailing the goals of the question and outlining the types of responses that are necessary to attain those goals.
- b) Ensuring that each participant has a similar understanding of the question's context. All respondents should be able to understand the question's key terms and have the same perspective of those notions.
- c) Ensuring that participants are asked questions for which they have the answers. Finding the solutions is impacted by at least three different sorts of challenges.:
 - · Not possessing the information required to address the question.
 - Possessing the information but not being able to recall it precisely or to the level of specificity required by the question.
 - The difficulty of precisely dating events for questions concerning interactions or occurrences during a certain time period.
- d) Presenting questions that respondents can respond to in the way that the question specifies. It is possible to ask participants queries to which they know the answer, but they are unable to respond in the manner the researcher desires because the researcher's preferences do not align with the reality that the participant is expressing.
- e) Asking queries that respondents would not object in responding. This is required to gather responses that can be combined to produce statistical information. The question must also be able to be stated effectively so that every person would understand what it meant.

Appendix I

Questionnaire

Questionnaire

Please mark the boxes with a \blacksquare as appropriate.

Section 1: General Questions

1	1.1 Which gen Male Female Other	der do you identify with?			
1	1.2 Which und Physics Electrical I Mechanica	lergraduate degree did you pursu Engineering 1 Engineering	e?		
1	1.3 Which area Radiation (Diagnostic Nuclear M	a do you specialise in? Oncology and Interventional Radiology edicine			
1	1.4 Are you ce and Non-Io Yes No	ertified as a Medical Physics Ex onising Radiation? □ □	pert by	the Commission for Protection	from Ionising
1	1.5 Are you o Ionising an Yes No	certified as a Radiation Protect nd Non-Ionising Radiation?	ion Exp	pert by the Commission for Pr	otection from
1	1.6 Are you ce Yes No	artified as MR Safety Expert (MR □ □	SE) by a	any accredited organisation?	
1	1.7 Are you ce Yes No	rtified as Laser Safety expert by □ □	any accr	redited organisation?	
1	1.8 Medical pl apply): Ionising ra Non-ionisi All medica None of th	hysicists in Malta are involved diation based medical devices ng radiation based medical devic 1 devices e above	in the sp es	pecification and procurement of	(tick all that
1	1.9 Medical ph Ionising ra Non-ionisi All medica None of th	nysicists in Malta are involved in diation based medical devices ng radiation based medical devic 1 devices e above	accepta: es	nce testing of (tick all that apply):

1.10 Ion No All No	Medical physicists in Malta are involved in the nising radiation based medical devices n-ionising radiation based medical devices medical devices ne of the above	commissioning of (tick all that apply):
1.11 Ion No All No	Medical physicists in Malta carry out quality as using radiation based medical devices n-ionising radiation based medical devices medical devices ne of the above	ssurance of (tick all that apply):
1.12 Ion No All No	Medical physicists in Malta are involved in the hising radiation based medical devices n-ionising radiation based medical devices medical devices ne of the above	decommissioning of (tick all that apply):
1.13 Ion No All No	Medical physicists in Malta are involved in opt ising radiation based medical devices n-ionising radiation based medical devices medical devices me of the above	imising clinical use of (tick all that apply):
1.14 ass (tic Ion No All No	Medical physicists in Malta provide advice sociated physical agents including protection fro the all that apply): using radiation n-ionising radiation l physical agents ne of the above	regarding patient risks from medical device m such physical agents in the following cases
1.15 (tic Ion No All No	Medical physicists in Malta are involved in t is all that apply): hising radiation based medical devices n-ionising radiation based medical devices medical devices ne of the above	he design and installation of new facilities of
1.16 acc Ion No All No	Medical physicists in Malta provide advice cidental exposures to physical agents from (tick a hising radiation based medical devices n-ionising radiation based medical devices medical devices ne of the above	regarding the prevention of unintended or III that apply):

- 1.17 The medical physics profession in Malta is seen by stakeholders as contributing significantly to the over-arching mission of the wider healthcare system of which it forms part:
 - Strongly Agree
 □

 Agree
 □

 Not Sure
 □

 Disagree
 □

 Strongly Disagree
 □
- 1.18 I feel that I am aware of what is internationally considered best practice in my specialty area of Medical Physics:
 - Strongly Agree
 □

 Agree
 □

 Not Sure
 □

 Disagree
 □

 Strongly Disagree
 □
- 1.19 There has never been a shortage of material resources that are required to enable me to deliver my professional services to the healthcare system:
 - Strongly Agree
 □

 Agree
 □

 Not Sure
 □

 Disagree
 □

 Strongly Disagree
 □
- 1.20 What do you believe are the specific unique benefits the Medical Physics profession in Malta provides and to whom (patients, other healthcare professionals, hospital management)? Please explain.

Section 2: Possible Strengths of the MP profession in Malta

2.1 Medical Physicists in Malta have a deep techno-scientific expertise regarding medical devices used in their specialty and their clinical use:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

2.2 Medical Physicists in Malta have a deep techno-scientific expertise regarding the protection of patients, workers and the general public from ionising radiation and other physical agents used in their specialty:

- 2.3 Medical Physicists in Malta have strong analytical, problem-solving, and trouble-shooting skills:
 - Strongly AgreeIAgreeINot SureIDisagreeIStrongly DisagreeI

2.4 Medical Physicists in Malta have strong mathematical, statistical and data analysis skills:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

2.5 The Medical Physics profession in Malta has a strong legal foundation:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

2.6 Medical Physicists in Malta have strong scientific research skills:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

2.7 Medical Physicists in Malta have strong information and communication technology skills:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	
Subligiy Disagice	

2.8 Present Medical Physics services in Malta compare well with other EU countries:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

2.9 Do you consider that the qualification and curriculum frameworks leading to the certification of Medical Physics professionals in Malta (Bachelor in Physics/Engineering, Masters in Medical Physics, and 2 years clinical training) are sufficient to ensure the production of high calibre professionals?

Yes	
No	

2.10 Which element might be improved and how?

2.11 Are there any other notable strengths that you can highlight for the MP profession in Malta?

Section 3: Possible Weaknesses of the MP profession in Malta

3.1 One of the problems that the profession faces in Malta is the absence of a universally known mission statement for the profession for Malta:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment.

3.2 The Medical Physics profession in Malta has a narrow range of specialisations:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

2.9 Do you consider that the qualification and curriculum frameworks leading to the certification of Medical Physics professionals in Malta (Bachelor in Physics/Engineering, Masters in Medical Physics, and 2 years clinical training) are sufficient to ensure the production of high calibre professionals?

Yes	
No	

2.10 Which element might be improved and how?

2.11 Are there any other notable strengths that you can highlight for the MP profession in Malta?

Section 3: Possible Weaknesses of the MP profession in Malta

3.1 One of the problems that the profession faces in Malta is the absence of a universally known mission statement for the profession for Malta:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment.

3.2 The Medical Physics profession in Malta has a narrow range of specialisations:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

3.3 The Medical Physics profession in Malta still has not realised the importance of strategic and robust leadership:

Strongly AgreeIAgreeINot SureIDisagreeIStrongly DisagreeI

Please feel free to add comment

3.4 Medical Physicists in Malta have low marketing skills leading to too low profile of the profession within the healthcare:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

3.5 Medical physicists in Malta take part in regional/international networking:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

3.6 The lack of an independent MP department will negatively impact the achievement of the vision:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

3.7 There is a reluctance by some Medical Physicists in Malta to be part of the wider healthcare picture:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

3.8 Medical Physicists in Malta have insufficient communication and pedagogical skills:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

3.9 Research by Medical Physicists in Malta has barely taken off:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

3.10 Our work duties in Malta are too overwhelming to be able to have enough free time to enrol in new training courses or attend conferences regarding the profession:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

3.11 Are there any legal problems which may hinder us from achieving the vision?

Yes	
No	

3.12 If in 3.11 you answered YES, what do you consider are the problems?

3.13 Are there any other notable weaknesses that you can highlight in the MP profession in Malta?

3.14 How can these weaknesses be eliminated?

Section 4: Opportunities of the MP profession in Malta

4.1 The Medical Physics profession can expand as a result of the increase in number and sophistication of hospital medical devices:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

4.2 Medical Physicists in Malta should link up with patient organisations and project themselves as patient advocates regarding safety standards in healthcare:

Strongly Agree	ш
Agree	
Not Sure	
Disagree	
Strongly Disagree	

4.3 Medical Physicists in Malta should link up with occupational health and safety organisations and environmental groups and project themselves as occupational/public advocates with regard to safety from physical agents:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

4.4 The Medical Physics profession should create further opportunities for itself by exerting pressure on legislators to recognise that patient safety should not be limited to ionising radiation and strive to ensure that the rigorous standards of safety to be found in ionising radiation are extended to other areas such as magnetic resonance imaging, lasers and in the future nanodevices:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

4.5 Due to the escalating cost of healthcare, the need for efficient use of medical devices and Health Technology Assessment is vital, and we as Medical Physicists need to take this opportunity and involve ourselves in such issues, as it would raise our profile with health economists:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

4.6 The rapid expansion in the number of home-use, self-testing and wearable devices are not only a clinical opportunity for Medical Physicists but also can be turned into a business opportunity. Shops selling such devices should be manned by Medical Physicists who can give advice to customers on the most appropriate devices to purchase, proper and safe use, re-calibration frequencies and other information. Medical Physicists can also offer quality control services for such devices:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

4.7 The Government of Malta makes available a lot of funding for me to be able to develop my competences:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

4.8 I take part in IPEM courses:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

4.9 I take part in EFOMP school for medical physics experts:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

4.10 I take part in IAEA courses:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

4.11 I take the opportunity to go to conferences regarding the profession:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

4.12 Are there opportunities for research in your work area?

Yes	
No	

Please feel free to add comment

4.13 Are there any other notable opportunities that you can highlight in the MP profession in Malta?

4.14 How can we benefit from your aforementioned opportunities?

Section 5: Threats of the MP profession in Malta

5.1 The BSc(Hons) Physics, Medical Physics and Radiation Protection has solved the threat of low number of medical physicists due to low number of Physics/Engineering graduates:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

5.2 Malta needs more Medical Physicists to achieve western European standards in my specialty area:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

5.3 Austerity economics and unrestrained commoditisation, are a threat to the Medical Physics profession in Malta:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment.

5.4 There is role poaching from other professions in Malta:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

Please feel free to add comment

5.5 The profession is facing unfair competition from other professions:

Please feel free to add comment

5.6 Are there any other notable present or future threats that you can highlight in the MP profession in Malta?

5.7 How can we eliminate or reduce the impact of these threats?

Section 6: Concluding Questions

6.1 Do you have any ideas on how this vision statement can be developed further?

Yes □ No □

6.2 If in 6.1 you answered YES, what are these ideas?

6.3 As a profession we are well prepared to face the challenges of future technologies:

Strongly Agree	
Agree	
Not Sure	
Disagree	
Strongly Disagree	

6.4 Thank you for dedicating your time in filling this questionnaire. Are there any other points which I have not mentioned regarding this subject that you think would be helpful for me to know?

Yes □ No □

6.5 If in 6.4 you answered YES, what are these points?

Thank you for your time.

Appendix J

Questionnaire Evaluation Sheet

Questionnaire Evaluation Sheet

Please mark the boxes with a ☑ as appropriate.

1. Do you think the questionnaire was too short/long/appropriately long given the research subject?

Too short □ Too long □ Appropriately long given the research subject □

2. Was it easy to follow the general instructions of the questionnaire?

Yes□ No □

If you answered NO, which instructions were not clear and why?

3. In your opinion was the questionnaire well structured?

Yes□ No □

If you answered NO, what would you suggest should be improved?

4. Were there any questions that you did not understand?

Yes□ No □

If you answered YES, kindly specify which and suggest improvement.

5. Did you find any objection in answering any particular question/s?

Yes□ No □

If you answered YES, kindly specify which.

6. Were there any questions which you consider could have been omitted?

Yes□ No □ If you answered YES, can you kindly specify which and your reason as to why they should be removed?

7. Do you have any further comments or suggestions?

Thank you for your feedback.

From the responses obtained, seven out of eight participants (87.5%) perceived the questionnaire to be appropriately long given the research subject whilst one participant (12.5%) found it too long. The general instructions of the questionnaire were easy to follow as perceived by all the participants (100%). Moreover, the questionnaire was perceived by all (100%) as being well structured. When asked if there were any questions which the participants did not understand, one participant (12.5%) identified Q3.6 "*The lack of an independent MP department will negatively impact the achievement of the vision*", however they did not specify how the question could have been improved. None of the participants found objection in answering any questions. Lastly, none of the participants thought that a question could have been omitted.

Appendix K

FREC Approval

9/4/22, 6:14 AM

University of Malta Mail - FHS-2022-00075 Jason Schembri



Jason Schembri <jason.schembri.17@um.edu.mt>

FHS-2022-00075 Jason Schembri

 Rita Pace Parascandalo <rita.pace-parascandalo@um.edu.mt>
 27 May 2022 at 07:50

 To: Jason Schembri <jason.schembri.17@um.edu.mt>
 27 May 2022 at 07:50

 Cc: Research Ethics HEALTHSCI <research-ethics.healthsci@um.edu.mt>, Carmel J Caruana

 <carmel.j.caruana@um.edu.mt>

Dear Jason,

your recently submitted amendments have been reviewed. Approval for your study is granted oBo FREC. You may proceed with your study and collect the data.

Good luck

Regards Dr Rita PP



Dr Rita Pace Parascandalo PhD (UCLan) BSc(Hons) (Melit.), MSc(Melit.), RM

Senior Lecturer, Department of Midwifery Chairperson, Faculty Research Ethics Committee

Faculty of Health Sciences Office No. 48 +356 2340 1176

rita.pace-parascandalo@um.edu.mt

[Quoted text hidden]

https://mail.google.com/mail/u/1/?ik=d6d11644d2&view=pt&search=all&permmsgid=msg-f%3A1733957408885473225&simpl=msg-f%3A1733957... 1/1